UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): April 26, 2023

FIRST WESTERN FINANCIAL, INC.

(Exact name of registrant as specified in its charter)

001-38595 Colorado 37-1442266 (State or other jurisdiction of (Commission (I.R.S. Employer incorporation or organization) File Number) Identification No.)

1900 16th Street, Suite 1200 Denver, Colorado

80202

(Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code: (303) 531-8100

Former name or former address, if changed since last report: **Not Applicable**

Securities registered pursuant to Section 12(b) of the Act:

Title of each class **Trading Symbol** Name of each exchange on which registered Common Stock, no par value **MYFW** The Nasdag Stock Market LLC

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- O Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
 O Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

- \times Emerging growth company
- If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 5.01 Compensatory Arrangements of Certain Officers

Chairman, President and Chief Executive Officer

The employment agreement by and between First Western Financial, Inc. (the "Company") and Scott C. Wylie, Chairman, President and Chief Executive Officer of First Western Financial, Inc., effective as of January 1, 2017 (the "Agreement"), was amended effective April 26, 2023 (this "Amendment").

In this Amendment, the parties agreed to amend the Agreement to (among other things) more closely align their cash- and equity-based severance compensation to current market practice based on the competitive market data compiled by Alvarez & Marsal.

Except as set forth in this Amendment, the terms of Mr. Wylie's Agreement, remain in full effect. For the other terms of the Agreement, which are not changed by this Amendment, refer to First Western Financial, Inc.'s Exhibit 10.3 on Form S-1, which was filed on June 19, 2018, as amended on January 30, 2020.

The foregoing summary of this Amendment does not purport to be a complete description of the terms and conditions of this Amendment and is qualified in its entirety by the full text of this Amendment attached as Exhibit 10.1, which is incorporated herein by reference.

Treasurer, Chief Operating Officer, and Chief Financial Officer

The employment agreement by and between First Western Financial, Inc. (the "Company") and Julie Courkamp, Treasurer, Chief Operating Officer, and Chief Financial Officer of First Western Financial, Inc., effective as of January 1, 2017, amended and restated as of March 5, 2018 (the "Amended and Restated Employment Agreement"), amended on May 2, 2019 (referred to herein as the "Agreement"), amended January 30, 2020 (the "First Amendment"), was amended effective April 26, 2023 (the "Second Amendment").

In this Second Amendment, the parties agreed to (among other things) more closely align their cash- and equity-based severance compensation to current market practice based on the competitive market data compiled by Alvarez & Marsal.

Except as set forth in this Second Amendment, the terms of Ms. Courkamp's Agreement, remain in full effect. For the other terms of the Agreement, which are not changed by this Second Amendment, refer to First Western Financial, Inc.'s Exhibit 10.4 on Form S-1, which was filed on June 19, 2018, as amended on May 2, 2019 and January 30, 2020.

The foregoing summary of this Second Amendment does not purport to be a complete description of the terms and conditions of this Amendment and is qualified in its entirety by the full text of this Amendment attached as Exhibit 10.2, which is incorporated herein by reference.

Chief Investment Officer

The employment agreement by and between First Western Financial, Inc. (the "Company") and John Sawyer, Chief Investment Officer of First Western Financial, Inc., effective as of April 8, 2020, was amended effective April 26, 2023 (this "Amendment").

In this Amendment, the parties agreed to (among other things) more closely align their cash- and equity-based severance compensation to current market practice based on the competitive market data compiled by Alvarez & Marsal.

Except as set forth in this Amendment, the terms of Mr. Sawyer's Agreement, remain in full effect. For the other terms of the Agreement, which are not changed by this Amendment, refer to First Western Financial, Inc.'s Exhibit 10.4 on Form S-1, which was filed on April 10, 2020.

The foregoing summary of this Amendment does not purport to be a complete description of the terms and conditions of this Amendment and is qualified in its entirety by the full text of this Amendment attached as Exhibit 10.3, which is incorporated herein by reference.

Item 8.01 Other Events

Amendment and Restatement of the First Western Financial, Inc. 2016 Omnibus Incentive Plan ("the 2016 Plan")

On April 26, 2023, the Board approved an amendment and restatement of the 2016 Plan to provide for continued vesting of Awards held by a "retirement eligible employee" following his or her "retirement". "Retirement" is defined as a retirement eligible employee's voluntary resignation if certain conditions are met, including: (a) the retirement eligible employee has provided at least one year advanced written notice of resignation to the Company; (b) the retirement eligible employee enters into a non-compete agreement or a non-solicitation agreement if the retirement eligible employee is not subject to a non-compete agreement or a non-solicitation agreement; and (c) the retirement eligible employee delivering and not revoking general release and waiver on a form approved by the Company. A "retirement eligible employee" is generally defined as an employee who has attained age 65 and completed 10 years of employment with Company and its Subsidiaries.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits.

Date: May 2, 2023

| Exhibit Number | Description |
|-------------------|--|
| 10.1 | Amended and Restated Employment Agreement dated April 26, 2023 by and between First Western Financial, Inc. and Scott C. Wylie |
| | |
| 10.2 | Second Amended and Restated Employment Agreement dated April 26, 2023 by and between First Western Financial, Inc. and Julie |
| | <u>Courkamp</u> |
| 10.3 | Amended and Restated Employment Agreement dated April 26, 2023 by and between First Western Financial, Inc. and John Sawyer |
| 10.4 | Amendment to 2016 Omnibus Incentive Plan dated April 26, 2023 |

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

FIRST WESTERN FINANCIAL, INC.

By: /s/ Julie A. Courkamp

Julie A. Courkamp Chief Financial Officer

AMENDED AND RESTATED EMPLOYMENT AGREEMENT

THIS AMENDED AND RESTATED EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") effective as of April 26, 2023 (the "<u>Effective Date</u>") is by and between First Western Financial Inc., a Colorado corporation (the "<u>Company</u>" or "<u>FWFI</u>"), and Scott C. Wylie, an individual resident of the State of Colorado (the "<u>Executive</u>"). This Agreement amends, supersedes and replaces, in its entirety, the Employment Agreement between the Company and the Executive dated effective as of January 1, 2017, as amended by that certain Amendment to Employment Agreement dated January 30, 2020 (together, the "<u>Prior Agreement</u>").

WHEREAS, the Executive is currently employed by the Company as its Chairman, President and Chief Executive Officer; and

WHEREAS, the Executive also serves as Chairman, President and CEO of the Company's wholly owned subsidiary bank, First Western Trust Bank, as well as any other any other subsidiaries unless specifically excluded (collectively, the "Affiliates"); and

WHEREAS, the Executive and the Company would like to amend and restate the Prior Agreement to more closely align the Executive's severance compensation to current market practice.

NOW, THEREFORE, in consideration of the foregoing, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

1. <u>Employment and Duties</u>.

- (a) <u>General</u>. As of the Effective Date, the Executive shall continue to serve as the Chairman, President and Chief Executive Officer of the Company and the Bank. The Executive shall report directly to the Board of Directors of the Company (the "<u>Board</u>"). The Executive shall have such duties and responsibilities, commensurate with the Executive's position, as may be assigned to the Executive from time to time by the Board, and he shall serve as a member of the Board and on the boards of directors or other governing bodies of the Affiliates during the Term (as defined herein). The Executive shall so serve on the aforementioned Board or boards, unless otherwise removed, without additional compensation therefor; <u>provided</u>, <u>however</u>, that the Executive shall be entitled to reimbursement of expenses incurred by the Executive in connection with such Board service. The Executive hereby accepts such employment and agrees to render the services described above.
- (b) Exclusive Services. For so long as the Executive is employed by the Company, the Executive shall devote his full-time working time to his duties hereunder, shall faithfully serve the Company, shall in all material respects conform to and comply with the lawful directions and instructions given to him by the Board and shall use his reasonable best efforts to promote and serve the interests of the Company. Further, the Executive shall not, directly or indirectly, render services to any other person or organization without the consent of the Board or otherwise engage in activities that would interfere in any material respect with his faithful performance of his duties hereunder. Notwithstanding the foregoing, (i) the Executive may serve on such other for-profit corporate boards as may be consented to by the Board, provided that such activity does not contravene the first sentence of this Section 1(b), and (ii) the Executive may serve on not-for-profit corporate, civic or charitable boards or engage in charitable activities without remuneration therefor as may be consented to by the Board, provided that such activity does not contravene the first sentence of this Section 1(b).

121986034v.

2. <u>Term.</u> The term of the Executive's employment under this Agreement shall commence on the Effective Date and shall expire on the December 31, 2023. This Agreement shall automatically renew for successive one-year terms commencing January I, 2024, unless either party gives notice to the other 90 days before the end of a particular term. The period during which the Agreement is in effect shall be referred to as the "<u>Term</u>."

3. Definitions.

- (a) "Base Amount" means the amount equal to the sum of (i) the Executive's Base Salary at the rate then in effect and (ii) amount equal to the greater of (A) the Executive's target Annual Bonus at the rate then if effect or (B) the average Annual Bonus for the three fiscal years immediately preceding the date on which the Executive's termination occurs.
 - (b) Termination for "Cause" means termination of the Executive's employment because of:
 - (i) the willful failure by the Executive to perform the Executive's duties with the Company;
 - (ii) gross incompetence or gross negligence in the discharge of the Executive's duties;
 - (iii) willful dishonesty, theft, embezzlement, fraud, breach of confidentiality, or unauthorized disclosure or use of financial information, confidential client information, client or employee lists, trade secrets, or other Company confidential or proprietary information;
 - (iv) willful violation of any law, rule or regulation of any governing authority or of the Company's policies and procedures, including, without limitation, the Company's employee handbook or similar document;
 - (v) the willful refusal of Executive to follow the lawful directions of the Company's Board of Directors within a reasonable period after delivery to Executive of written notice of such directions;
 - (vi) willful conduct that is grossly injurious to the reputation, financial condition, business or assets of the Company; or
 - (vii) willful breach of any material provision in an agreement with the Company.

In each of (i) through (vii) the Executive shall be given written notice of such cause for termination, and in each of (i) and (vii) the Executive shall be given an opportunity to remedy such cause for termination within sixty (60) business days of receipt of such notice.

- (c) <u>"Change in Control"</u> shall have the same meaning as the definition contained in Section 12.2 of the First Western Financial, Inc. 2016 Omnibus Incentive Plan.
- (d) Resignation for "<u>Good Reason</u>" means termination of employment by the Executive because of the occurrence of any of the following events:
 - (i) there is a material reduction in the Executive's Base Salary, unless agreed to in writing by the Executive;

- (ii) there is a material reduction in the Executive's authority, duties, or responsibilities;
- (iii) the Executive does not continue to retain the title of Chairman of the Board of the Company, unless agreed to in writing by the Executive;
 - (iv) the failure of any successor to assume this Agreement; and
- (v) any other action or inaction that constitutes a material breach by the Company of this Agreement after the Executive provides written notice to the Company of the facts which constitute the grounds within sixty (60) business days following the initial existence of the grounds and the Company thereafter fails to cure such grounds within sixty (60) business days following its receipt of such notice (or, in the event that such grounds cannot be corrected within such sixty (60) day period, the Company has not taken all reasonable steps within such sixty (60) day period to correct such grounds as promptly as practicable thereafter).
- 4. <u>Compensation and Other Benefits</u>. Subject to the provisions of this Agreement, the Company shall pay and provide the following compensation and other benefits to the Executive during the Term as compensation for services rendered hereunder:
- (a) <u>Base Salary.</u> The Company shall pay to the Executive an annual salary (the "<u>Base Salary</u>") payable in substantially equal installments at such intervals as may be determined by the Company in accordance with its ordinary payroll practices, as established from time to time. The Base Salary shall be reviewed by the Board of Directors annually, and increased as appropriate, for market changes. Commencing effective as of the Effective Date, Executive's annualized Base Salary is \$575,000. The Base Salary shall not be decreased by the Company except with the prior written consent of the Executive.
- (b) <u>Annual Bonus</u>. Executive is eligible to receive annual incentive compensation (the "<u>Annual Bonus</u>") as governed by the terms set forth in the First Western Financial, Inc. Incentive Plan for Senior Executive Officers, as may be amended from time to time. The incentive compensation performance measures and goals are reviewed by the Board of Directors annually, and adjusted as appropriate, according to the needs of the business.
- (c) <u>Long-Term Incentive Plan</u>. The Executive shall be eligible for grants under the First Western Financial, Inc. 2016 Omnibus Incentive Plan, including, but not limited to, grants of stock options, market conditioned performance share units, financial conditioned performance stock units and restricted stock units, as the Compensation Committee of the Board shall determine from time to time.
- (d) <u>Savings and Retirement Plans</u>. The Executive shall be eligible to participate in all savings and retirement plans applicable generally to other executives of the Company, in accordance with the terms of the plans, as may be amended from time to time.
- (e) <u>Welfare Benefit Plans</u>. The Executive and his eligible dependents shall be eligible to participate in and shall receive all benefits under the Company's welfare benefit plans and programs applicable generally to other executives of the Company, in accordance with the terms of the plans, as may be amended from time to time.
- (f) <u>Expenses</u>. Upon presentation of written documentation thereof, in accordance with the applicable expense reimbursement policies and procedures of the Company as in effect from time to time, the Company shall reimburse the Executive for reasonable business-related expenses incurred by the Executive in the fulfillment of his duties. Payments

with respect to reimbursements of expenses shall be made promptly and in accordance with the applicable expense reimbursement policies and procedures of the Company, but in any event, on or before the last day of the calendar month following the calendar month in which the relevant expense is incurred.

- (g) <u>Vacation</u>. The Executive shall be entitled to vacation each calendar year during the Term, subject to and in accordance with the Company's vacation policy in effect from time to time.
- 5. <u>Termination of Employment</u>. The terms of any equity compensation grants outstanding as of the Effective Date, as well as any future equity compensation grants made under the First Western Financial, Inc. 2016 Omnibus Incentive Plan shall govern what the Executive receives on termination of employment, in terms of equity compensation, except as expressly provided in Section 5(b)(i)(C) and Section 5(b)(ii)(C) hereof. All other forms of remuneration the Executive is eligible for on termination of employment are set forth in this Section 5.
- (a) Termination for Cause; Resignation Without Good Reason. If, prior to the expiration of the Term, the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code"), and the Treasury Regulations ("Regulations") thereunder, by reason of the Company's termination of the Executive's employment for Cause, or if the Executive resigns from his employment hereunder other than for Good Reason, the Executive shall be entitled only to payment of (i) any unpaid Base Salary through and including the date of termination or resignation, (ii) any Annual Bonus earned, but unpaid, for the year immediately preceding the year in which the termination date occurs (which unpaid Annual Bonus amount shall be paid no later than March 15 of the year following the year in which the amount was earned), and (iii) any other amounts or benefits required to be paid or provided by law or under any plan, program, policy or practice of the Company (the amounts or benefits in (i) through (iii) being referred to collectively as the "Other Accrued Compensation and Benefits"). Except as set forth in this subsection (a), the Executive shall have no further right to receive any other compensation or benefits after such termination or resignation of employment.

(b) <u>Termination without Cause; Resignation for Good Reason.</u>

- (i) If the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Code and the Regulations thereunder, by reason of the Company's termination of the Executive's employment without Cause, or if the Executive resigns from his employment hereunder for Good Reason, the Executive shall be entitled to the following:
 - (A) The Other Accrued Compensation and Benefits.
- (B) An amount equal to two (2) times the Executive's Base Amount, which shall be payable in equal installments pursuant to the Company's normal payroll practices and subject to all legally required and customary withholdings for the twenty-four (24) month period following termination.
- (C) A pro-rated portion of each outstanding unvested equity award, including, without limitation, stock options, restricted stock and restricted stock units, held by the Executive shall automatically become vested and, if applicable, exercisable and any restrictions thereon shall immediately lapse. The pro-rated portion, determined separately for each vesting tranche for each such equity award, shall be based on a fraction, (I) the numerator of which is number of days that elapse from the grant date

of the equity award until the Executive's termination date and the (II) the denominator of which is the total number of days in the period that begins on the date of grant of the applicable equity award and ends on the scheduled vesting date of such equity award (or, if applicable, the scheduled vesting date of the applicable vesting tranche). For purposes of determining the number of shares subject to any outstanding unvested equity awards subject to performance vesting conditions (as opposed to solely service-based vesting conditions) that would otherwise vest pursuant to the foregoing sentence, the applicable performance goals shall be deemed achieved at the level of the Company's actual achievement of the applicable performance goals as measured as of the Executive's termination date, as determined by the Compensation Committee in its reasonable discretion. Each of the Executive's outstanding equity awards as of the Effective Date is hereby amended to incorporate the terms of this Section 5.1(b)(i)(C).

(D) The Company shall monthly pay to the Executive the amount equal to the full premium amount (determined as of the date of termination) for continued coverage under the Company's health plan pursuant to the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), for the Executive, and, to the extent that the Executive is providing coverage for his spouse or eligible dependents as of the termination date, for such individuals; provided, however, that the Company's obligation to pay such premiums shall cease immediately upon the earlier of (i) the passage of eighteen (18) months (ii) the expiration of the statutory COBRA period and (iii) the date the Executive becomes eligible for coverage under any other group health plan (as an employee or otherwise) or Medicare.

Notwithstanding the foregoing, if the Company terminates the Executive's employment without Cause, the Company shall provide the Executive with no less than ninety (90) days' written notice or payment of three (3) months Base Salary in lieu of ninety (90) days' written notice, which shall be in addition to payments described under this Section 5(b)(i).

- (E) All payments and benefits provided under this Section 5(b)(i)(B) and (D) shall commence on the first payroll date following the 60th day after the Executive's termination of employment, with the first installment payment to include any payments that would have otherwise been paid to the Executive if such payments commenced on the first payroll date following the Executive's termination date. The Company shall not be required to make the payments and provide the benefits provided for under this Section 5(b)(i)(A), (B), (C) or (D) unless the Executive executes and delivers to the Company, within sixty (60) days following the Executive's termination of employment, a release substantially in the form attached hereto as Exhibit A (the "Release"), and the Release has become effective and irrevocable in its entirety in such 60-day period. The Executive's failure or refusal to sign the Release (or the Executive's revocation of such Release in accordance with applicable laws) will result in the forfeiture of the payments and benefits under this Section 5(b)(i)(A), (B), (C) or (D). To the extent any amount payable under this Section 5 is deferred compensation subject to the Code, if the period during which the Executive has discretion to execute or revoke the Release straddles two of the Executive's taxable years, then the Company shall make the severance payments starting in the second of such taxable years, regardless of which taxable year the Executive actually delivers the executed Release to the Company. The Executive may not, directly or indirectly, designate the calendar year or timing of payments. This Section 5(b)(i) shall expressly not apply to payments made on account of a Change in Control, pursuant to Section 5(b)(ii) hereof.
- (F) If, following a termination of employment without Cause or a resignation for Good Reason, the Executive breaches the provisions of Sections 6

through 10 hereof or breaches any provision set forth in the executed copy of the general release of claims, the Executive shall not be eligible, as of the date of such breach, for the payments and benefits described in Section 5(b)(i)(A), (B), (C) or (D), and any and all obligations and agreements of the Company with respect to such payments shall thereupon cease. This Section 5(b)(i)(F) shall expressly not apply to payments made on account of a Change in Control, pursuant to Section 5(b)(i) hereof.

- (ii) If the Company undergoes a Change in Control and, within 24 months of such Change in Control, the Executive is terminated without Cause or resigns for Good Reason, then, in lieu of (and not in addition to) any payments or benefits payable to the Executive pursuant to Section 5(b)(i), the Executive shall be entitled to the following:
 - (A) The Other Accrued Compensation and Benefits.
- (B) An amount equal to 2.99 times the Executive's Base Amount, payable in a single lump sum payment.
- (C) Each outstanding equity award, including, without limitation, stock options, restricted stock and restricted stock units, held by the Executive shall automatically become vested and, if applicable, exercisable and any restrictions thereon shall immediately lapse, in each case, with respect to that number of shares of Company common stock that would otherwise vest based on Executive's continued employment. For purposes of determining the number of shares subject to any outstanding equity awards subject to performance vesting conditions that would otherwise vest pursuant to the foregoing sentence, the applicable performance goals shall be deemed achieved at the "target" level. Each of the Executive's outstanding equity awards as of the Effective Date is hereby amended to incorporate the terms of this Section 5.1(b)(ii)(C).
- (D) Monthly payments to the Executive equal to the full premium amount (determined as of the date of termination) for continued coverage under the Company's health plan pursuant to COBRA for the Executive, and, to the extent that the Executive is providing coverage for his spouse or eligible dependents as of the termination date, for such individuals; provided, however, that the Company's obligation to pay such premiums shall cease immediately upon the earlier of (i) the passage of twenty-four (24) months (ii) the expiration of the statutory COBRA period and (iii) the date the Executive becomes eligible for coverage under any other group health plan (as an employee or otherwise) or Medicare.

Notwithstanding the foregoing, if the Company (or its successor) terminates the Executive's employment without Cause, the Company (or its successor) shall provide the Executive with no less than ninety (90) days' written notice or payment of three (3) months Base Salary in lieu of ninety (90) days' written notice, which shall be in addition to payments described under this Section 5(b)(ii).

(E) Unless otherwise provided herein, all payments and benefits provided under this Section 5(b)(ii)(B) and (D) shall be paid (or, in the case of the payments described in Section 5(b)(ii)(D), commence to be paid) on the first payroll date following the 60th day after the Executive's termination of employment, with the first payment to include any payments provided under Section 5(b)(ii)(D) that would have otherwise been paid to the Executive if such payments commenced on the first payroll date following the Executive's termination date. The Company shall not be required to make the payments and provide the benefits provided for under this Section

- 5(b)(ii)(A), (B), (C) or (D) unless the Executive executes and delivers the Release to the Company, within sixty (60) days following the Executive's termination of employment, and the Release has become effective and irrevocable in its entirety in such 60-day period. The Executive's failure or refusal to sign the Release (or the Executive's revocation of such Release in accordance with applicable laws) will result in the forfeiture of the payments and benefits under this Section 5(b)(ii)(A), (B), (C) or (D). To the extent any amount payable under this Section 5 is deferred compensation subject to the Code, if the period during which the Executive has discretion to execute or revoke the Release straddles two of the Executive's taxable years, then the Company shall make the severance payments starting in the second of such taxable years, regardless of which taxable year the Executive actually delivers the executed Release to the Company. The Executive may not, directly or indirectly, designate the calendar year or timing of payments.
- (F) In the event that it is determined that any payment or distribution of any type to or for the benefit of an Executive made by the Company, by any of its Affiliates, by any person who acquires ownership or effective control or ownership of a substantial portion of the Company's assets (within the meaning of Code Section 280G) or by any affiliate of such person, whether paid or payable or distributed or distributable pursuant to the terms of any equity compensation plan, this Agreement or otherwise (the "Total Payments"), would be subject to the excise tax imposed by Code Section 4999 or any interest or penalties with respect to such excise tax (the "Excise Tax"), then, notwithstanding any other provision of this Agreement or any equity compensation plan to the contrary, any right of the Executive to any payment or benefit under this Agreement or any such equity compensation plan shall be reduced or eliminated, but only to the extent necessary to avoid imposition of the Excise Tax. In no case, however, shall such cutback be made if Total Payments after the imposition of the Excise Tax are greater than Total Payments cut back as provided in this Section 5(b) (ii) to avoid the Excise Tax.
- (G) In the event that a cutback of Total Payments is permitted under Section 5(b)(ii)(F), and except as required by Code Section 409A or to the extent that Code 409A permits discretion, the Compensation Committee shall have the right, in the Compensation Committee's sole discretion, to designate those rights, payments, or benefits and all other agreements that should be reduced or eliminated so as to provide the Executive with the maximum pre-tax amount which avoids imposition of the Excise Tax. For example, the Compensation Committee may choose to cut back cash severance, if that would yield a higher pre-tax amount than cutting back equity. Notwithstanding the foregoing, to the extent any payment or benefit constitutes deferred compensation under Code Section 409A, in order to comply with Code Section 409A the Compensation Committee shall instead accomplish such reduction by first reducing or eliminating any cash payments (with the payments to be made furthest in the future being reduced first), then by reducing or eliminating any accelerated vesting of options or stock appreciation rights, then by reducing or eliminating any accelerated vesting of restricted stock or restricted stock units.
- (c) <u>Termination Due to Death or Disability</u>. The Executive's employment with the Company shall terminate automatically on the Executive's death. In the event of the Executive's Disability (as defined herein), the Company shall be entitled to terminate his employment. In the event of the Executive's death or if the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Code, or the Regulations thereunder, by reason of the Executive's Disability, the Company shall pay to the Executive (or his estate, as applicable), (i) the Executive's Base Salary through and including the date of termination and any Other Accrued Compensation and Benefits (ii) a pro-rata Annual Bonus for the year of

termination, based on actual audited year-end results and payable when bonuses are normally paid to employees, and (iii) three (3) months Base Salary at the rate then in effect, payable in equal installments pursuant to the Company's normal payroll practices and subject to all legally required and customary withholdings for the three (3) month following termination.

- (i) For purposes of this Agreement, "<u>Disability</u>" means a physical or mental disability or infirmity of the Executive that prevents the normal performance of substantially all his duties for a period in excess of ninety (90) consecutive days or for more than ninety (90) days in any consecutive twelve (12)-month period. Evidence of such physical or mental disability or infirmity shall be certified by a physician licensed to practice in the state of residence of the Executive, which physician is mutually agreeable to the Board and the Executive. If there is no agreement on the selection of the physician, then the Board shall select one physician and the Executive shall select one physician, and the two physicians shall attempt to mutually agree upon such physical or mental disability or infirmity. If the two physicians cannot agree, then the two physicians shall jointly select a third physician, whose opinion on such physical or mental disability or infirmity shall control.
- (d) Notice of Termination. Any termination of employment by the Company or the Executive shall be communicated by a written "Notice of Termination" to the other party hereto given in accordance with Section 24 of this Agreement. In the event of a termination by the Company for Cause, or by the Executive for Good Reason, the Notice of Termination shall (i) indicate the specific termination provision in this Agreement relied upon, (ii) set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of the Executive's employment under the provision so indicated and (iii) specify the date of termination, which date shall not be more than thirty (30) business days after the giving of such notice, provided that the date of termination will not occur before the expiration of any applicable cure period.

The failure by the Executive or the Company to set forth in the Notice of Termination any fact or circumstance which contributes to a showing of Good Reason or Cause shall not waive any right of the Executive or the Company, respectively, hereunder or preclude the Executive or the Company, respectively, from asserting such fact or circumstance in enforcing the Executive's or the Company's rights hereunder.

- (e) <u>No further Rights</u>. The Executive shall have no further rights under this Agreement or otherwise to receive any other compensation or benefits after such termination or resignation of employment.
 - Confidentiality.
 - (a) <u>Confidential Information</u>.
 - (i) The Executive agrees that he will not at any time, except with the prior written consent of the Company its or its Affiliates or as required by applicable law, directly or indirectly, reveal to any person, entity or other organization (other than the Company its or its Affiliates or its respective employees, officers, directors, shareholders or agents) or use for the Executive's own benefit any confidential or proprietary information of any member of the Company its or its Affiliates ("Confidential Information") relating to the assets, liabilities, employees, goodwill, business or affairs of any member of the Company its or its Affiliates, including, without limitation, any information concerning past, present or prospective clients, intellectual capital, marketing data, or other confidential information used by, or useful to, any member of and known to the Executive by reason of the Executive's employment by, shareholdings in or other

association with the Company its or its Affiliates, other than disclosure while employed by the Company which the Executive reasonably and in good faith believes to be in or not opposed to the interests of the Company; <u>provided</u> that such Confidential Information does not include any information which is available to the general public or is generally available within the relevant business or industry other than as a result of the Executive's breach of this Agreement. Confidential Information may be in any medium or form, including, without limitation, physical documents, computer files or disks, videotapes, audiotapes, and oral communications.

- (ii) In the event that the Executive becomes legally compelled to disclose any Confidential Information, the Executive shall, if permitted by law, provide the Company with prompt written notice so that the Company may seek a protective order or other appropriate remedy. In the event that such protective order or other remedy is not obtained, the Executive shall furnish only that portion of such Confidential Information or take only such action as is legally required by binding order and shall exercise his reasonable efforts to obtain reliable assurance that confidential treatment shall be accorded any such Confidential Information, The Company shall promptly pay (upon receipt of invoices and any other documentation as may be requested by the Company) all reasonable expenses and fees incurred by the Executive, including attorneys' fees, in connection with his compliance with the immediately preceding sentence. Notwithstanding anything herein to the contrary, nothing in this Agreement shall (A) prohibit the Executive from making reports of possible violations of federal law or regulation to any governmental agency or entity in accordance with the provisions of and rules promulgated under Section 21F of the Securities Exchange Act of 1934 or Section 806 of the Sarbanes-Oxley Act of 2002, or of any other whistleblower protection provisions of state or federal law or regulation, or (B) require notification or prior approval by the Company of any reporting described in clause (A).
- (b) <u>Confidentiality of Agreement</u>. The Executive agrees that, except as may be required by applicable law or legal process, during the Term and thereafter, he shall not disclose the terms of this Agreement to any person or entity other than the Executive's accountants, financial advisors, attorneys or spouse, provided that such accountants, financial advisors, attorneys and spouse agree not to disclose the terms of this Agreement to any other person or entity.
- (c) <u>Exclusive Property</u>. The Executive confirms that all Confidential Information is and shall remain the exclusive property of the Company its or its Affiliates. All business records, papers and documents kept or made by the Executive relating to the business of the Company its or its Affiliates shall be and remain the property of the Company its or its Affiliates. Upon the request and at the expense of the Company its or its Affiliates, the Executive shall promptly make all disclosures, execute all instruments and papers and perform all acts reasonably necessary to vest and confirm in the Company its or its Affiliates, fully and completely, all rights created or contemplated by this Section 6.
- 7. Noncompetition. The Executive agrees that, for a period commencing on January 1, 2017, and ending two (2) years following the Executive's termination of employment (the "Restricted Period"), the Executive shall not, without the prior written consent of the Company, directly or indirectly, and whether as principal or investor or as an employee, officer, director, manager, partner, consultant, agent or otherwise, alone or in association with any other person, firm, corporation or other business organization, carry on a Competing Business (as defined herein) within the County of Denver and contiguous counties. For purposes of this Section 7: (a) carrying on a "Competing Business" means to engage in the competing business of any business carried on by the Company its or its Affiliates. Notwithstanding the foregoing, nothing herein shall limit the Executive's right to own not more than one (1) percent of any of

the debt or equity securities of any business organization that is then filing reports with the Securities and Exchange Commission pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended. The Executive acknowledges that this noncompetition provision will not unreasonably impair or infringe upon the Executive's right to work or earn a living during the Restricted Period, and the Executive further acknowledges that these restrictions are necessary to protect the trade secrets of the Company as the Executive's breach of this provision would necessarily involve the Executive's use of the Company's trade secrets.

Initial

8. Non-Solicitation. The Executive agrees that for the Restricted Period the Executive shall not, directly or indirectly, (a) interfere with or attempt to interfere with the relationship between any person who is, or was during the Restricted Period or the 3-month period immediately preceding the commencement of the Restricted Period, an employee, officer, representative or agent of the Company its or its Affiliates and any member of the Company its or its Affiliates, or solicit, induce or attempt to solicit or induce any of them to terminate their employment or service relationship with any member of the Company its or its Affiliates or violate the terms of their respective service contracts, or any employment arrangements, with such entities, provided that the foregoing shall not prevent general employment or service solicitations that do not specifically target any such persons; or (b) induce or attempt to induce any customer or client of any member of the Company its or its Affiliates to cease doing business with any member of the Company its or its Affiliates, or in any way interfere with the relationship between any member of the Company its or its Affiliates and any customer or client of any member of the Company its or its Affiliates. The Executive acknowledges that this non-solicitation provision will not unreasonably impair or infringe upon the Executive's right to work or earn a living during the Restricted Period, and the Executive further acknowledges that these restrictions are necessary to protect the trade secrets of the Company as the Executive's breach of this provision would necessarily involve the Executive's use of the Company's trade secrets.

Initial ____

- 9. <u>No Conflicting Agreement</u>. The Executive represents, warrants and covenants to the Company that the Executive is not a party to any agreement, whether written or oral, that would be breached by or would prevent or interfere with the executive of the Executive of this Agreement or the fulfillment by the Executive of the Executive's obligations hereunder.
- 10. <u>Nondisparagement</u>. Each party represents, warrants and covenants to the other that at no time during the Term or thereafter shall such party make, or cause or assist any other person to make, any statement or other communication to any third party which impugns or attacks, or is otherwise critical of, the reputation, business or character of the other party or any of its respective directors, officers or employees, as applicable; provided this Section shall not prohibit truthful testimony by or on behalf of either party in any judicial or administrative proceeding.
- 11. Section 409A of the Code. This Agreement is intended to meet the requirements of Section 409A of the Code and shall be interpreted and construed consistent with that intent. Notwithstanding any other provision of this Agreement, to the extent that the right to any payment (including the provision of benefits) hereunder provides for the "deferral of compensation" within the meaning of Section 409A(d)(l) of the Code, if the Executive is a "Specified Employee" within the meaning of Section 409A(a)(2)(B)(i) of the Code on the date of the Executive's "Separation from Service" within the meaning of Section 409A(a)(2)(A)(i) of the Code (the "Separation Date"), then no such payment shall be made or commence during the period beginning on the Separation Date and ending on the date that is six months following the Separation Date or, if earlier, on the date of the Executive's death. The amount of any payment

that would otherwise be paid to the Executive during this period shall instead be paid to the Executive on the fifteenth day of the first calendar month following the end of the period.

12. Certain Remedies.

- (a) <u>Forfeiture/Payment Obligations</u>. In the event the Executive fails to comply with Sections 6 through 10., other than any isolated, insubstantial and inadvertent failure, the Executive agrees that he will forfeit any amounts not already paid pursuant to Section 5(b)(i)(A), (B), (C) or (D) of this Agreement. Notwithstanding the previous sentence, the Executive shall be given written notice of each alleged failure to comply with Sections 6 through 10, and the Executive shall be given an opportunity to remedy such failure within (60) sixty business days of the receipt of such notice. For purposes of clarity, the Executive's failure to comply with Sections 6 through 10 shall not result in forfeiture of amounts required to be paid but not already paid on account of a Change in Control pursuant to Section 5(b)(ii) of this Agreement.
- (b) <u>Injunctive Relief.</u> Without intending to limit the remedies available to the Company its or its Affiliates, including, but not limited to, that set forth in Section 12(a) hereof, the Executive agrees that a breach of any of the covenants contained in Sections 6 through 10 of this Agreement may result in material and irreparable injury to the Company its or its Affiliates for which there is no adequate remedy at law, that it will not be possible to measure damages for such injuries precisely and that, in the event of such a breach or threat thereof, any member of the Company its or its Affiliates shall be entitled to seek a temporary restraining order or a preliminary or permanent injunction, or both, without bond or other security, restraining the Executive from engaging in activities prohibited by the covenants contained in Sections 6 through 10 of this Agreement or such other relief as may be required specifically to enforce any of the covenants contained in this Agreement. Such injunctive relief in any court shall be available to the Company its or its Affiliates in lieu of, or prior to or pending determination in, any arbitration proceeding.
- 13. <u>Defense of Claims</u>. The Executive agrees that, during the Term, and for a period of seven (7) years after termination of the Executive's employment, upon request from the Company, the Executive will reasonably cooperate with the Company in the defense of any claims or actions that may be made by or against the Company that affect the Executive's prior areas of responsibility, except if the Executive's reasonable interests are adverse to the Company in such claim or action. The Company agrees to promptly pay in advance or reimburse the Executive for, as requested by the Executive, all of the Executive's reasonable travel and other direct costs and expenses incurred, or to be reasonably incurred, to comply with the Executive's obligations under this Section 13, including, but not limited to, legal costs and expenses.
- 14. Alternative Dispute Resolution. The Company and the Executive agree that any dispute that arises out of or relates to Executive's employment or termination of employment with the Company, including any dispute that the Executive may have with any present or former officer, manager, director, employee, agent, attorney or insurer of the Company, shall first be submitted to mediation through the Institute for Conflict Prevention & Resolution ("CPR") (or such other nationally-recognized alternative dispute resolution service as the Executive and Company may agree). The Executive and the Company shall use their reasonable efforts to commence and conclude such mediation in a prompt manner. If the dispute is not resolved through mediation within thirty (30) days after notice thereof, such dispute shall be resolved by binding arbitration in accordance with the rules and procedures of the CPR (or such other nationally recognized alternative dispute resolution service as the Executive and the Company may agree). Judgment upon the award rendered by the arbitrator may be entered in any court having in person and subject matter jurisdiction. The Company and the Executive hereby submit to the jurisdiction of the federal and state courts in Denver, Colorado, for the

purpose of confirming any such award and entering judgment thereon. The Company shall pay for all administrative costs and fees charged by the CPR (or such other nationally recognized alternative dispute resolution service) as well as the fees charged by the arbitrator. Each party shall pay for his or its attorneys' fees and costs.

- 15. <u>Nonassignability; Binding Agreement</u>.
- (a) <u>By the Executive</u>. This Agreement and any and all rights, duties, obligations or interests hereunder shall not be assignable or delegable by the Executive.
- (b) <u>By the Company</u>. This Agreement and all of the Company's rights and obligations hereunder shall not be assignable by the Company except as incident to a reorganization, merger or consolidation, or transfer of all or substantially all of the Company's assets.
- (c) <u>Binding Effect</u>. This Agreement shall be binding upon, and inure to the benefit of, the parties hereto, any successors to or assigns of the Company and the Executive's heirs and the personal representatives of the Executive's estate.
- 16. <u>Withholding</u>. Any payments made or benefits provided to the Executive under this Agreement shall be reduced by any applicable withholding taxes or other amounts required to be withheld by law or contract.
- 17. <u>Amendment: Waiver</u>. This Agreement may not be modified, amended or waived in any manner, except by an instrument in writing signed by both parties hereto. The waiver by either party of compliance with any provision of this Agreement by the other party shall not operate or be construed as a waiver of any other provision of this Agreement, or of any subsequent breach by such party of a provision of this Agreement.
- 18. <u>Governing Law and Forum</u>. The Executive and the Company agree that this Agreement and all matters or issues arising out of or relating to the Executive's employment with the Company shall be governed by the laws of the State of Colorado applicable to contracts entered into and performed entirely therein. Any action to enforce this Agreement shall be brought solely in the state or federal courts located in the County of Denver, Colorado.
- 19. <u>Survival of Certain Provisions</u>. Unless expressly provided otherwise, the rights and obligations set forth in this Agreement shall survive any termination or expiration of this Agreement.
- 20. <u>Entire Agreement: Supersedes Previous Agreements</u>. Except as provided in Section 5(f) hereof, this Agreement contains the entire agreement and understanding of the parties hereto with respect to the matters covered herein, and supersedes all prior or contemporaneous negotiations, commitments, agreements and writings with respect to the subject matter hereof (including, without limitation, the Prior Agreement), all such other negotiations, commitments, agreements and writings shall have no further force or effect, and the parties to any such other negotiation, commitment, agreement or writing shall have no further rights or obligations thereunder.
- 21. <u>Severability</u>. The provisions of the Agreement shall be deemed severable, and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions herein.

- 22. <u>Counterparts</u>. This Agreement may be executed by cither of the parties hereto in counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same instrument.
- 23. <u>Headings</u>. The headings of sections herein arc included solely for convenience of reference and shall not control the meaning or interpretation of any of the provisions of this Agreement.
 - 24. <u>Notices</u>. All notices or communications hereunder shall be in writing, addressed as follows:

To the Company:

First Western Financial, Inc. 1900 16th St Suite #1200 Denver, CO 80202 Attention: Secretary and General counsel

To the Executive:

Scott C. Wylie 435 Monroe Street Denver, CO 80206

All such notices shall be conclusively deemed to be received and shall be effective (i) if sent by hand delivery or nationally recognized courier, upon receipt or (ii) if sent by electronic mail or facsimile, upon receipt by the sender of such transmission.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

| IN WITNESS WHEREOF, the Company has caused this Agreement to be signed by its officer pursuant to | o the |
|--|-------|
| authority of its Board, and the Executive has executed this Agreement, as of the day and year first written above. | |

| FIRST WESTERN FINANCIAL, INC. |
|-------------------------------|
| By: |
| SCOTT C. WYLIE |
| _ |

GENERAL RELEASE OF CLAIMS

| | This Gene | ral Release | of all Claim | ıs (this | "Agreement" |) is enter | ed into by S | Scott C. | Wylie (the | "Executive" |) and |
|--------------|-----------|-------------|--------------|----------|-------------|------------|--------------|----------|------------|-------------|-------|
| FWFI (the "C | | | | | | | | | | | |
| the Company | | | | | | | | | | | |

In consideration of the promises set forth in the Amended and Restated Employment Agreement between the Executive and the Company, dated effective January 1, 2017 (the "Employment Agreement"), the Executive and the Company agree as follows:

1. <u>Return of Property</u>. All Company files, access keys and codes, desk keys, ID badges, computers, records, manuals, electronic devices, computer programs, papers, electronically stored information or documents, telephones and credit cards, and any other property of the Company in the Executive's possession must be returned no later than the date of the Executive's termination from the Company; provided, that, after the notification of a consultation with the Company, the Executive may keep one copy of such items as he may reasonably expect to use to protect his rights under this Agreement.

2. General Release and Waiver of Claims.

- Release. In consideration of the payments and benefits provided to the Executive under the Employment Agreement and after consultation with counsel, the Executive and each of the Executive's respective heirs, executors, administrators, representatives, agents, insurers, successors and assigns (collectively, the "Releasors") hereby irrevocably and unconditionally release and forever discharge the Company, its subsidiaries and affiliates and each of their respective officers, employees, directors, shareholders and agents ("Releasees") from any and all claims, actions, causes of action, rights, judgments, obligations, damages, demands, accountings or liabilities of whatever kind or character (collectively, "Claims"), including, without limitation, any Claims under any federal, state, local or foreign law, that the Releasors may have, or in the future may possess, arising out of (i) the Executive's employment relationship with and service as an employee, officer or director of the Company or any subsidiaries or affiliated companies and the termination of such relationship or service, and any event, condition, circumstance or obligation that occurred, existed or arose on or prior to the date hereof; provided, however, that the Executive does not release, discharge or waive any rights to (i) payments and benefits provided under the Employment Agreement that are contingent upon the execution by the Executive of this Agreement and (ii) any indemnification rights the Executive may have under the Employment Agreement, in accordance with the Company's governance instruments or under any director and officer liability insurance maintained by the Company with respect to liabilities arising as a result of the Executive's service as an officer and employee of the Company. This Section 2(a) does not apply to any Claims that the Releasors may have as of the date the Executive signs this Agreement arising under the Federal Age Discrimination in Employment Act of 1967, as amended, and the applicable rules and regulations promulgated thereunder ("AĎEA"). Claims arising under ADEA are addressed in Section 2(b) of this Agreement.
- (b) <u>Specific Release of ADEA Claims</u>. In further consideration of the payments and benefits provided to the Executive under the Employment Agreement, the Releasors hereby unconditionally release and forever discharge the Releasees from any and all

Claims arising under ADEA that the Releasors may have as of the date the Executive signs this Agreement. By signing this Agreement, the Executive hereby acknowledges and confirms the following: (i) the Executive was advised by the Company in connection with his termination to consult with an attorney of his choice prior to signing this Agreement and to have such attorney explain to the Executive the terms of this Agreement, including, without limitation, the terms relating to the Executive's release of claims arising under ADEA, and the Executive has in fact consulted with an attorney; (ii) the Executive was given a period of not fewer than 21 days to consider the terms of this Agreement and to consult with an attorney of his choosing with respect thereto; (iii) the Executive knowingly and voluntarily accepts the terms of this Agreement; and (iv) the Executive is providing this release and discharge only in exchange for consideration in addition to anything of value to which the Executive is already entitled. The Executive also understands that he has seven days following the date on which he signs this Agreement within which to revoke the release contained in this Section, by providing the Company with a written notice of his revocation of the release and waiver contained in this Section.

(c) <u>No Assignment</u>. The Executive represents and warrants that he has not assigned any of the Claims being released under this Agreement. The Company may assign this Agreement, in whole or in part, to any affiliated company or subsidiary of, or any successor in interest to, the Company.

3. Proceedings.

- (a) <u>General Agreement Relating to Proceedings</u>. The Executive has not filed, and except as provided in <u>Sections 3(b)</u> and <u>3(c)</u>, the Executive agrees not to initiate or cause to be initiated on his behalf, any complaint, charge, claim or proceeding against the Releasees before any local, state or federal agency, court or other body relating to his employment or the termination of his employment, other than with respect to the obligations of the Company to the Executive under the Employment Agreement (each, individually, a "<u>Proceeding</u>"), and agrees not to participate voluntarily in any Proceeding. The Executive waives any right he may have to benefit in any manner from any relief (whether monetary or otherwise) arising out of any Proceeding.
- (b) <u>Proceedings Under ADEA</u>. <u>Section 3(a)</u> shall not preclude the Executive from filing any complaint, charge, claim or proceeding challenging the validity of the Executive's waiver of Claims arising under ADEA (which is set forth in <u>Section 2(b)</u> of this Agreement). However, both the Executive and the Company confirm their belief that the Executive's waiver of claims under ADEA is valid and enforceable, and that their intention is that all claims under ADEA will be waived.
- (c) <u>Certain Administrative Proceedings</u>. In addition, <u>Section 3(a)</u> shall not preclude the Executive from filing a charge with or participating in any administrative investigation or proceeding by the Equal Employment Opportunity Commission or another Fair Employment Practices agency. The Executive is, however, waiving his right to recover money in connection with any such charge or investigation. The Executive is also waiving his right to recover money in connection with a charge filed by any other entity or individual, or by any federal, state or local agency.
- 4. <u>Remedies</u>. In the event the Executive initiates or voluntarily participates in any Proceeding in violation of this Agreement, or if he fails to abide by any of the terms of this Agreement or his post-termination obligations contained in the Employment Agreement, or if he revokes the ADEA release contained in <u>Section 2(b)</u> within the seven-day period provided under <u>Section 2(b)</u>, the Company may, in addition to any other remedies it may have, reclaim any amounts paid to him under Sections 5(b)(i)(A), (B) or (C) of the Employment Agreement or

terminate any benefits or payments that are subsequently due under Sections 5(b)(i)(A), (B) and (C) of the Employment Agreement, without waiving the release granted herein. The Executive acknowledges and agrees that the remedy at law available to the Company for breach of any of his post-termination obligations under the Employment Agreement or his obligations under Sections 2 and 3 herein would be inadequate and that damages flowing from such a breach may not readily be susceptible to measurement in monetary terms. Accordingly, the Executive acknowledges, consents and agrees that, in addition to any other tights or remedies that the Company may have at law or in equity or as may otherwise be set forth in the Employment Agreement, the Company shall be entitled to seek a temporary restraining order or a preliminary or permanent injunction, or both, without bond or other security, restraining the Executive from breaching his post-termination obligations under the Employment Agreement or his obligations under Sections 2 and 3 herein. Such injunctive relief in any court shall be available to the Company, in lieu of, or prior to or pending determination in, any arbitration proceeding.

The Executive understands that by entering into this Agreement he shall be limiting the availability of certain remedies that he may have against the Company and limiting also his ability to pursue certain claims against the Company.

- 5. <u>Severability Clause</u>. In the event that any provision or part of this Agreement is found to be invalid or unenforceable, only that particular provision or part so found, and not the entire Agreement, shall be inoperative.
- 6. <u>Non-Admission</u>. Nothing contained in this Agreement shall be deemed or construed as an admission of wrongdoing or liability on the part of the Company or the Executive.
- 7. <u>Governing Law and Forum.</u> The Executive and the Company agree that this Agreement and all matters or issues arising out of or relating to the Executive's employment with the Company shall be resolved pursuant to the provisions of Section 14 of the Employment Agreement.
 - 8. Notices. All notices or communications hereunder shall be in writing, addressed as follows:

To the Company;

First Western Financial, Inc. 1900 16th St Suite #1200 Denver, CO 80202 Attention: Secretary and General counsel

To the Executive:

Scott C. Wylie 435 Monroe Street Denver, CO 80206

All such notices shall be conclusively deemed to be received and shall be effective (i) if sent by hand delivery or nationally recognized courier, upon receipt or (ii) if sent by electronic mail or facsimile, upon receipt by the sender of such transmission.

THE EXECUTIVE ACKNOWLEDGES THAT HE HAS READ THIS AGREEMENT AND THAT HE FULLY KNOWS, UNDERSTANDS AND APPRECIATES ITS CONTENTS, AND THAT HE HEREBY EXECUTES THE SAME AND MAKES THIS AGREEMENT AND THE RELEASE AND AGREEMENTS PROVIDED FOR HEREIN VOLUNTARILY AND OF HIS OWN FREE WILL.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first set forth above.

First Western Financial, Inc.

By: <u>/s/ Julie Courkamp</u>
Julie Courkamp, Treasurer, COO, and CFO

THE EXECUTIVE

/s/ Scott C. Wylie Scott C. Wylie, Chairman, President, and CEO

Dated: April 26, 2023

[SIGNATURE PAGE TO GENERAL RELEASE OF CLAIMS]

SECOND AMENDED AND RESTATED EMPLOYMENT AGREEMENT

THIS AMENDED AND RESTATED EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") effective as of April 26, 2023 (the "<u>Effective Date</u>") is by and between First Western Financial Inc., a Colorado corporation (the "<u>Company</u>" or "<u>FWFI</u>"), and Julie Courkamp, an individual resident of the State of Colorado (the "<u>Executive</u>"). This Agreement amends, supersedes and replaces, in its entirety, the Amended & Restated Employment Agreement between the Company and the Executive dated effective as of March 5, 2018, as amended by that certain Amendment to Employment Agreement dated May 2, 2019, and that certain Second Amendment to Employment Agreement dated January 30, 2020 (collectively, the "<u>Prior Agreement</u>").

WHEREAS, the Executive is currently employed by the Company as its Chief Financial Officer, Chief Operating Officer and Treasurer; and

WHEREAS, the Executive and the Company would like to amend and restate the Prior Agreement to more closely align the Executive's severance compensation to current market practice.

NOW, THEREFORE, in consideration of the foregoing, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

- 1. <u>Employment and Duties</u>.
- (a) <u>General</u>. As of the Effective Date, the Executive shall continue to serve as the Chief Financial Officer, Chief Operating Officer and Treasurer of the Company. The Executive shall report directly to the Chief Executive Officer of the Company (the "<u>CEO</u>"). The Executive shall have such duties and responsibilities, commensurate with the Executive's position, as may be assigned to the Executive from time to time by the CEO. The Executive hereby accepts such employment and agrees to render the services described above.
- (b) Exclusive Services. For so long as the Executive is employed by the Company, the Executive shall devote her full-time working time to her duties hereunder, shall faithfully serve the Company, shall in all material respects conform to and comply with the lawful directions and instructions given to her by the CEO and shall use her reasonable best efforts to promote and serve the interests of the Company. Further, the Executive shall not, directly or indirectly, render services to any other person or organization without the consent of the CEO or otherwise engage in activities that would interfere in any material respect with her faithful performance of her duties hereunder. Notwithstanding the foregoing, (i) the Executive may serve on such other for-profit corporate boards as may be consented to by the CEO, provided that such activity does not contravene the first sentence of this Section 1(b), and (ii) the Executive may serve on not-for-profit corporate, civic or charitable boards or engage in charitable activities without remuneration therefor as may be consented to by the CEO, provided that such activity does not contravene the first sentence of this Section 1(b).
- 2. <u>Term.</u> The term of the Executive's employment under this Agreement commenced on the Effective Date, and shall expire on the December 31, 2023. This Agreement shall automatically renew for successive one-year terms commencing January 1, 2024, unless either party gives notice to the other 90 days before the end of a particular term. The period during which the Agreement is in effect shall be referred to as the "<u>Term</u>".

121986569.5

3. <u>Definitions</u>.

- (a) "Base Amount" means the amount equal to the sum of (i) the Executive's Base Salary at the rate then in effect and (ii) amount equal to the greater of (A) the Executive's target Annual Bonus at the rate then if effect or (B) the average Annual Bonus for the three fiscal years immediately preceding the date on which the Executive's termination occurs.
 - (b) Termination for "Cause" means termination of the Executive's employment because of:
 - (i) the willful failure by the Executive to perform the Executive's duties with the Company;
 - (ii) gross incompetence or gross negligence in the discharge of the Executive's duties;
 - (iii) willful dishonesty, theft, embezzlement, fraud, breach of confidentiality, or unauthorized disclosure or use of financial information, confidential client information, client or employee lists, trade secrets, or other Company confidential or proprietary information;
 - (iv) willful violation of any law, rule or regulation of any governing authority or of the Company's policies and procedures, including, without limitation, the Company's employee handbook or similar document;
 - (v) the willful refusal of Executive to follow the lawful directions of the CEO within a reasonable period after delivery to Executive of written notice of such directions;
 - (vi) willful conduct that is grossly injurious to the reputation, financial condition, business or assets of the Company; or
 - (vii) willful breach of any material provision in an agreement with the Company.

In each of (i) through (vii) the Executive shall be given written notice of such cause for termination, and in each of (i) and (vii) the Executive shall be given an opportunity to remedy such cause for termination within sixty (60) business days of receipt of such notice.

- (c) "<u>Change in Control</u>" shall have the same meaning as the definition contained in Section 12.2 of the First Western Financial, Inc. 2016 Omnibus Incentive Plan.
- (d) Resignation for "<u>Good Reason</u>" means termination of employment by the Executive because of the occurrence of any of the following events:
 - (i) there is a material reduction in the Executive's Base Salary, unless agreed to in writing by the Executive;
 - (ii) there is a material reduction in the Executive's authority, duties, or responsibilities, provided that the Company's appointment of a new Chief Financial Officer to replace the Executive in that position shall not be considered a material reduction in the Executive's authority, duties, or responsibilities;

- (iii) the failure of any successor to assume this Agreement; and
- (iv) any other action or inaction that constitutes a material breach by the Company of this Agreement after the Executive provides written notice to the Company of the facts which constitute the grounds within sixty (60) business days following the initial existence of the grounds and the Company thereafter fails to cure such grounds within sixty (60) business days following its receipt of such notice (or, in the event that such grounds cannot be corrected within such sixty (60) day period, the Company has not taken all reasonable steps within such sixty (60) day period to correct such grounds as promptly as practicable thereafter).
- 4. <u>Compensation and Other Benefits</u>. Subject to the provisions of this Agreement, the Company shall pay and provide the following compensation and other benefits to the Executive during the Term as compensation for services rendered hereunder:
- (a) <u>Base Salary</u>. The Company shall pay to the Executive an annual salary (the "<u>Base Salary</u>") payable in substantially equal installments at such intervals as may be determined by the Company in accordance with its ordinary payroll practices, as established from time to time. The Base Salary shall be reviewed by the Board of Directors annually, and increased as appropriate, for market changes. Commencing effective as of the Effective Date, Executive's annualized Base Salary is \$350,000. The Base Salary shall not be decreased by the Company except with the prior written consent of the Executive.
- (b) <u>Annual Bonus</u>. Executive is eligible to receive annual incentive compensation (the "<u>Annual Bonus</u>") as governed by the terms set forth in the First Western Financial, Inc. Incentive Plan for Senior Executive Officers, as may be amended from time to time. The incentive compensation performance measures and goals are reviewed by the Board of Directors annually, and adjusted as appropriate, according to the needs of the business.
- (c) <u>Long-Term Incentive Plan</u>. The Executive shall be eligible for grants under the First Western Financial, Inc. 2016 Omnibus Incentive Plan, including, but not limited to, grants of stock options, market conditioned performance share units, financial conditioned performance stock units and restricted stock units, as the Compensation Committee of the Board shall determine from time to time.
- (d) <u>Savings and Retirement Plans</u>. The Executive shall be eligible to participate in all savings and retirement plans applicable generally to other executives of the Company, in accordance with the terms of the plans, as may be amended from time to time.
- (e) <u>Welfare Benefit Plans</u>. The Executive and her eligible dependents shall be eligible to participate in and shall receive all benefits under the Company's welfare benefit plans and programs applicable generally to other executives of the Company, in accordance with the terms of the plans, as may be amended from time to time.
- (f) <u>Expenses</u>. Upon presentation of written documentation thereof, in accordance with the applicable expense reimbursement policies and procedures of the Company as in effect from time to time, the Company shall reimburse the Executive for reasonable business-related expenses incurred by the Executive in the fulfillment of her duties. Payments with respect to reimbursements of expenses shall be made promptly and in accordance with the applicable expense reimbursement policies and procedures of the Company, but in any event, on or before the last day of the calendar month following the calendar month in which the relevant expense is incurred.

- (g) <u>Vacation</u>. The Executive shall be entitled to vacation each calendar year during the Term, subject to and in accordance with the Company's vacation policy in effect from time to time.
- 5. <u>Termination of Employment</u>. The terms of any equity compensation grants outstanding as of the Effective Date, as well as any future equity compensation grants made under the First Western Financial, Inc. 2016 Omnibus Incentive Plan shall govern what the Executive receives on termination of employment, in terms of equity compensation, except as expressly provided in Section 5(b)(i)(C) and Section 5(b)(ii)(C) hereof. All other forms of remuneration the Executive is eligible for on termination of employment are set forth in this Section 5.
- (a) Termination for Cause; Resignation Without Good Reason. If, prior to the expiration of the Term, the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code"), and the Treasury Regulations ("Regulations") thereunder, by reason of the Company's termination of the Executive's employment for Cause, or if the Executive resigns from her employment hereunder other than for Good Reason, the Executive shall be entitled only to payment of (i) any unpaid Base Salary through and including the date of termination or resignation, (ii) any Annual Bonus earned, but unpaid, for the year immediately preceding the year in which the termination date occurs (which unpaid Annual Bonus amount shall be paid no later than March 15 of the year following the year in which the amount was earned), and (iii) any other amounts or benefits required to be paid or provided by law or under any plan, program, policy or practice of the Company (the amounts or benefits in (i) through (iii) being referred to collectively as the "Other Accrued Compensation and Benefits"). Except as set forth in this subsection (a), the Executive shall have no further right to receive any other compensation or benefits after such termination or resignation of employment.

(b) <u>Termination without Cause; Resignation for Good Reason.</u>

- (i) If the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Code and the Regulations thereunder, by reason of the Company's termination of the Executive's employment without Cause, or if the Executive resigns from her employment hereunder for Good Reason, the Executive shall be entitled to the following:
 - (A) The Other Accrued Compensation and Benefits.
- (B) An amount equal to one (1) times the Executive's Base Amount, which shall be payable in equal installments pursuant to the Company's normal payroll practices and subject to all legally required and customary withholdings for the twelve (12) month period following termination.
- (C) A pro-rated portion of each outstanding unvested equity award, including, without limitation, stock options, restricted stock and restricted stock units, held by the Executive shall automatically become vested and, if applicable, exercisable and any restrictions thereon shall immediately lapse. The pro-rated portion, determined separately for each vesting tranche for each such equity award, shall be based on a fraction, (I) the numerator of which is number of days that elapse from the grant date of the equity award until the Executive's termination date and the (II) the denominator of which is the total number of days in the period that begins on the date of grant of the applicable equity award and ends on the scheduled vesting date of such equity award (or, if applicable, the scheduled vesting date of the applicable vesting tranche). For purposes of determining the number of shares subject to any outstanding unvested equity awards

subject to performance vesting conditions (as opposed to solely service-based vesting conditions) that would otherwise vest pursuant to the foregoing sentence, the applicable performance goals shall be deemed achieved at the level of the Company's actual achievement of the applicable performance goals as measured as of the Executive's termination date, as determined by the Compensation Committee in its reasonable discretion. Each of the Executive's outstanding equity awards as of the Effective Date is hereby amended to incorporate the terms of this Section 5.1(b)(i)(C).

(D) The Company shall monthly pay to the Executive the amount equal to the full premium amount (determined as of the date of termination) for continued coverage under the Company's health plan pursuant to the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), for the Executive, and, to the extent that the Executive is providing coverage for her spouse or eligible dependents as of the termination date, for such individuals; provided, however, that the Company's obligation to pay such premiums shall cease immediately upon the earlier of (i) the passage of eighteen (18) months (ii) the expiration of the statutory COBRA period and (iii) the date the Executive becomes eligible for coverage under any other group health plan (as an employee or otherwise) or Medicare.

Notwithstanding the foregoing, if the Company terminates the Executive's employment without Cause, the Company shall provide the Executive with no less than ninety (90) days' written notice or payment of three (3) months Base Salary in lieu of ninety (90) days' written notice, which shall be in addition to payments described under this Section 5(b)(i).

- (E) All payments and benefits provided under this Section 5(b)(i)(B) and (D) shall commence on the first payroll date following the 60th day after the Executive's termination of employment, with the first installment payment to include any payments that would have otherwise been paid to the Executive if such payments commenced on the first payroll date following the Executive's termination date. The Company shall not be required to make the payments and provide the benefits provided for under this Section 5(b)(i)(A), (B), (C) or (D) unless the Executive executes and delivers to the Company, within sixty (60) days following the Executive's termination of employment, a release substantially in the form attached hereto as Exhibit A (the "Release"), and the Release has become effective and irrevocable in its entirety in such 60-day period. The Executive's failure or refusal to sign the Release (or the Executive's revocation of such Release in accordance with applicable laws) will result in the forfeiture of the payments and benefits under this Section 5(b)(i)(A), (B), (C) or (D). To the extent any amount payable under this Section 5 is deferred compensation subject to the Code, if the period during which the Executive has discretion to execute or revoke the Release straddles two of the Executive's taxable years, then the Company shall make the severance payments starting in the second of such taxable years, regardless of which taxable year the Executive actually delivers the executed Release to the Company. The Executive may not, directly or indirectly, designate the calendar year or timing of payments. This Section 5(b)(i) shall expressly not apply to payments made on account of a Change in Control, pursuant to Section 5(b)(ii) hereof.
- (F) If, following a termination of employment without Cause or a resignation for Good Reason, the Executive breaches the provisions of Sections 6 through 10 hereof or breaches any provision set forth in the executed copy of the general release of claims, the Executive shall not be eligible, as of the date of such breach, for the payments and benefits described in Section 5(b)(i)(A), (B), (C) or (D), and any and all obligations and agreements of the Company with respect to such payments shall

thereupon cease. This Section 5(b)(i)(F) shall expressly not apply to payments made on account of a Change in Control, pursuant to Section 5(b)(ii) hereof.

- (ii) If the Company undergoes a Change in Control and, within 24 months of such Change in Control, the Executive is terminated without Cause or resigns for Good Reason, then, in lieu of (and not in addition to) any payments or benefits payable to the Executive pursuant to Section 5(b)(i), the Executive shall be entitled to the following:
 - (A) The Other Accrued Compensation and Benefits.
- (B) An amount equal to two (2) times the Executive's Base Amount, payable in a single lump sum payment.
- (C) Each outstanding equity award, including, without limitation, stock options, restricted stock and restricted stock units, held by the Executive shall automatically become vested and, if applicable, exercisable and any restrictions thereon shall immediately lapse, in each case, with respect to that number of shares of Company common stock that would otherwise vest based on Executive's continued employment. For purposes of determining the number of shares subject to any outstanding equity awards subject to performance vesting conditions that would otherwise vest pursuant to the foregoing sentence, the applicable performance goals shall be deemed achieved at the "target" level. Each of the Executive's outstanding equity awards as of the Effective Date is hereby amended to incorporate the terms of this Section 5.1(b)(ii)(C).
- (D) Monthly payments to the Executive equal to the full premium amount (determined as of the date of termination) for continued coverage under the Company's health plan pursuant to COBRA for the Executive, and, to the extent that the Executive is providing coverage for her spouse or eligible dependents as of the termination date, for such individuals; provided, however, that the Company's obligation to pay such premiums shall cease immediately upon the earlier of (i) the passage of eighteen (18) months (ii) the expiration of the statutory COBRA period and (iii) the date the Executive becomes eligible for coverage under any other group health plan (as an employee or otherwise) or Medicare.

Notwithstanding the foregoing, if the Company (or its successor) terminates the Executive's employment without Cause, the Company (or its successor) shall provide the Executive with no less than ninety (90) days' written notice or payment of three (3) months Base Salary in lieu of ninety (90) days' written notice, which shall be in addition to payments described under this Section 5(b)(ii).

(E) Unless otherwise provided herein, all payments and benefits provided under this Section 5(b)(ii)(B) and (D) shall be paid (or, in the case of the payments described in Section 5(b)(ii)(D), commence to be paid) on the first payroll date following the 60th day after the Executive's termination of employment, with the first payment to include any payments provided under Section 5(b)(ii)(D) that would have otherwise been paid to the Executive if such payments commenced on the first payroll date following the Executive's termination date. The Company shall not be required to make the payments and provide the benefits provided for under this Section 5(b)(ii)(A), (B), (C) or (D) unless the Executive executes and delivers the Release to the Company, within sixty (60) days following the Executive's termination of employment, and the Release has become effective and irrevocable in its entirety in such 60-day period. The Executive's failure or refusal to sign the Release (or the Executive's

revocation of such Release in accordance with applicable laws) will result in the forfeiture of the payments and benefits under this Section 5(b)(ii)(A), (B), (C) or (D). To the extent any amount payable under this Section 5 is deferred compensation subject to the Code, if the period during which the Executive has discretion to execute or revoke the Release straddles two of the Executive's taxable years, then the Company shall make the severance payments starting in the second of such taxable years, regardless of which taxable year the Executive actually delivers the executed Release to the Company. The Executive may not, directly or indirectly, designate the calendar year or timing of payments.

- (F) In the event that it is determined that any payment or distribution of any type to or for the benefit of an Executive made by the Company, by any of its Affiliates, by any person who acquires ownership or effective control or ownership of a substantial portion of the Company's assets (within the meaning of Code Section 280G) or by any affiliate of such person, whether paid or payable or distributed or distributable pursuant to the terms of any equity compensation plan, this Agreement or otherwise (the "Total Payments"), would be subject to the excise tax imposed by Code Section 4999 or any interest or penalties with respect to such excise tax (the "Excise Tax"), then, notwithstanding any other provision of this Agreement or any equity compensation plan to the contrary, any right of the Executive to any payment or benefit under this Agreement or any such equity compensation plan shall be reduced or eliminated, but only to the extent necessary to avoid imposition of the Excise Tax. In no case, however, shall such cutback be made if Total Payments after the imposition of the Excise Tax are greater than Total Payments cut back as provided in this Section 5(b) (ii) to avoid the Excise Tax.
- (G) In the event that a cutback of Total Payments is permitted under Section 5(b)(ii)(F), and except as required by Code Section 409A or to the extent that Code 409A permits discretion, the Compensation Committee shall have the right, in the Compensation Committee's sole discretion, to designate those rights, payments, or benefits and all other agreements that should be reduced or eliminated so as to provide the Executive with the maximum pre-tax amount which avoids imposition of the Excise Tax. For example, the Compensation Committee may choose to cut back cash severance, if that would yield a higher pre-tax amount than cutting back equity. Notwithstanding the foregoing, to the extent any payment or benefit constitutes deferred compensation under Code Section 409A, in order to comply with Code Section 409A the Compensation Committee shall instead accomplish such reduction by first reducing or eliminating any cash payments (with the payments to be made furthest in the future being reduced first), then by reducing or eliminating any accelerated vesting of options or stock appreciation rights, then by reducing or eliminating any accelerated vesting of restricted stock or restricted stock units.
- (c) <u>Termination Due to Death or Disability</u>. The Executive's employment with the Company shall terminate automatically on the Executive's death. In the event of the Executive's Disability (as defined herein), the Company shall be entitled to terminate her employment. In the event of the Executive's death or if the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Code, or the Regulations thereunder, by reason of the Executive's Disability, the Company shall pay to the Executive (or her estate, as applicable), (i) the Executive's Base Salary through and including the date of termination and any Other Accrued Compensation and Benefits (ii) a pro-rata Annual Bonus for the year of termination, based on actual audited year-end results and payable when bonuses are normally paid to employees, and (iii) three (3) months Base Salary at the rate then in effect, payable in equal installments pursuant to the Company's normal payroll practices and subject to all legally required and customary withholdings for the three (3) month following termination.

- (i) For purposes of this Agreement, "<u>Disability</u>" means a physical or mental disability or infirmity of the Executive that prevents the normal performance of substantially all her duties for a period in excess of ninety (90) consecutive days or for more than ninety (90) days in any consecutive twelve (12)-month period. Evidence of such physical or mental disability or infirmity shall be certified by a physician licensed to practice in the state of residence of the Executive, which physician is mutually agreeable to the Board and the Executive. If there is no agreement on the selection of the physician, then the Board shall select one physician and the Executive shall select one physician, and the two physicians shall attempt to mutually agree upon such physical or mental disability or infirmity. If the two physicians cannot agree, then the two physicians shall jointly select a third physician, whose opinion on such physical or mental disability or infirmity shall control.
- (d) <u>Notice of Termination</u>. Any termination of employment by the Company or the Executive shall be communicated by a written "<u>Notice of Termination</u>" to the other party hereto given in accordance with Section 24 of this Agreement. In the event of a termination by the Company for Cause, or by the Executive for Good Reason, the Notice of Termination shall (i) indicate the specific termination provision in this Agreement relied upon, (ii) set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of the Executive's employment under the provision so indicated and (iii) specify the date of termination, which date shall not be more than thirty (30) business days after the giving of such notice, provided that the date of termination will not occur before the expiration of any applicable cure period.

The failure by the Executive or the Company to set forth in the Notice of Termination any fact or circumstance which contributes to a showing of Good Reason or Cause shall not waive any right of the Executive or the Company, respectively, hereunder or preclude the Executive or the Company, respectively, from asserting such fact or circumstance in enforcing the Executive's or the Company's rights hereunder.

- (e) <u>No Further Rights</u>. The Executive shall have no further rights under this Agreement or otherwise to receive any other compensation or benefits after such termination or resignation of employment.
 - 6. <u>Confidentiality</u>.
 - (a) <u>Confidential Information</u>.
 - (i) The Executive agrees that she will not at any time, except with the prior written consent of the Company or its Affiliates or as required by applicable law, directly or indirectly, reveal to any person, entity or other organization (other than the Company or its Affiliates or its respective employees, officers, directors, shareholders or agents) or use for the Executive's own benefit any confidential or proprietary information of any member of the Company or its Affiliates ("Confidential Information") relating to the assets, liabilities, employees, goodwill, business or affairs of any member of the Company or its Affiliates, including, without limitation, any information concerning past, present or prospective clients, intellectual capital, marketing data, or other confidential information used by, or useful to, any member of and known to the Executive by reason of the Executive's employment by, shareholdings in or other association with the Company or its Affiliates, other than disclosure while employed by the Company which the Executive reasonably and in good faith believes to be in or not opposed to the interests of the Company; provided that such Confidential Information does not include any information which is available to the general public or is generally available within the relevant business or industry other than as a result of the Executive's breach of this

Agreement. Confidential Information may be in any medium or form, including, without limitation, physical documents, computer files or disks, videotapes, audiotapes, and oral communications.

- (ii) In the event that the Executive becomes legally compelled to disclose any Confidential Information, the Executive shall, if permitted by law, provide the Company with prompt written notice so that the Company may seek a protective order or other appropriate remedy. In the event that such protective order or other remedy is not obtained, the Executive shall furnish only that portion of such Confidential Information or take only such action as is legally required by binding order and shall exercise her reasonable efforts to obtain reliable assurance that confidential treatment shall be accorded any such Confidential Information. The Company shall promptly pay (upon receipt of invoices and any other documentation as may be requested by the Company) all reasonable expenses and fees incurred by the Executive, including attorneys' fees, in connection with her compliance with the immediately preceding sentence. Notwithstanding anything herein to the contrary, nothing in this Agreement shall (A) prohibit the Executive from making reports of possible violations of federal law or regulation to any governmental agency or entity in accordance with the provisions of and rules promulgated under Section 21F of the Securities Exchange Act of 1934 or Section 806 of the Sarbanes-Oxley Act of 2002, or of any other whistleblower protection provisions of state or federal law or regulation, or (B) require notification or prior approval by the Company of any reporting described in clause (A).
- (b) <u>Confidentiality of Agreement</u>. The Executive agrees that, except as may be required by applicable law or legal process, during the Term and thereafter, she shall not disclose the terms of this Agreement to any person or entity other than the Executive's accountants, financial advisors, attorneys and spouse agree not to disclose the terms of this Agreement to any other person or entity.
- (c) <u>Exclusive Property</u>. The Executive confirms that all Confidential Information is and shall remain the exclusive property of the Company or its Affiliates. All business records, papers and documents kept or made by the Executive relating to the business of the Company or its Affiliates shall be and remain the property of the Company or its Affiliates. Upon the request and at the expense of the Company or its Affiliates, the Executive shall promptly make all disclosures, execute all instruments and papers and perform all acts reasonably necessary to vest and confirm in the Company or its Affiliates, fully and completely, all rights created or contemplated by this Section 6.
- 7. Noncompetition. The Executive agrees that, for a period commencing on the January 1, 2017 and ending 365 days following the Executive's termination of employment (the "Restricted Period"), the Executive shall not, without the prior written consent of the Company, directly or indirectly, and whether as principal or investor or as an employee, officer, director, manager, partner, consultant, agent or otherwise, alone or in association with any other person, firm, corporation or other business organization, carry on a Competing Business (as defined herein) within the County of Denver and contiguous counties. For purposes of this Section 7: (a) carrying on a "Competing Business" means to engage in the competing business of any business carried on by the Company or its Affiliates. Notwithstanding the foregoing nothing herein shall limit the Executive's right to own not more than one (1) percent of any of the debt or equity securities of any business organization that is then filing reports with the Securities and Exchange Commission pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended. The Executive acknowledges that this noncompetition provision will not unreasonably impair or infringe upon the Executive's right to work or earn a living during the Restricted Period, and the Executive further acknowledges that these restrictions are necessary to

protect the trade secrets of the Company as the Executive's breach of this provision would necessarily involve the Executive's use of the Company's trade secrets.

Initial

8. Non-Solicitation. The Executive agrees that for the Restricted Period the Executive shall not, directly or indirectly, (a) interfere with or attempt to interfere with the relationship between any person who is, or was during the Restricted Period or the 3-month period immediately preceding the commencement of the Restricted Period, an employee, officer, representative or agent of the Company or its Affiliates and any member of the Company or its Affiliates, or solicit, induce or attempt to solicit or induce any of them to terminate their employment or service relationship with any member of the Company or its Affiliates or violate the terms of their respective service contracts, or any employment arrangements, with such entities, provided that the foregoing shall not prevent general employment or service solicitations that do not specifically target any such persons; or (b) induce or attempt to induce any customer or client of any member of the Company or its Affiliates to cease doing business with any member of the Company or its Affiliates, or in any way interfere with the relationship between any member of the Company or its Affiliates. The Executive acknowledges that this non-solicitation provision will not unreasonably impair or infringe upon the Executive's right to work or earn a living during the Restricted Period, and the Executive further acknowledges that these restrictions are necessary to protect the trade secrets of the Company as the Executive's breach of this provision would necessarily involve the Executive's use of the Company's trade secrets.

Initial

- 9. <u>No Conflicting Agreement</u>. The Executive represents, warrants and covenants to the Company that the Executive is not a party to any agreement, whether written or oral, that would be breached by or would prevent or interfere with the executive of the Executive of this Agreement or the fulfillment by the Executive of the Executive's obligations hereunder.
- 10. <u>Nondisparagement</u>. Each party represents, warrants and covenants to the other that at no time during the Term or thereafter shall such party make, or cause or assist any other person to make, any statement or other communication to any third party which impugns or attacks, or is otherwise critical of, the reputation, business or character of the other party or any of its respective directors, officers or employees, as applicable; provided this Section shall not prohibit truthful testimony by or on behalf of either party in any judicial or administrative proceeding.
- 11. Section 409A of the Code. This Agreement is intended to meet the requirements of Section 409A of the Code and shall be interpreted and construed consistent with that intent. Notwithstanding any other provision of this Agreement, to the extent that the right to any payment (including the provision of benefits) hereunder provides for the "deferral of compensation" within the meaning of Section 409A(d)(l) of the Code, if the Executive is a "Specified Employee" within the meaning of Section 409A(a)(2)(B)(i) of the Code on the date of the Executive's "Separation from Service" within the meaning of Section 409A(a)(2)(A)(i) of the Code (the "Separation Date"), then no such payment shall be made or commence during the period beginning on the Separation Date and ending on the date that is six months following the Separation Date or, if earlier, on the date of the Executive's death. The amount of any payment that would otherwise be paid to the Executive during this period shall instead be paid to the Executive on the fifteenth day of the first calendar month following the end of the period.
 - 12. Certain Remedies.

- (a) <u>Forfeiture/Payment Obligations</u>. In the event the Executive fails to comply with Sections 6 through 10, other than any isolated, insubstantial and inadvertent failure, the Executive agrees that she will forfeit any amounts not already paid pursuant to Section 5(b)(i) (A), (B), (C) or (D) of this Agreement. Notwithstanding the previous sentence, the Executive shall be given written notice of each alleged failure to comply with Sections 6 through 10, and the Executive shall be given an opportunity to remedy such failure within (60) sixty business days of the receipt of such notice. For purposes of clarity, the Executive's failure to comply with Sections 6 through 10 shall not result in forfeiture of amounts required to be paid but not already paid on account of a Change in Control pursuant to Section 5(b)(ii) of this Agreement.
- (b) <u>Injunctive Relief.</u> Without intending to limit the remedies available to the Company or its Affiliates, including, but not limited to, that set forth in Section 12(a) hereof, the Executive agrees that a breach of any of the covenants contained in Sections 6 through 10 of this Agreement may result in material and irreparable injury to the Company or its Affiliates for which there is no adequate remedy at law, that it will not be possible to measure damages for such injuries precisely and that, in the event of such a breach or threat thereof, any member of the Company or its Affiliates shall be entitled to seek a temporary restraining order or a preliminary or permanent injunction, or both, without bond or other security, restraining the Executive from engaging in activities prohibited by the covenants contained in Sections 6 through 10 of this Agreement or such other relief as may be required specifically to enforce any of the covenants contained in this Agreement. Such injunctive relief in any court shall be available to the Company or its Affiliates in lieu of, or prior to or pending determination in, any arbitration proceeding.
- 13. <u>Defense of Claims</u>. The Executive agrees that, during the Term, and for a period of seven (7) years after termination of the Executive's employment, upon request from the Company, the Executive will reasonably cooperate with the Company in the defense of any claims or actions that may be made by or against the Company that affect the Executive's prior areas of responsibility, except if the Executive's reasonable interests are adverse to the Company in such claim or action. The Company agrees to promptly pay in advance or reimburse the Executive for, as requested by the Executive, all of the Executive's reasonable travel and other direct costs and expenses incurred, or to be reasonably incurred, to comply with the Executive's obligations under this Section 13, including, but not limited to, legal costs and expenses.
- 14. Alternative Dispute Resolution. The Company and the Executive agree that any dispute that arises out of or relates to Executive's employment or termination of employment with the Company, including any dispute that the Executive may have with any present or former officer, manager, director, employee, agent, attorney or insurer of the Company, shall first be submitted to mediation through the Institute for Conflict Prevention & Resolution ("CPR") (or such other nationally-recognized alternative dispute resolution service as the Executive and Company may agree). The Executive and the Company shall use their reasonable efforts to commence and conclude such mediation in a prompt manner. If the dispute is not resolved through mediation within thirty (30) days after notice thereof, such dispute shall be resolved by binding arbitration in accordance with the rules and procedures of the CPR (or such other nationally recognized alternative dispute resolution service as the Executive and the Company may agree). Judgment upon the award rendered by the arbitrator may be entered in any court having in person and subject matter jurisdiction. The Company and the Executive hereby submit to the jurisdiction of the federal and state courts in Denver, Colorado, for the purpose of confirming any such award and entering judgment thereon. The Company shall pay for all administrative costs and fees charged by the CPR (or such other nationally recognized alternative dispute resolution service) as well as the fees charged by the arbitrator. Each party shall pay for her or its attorneys' fees and costs.
 - 15. <u>Nonassignability; Binding Agreement.</u>

11

- (a) <u>By the Executive</u>. This Agreement and any and all rights, duties, obligations or interests hereunder shall not be assignable or delegable by the Executive.
- (b) <u>By the Company</u>. This Agreement and all of the Company's rights and obligations hereunder shall not be assignable by the Company except as incident to a reorganization, merger or consolidation, or transfer of all or substantially all of the Company's assets.
- (c) <u>Binding Effect</u>. This Agreement shall be binding upon, and inure to the benefit of, the parties hereto, any successors to or assigns of the Company and the Executive's heirs and the personal representatives of the Executive's estate.
- 16. <u>Withholding</u>. Any payments made or benefits provided to the Executive under this Agreement shall be reduced by any applicable withholding taxes or other amounts required to be withheld by law or contract.
- 17. <u>Amendment; Waiver</u>. This Agreement may not be modified, amended or waived in any manner, except by an instrument in writing signed by both parties hereto. The waiver by either party of compliance with any provision of this Agreement by the other party shall not operate or be construed as a waiver of any other provision of this Agreement, or of any subsequent breach by such party of a provision of this Agreement.
- 18. <u>Governing Law and Forum</u>. The Executive and the Company agree that this Agreement and all matters or issues arising out of or relating to the Executive's employment with the Company shall be governed by the laws of the State of Colorado applicable to contracts entered into and performed entirely therein. Any action to enforce this Agreement shall be brought solely in the state or federal courts located in the County of Denver, Colorado.
- 19. <u>Survival of Certain Provisions</u>. Unless expressly provided otherwise, the rights and obligations set forth in this Agreement shall survive any termination or expiration of this Agreement.
- 20. <u>Entire Agreement; Supersedes Previous Agreements</u>. This Agreement contains the entire agreement and understanding of the parties hereto with respect to the matters covered herein, and supersedes all prior or contemporaneous negotiations, commitments, agreements and writings with respect to the subject matter hereof (including, without limitation, the Prior Agreement), all such other negotiations, commitments, agreements and writings shall have no further force or effect, and the parties to any such other negotiation, commitment, agreement or writing shall have no further rights or obligations thereunder.
- 21. <u>Severability</u>. The provisions of the Agreement shall be deemed severable, and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions herein.
- 22. <u>Counterparts</u>. This Agreement may be executed by either of the parties hereto in counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same instrument.
- 23. <u>Headings</u>. The headings of sections herein are included solely for convenience of reference and shall not control the meaning or interpretation of any of the provisions of this Agreement.
 - 24. Notices. All notices or communications hereunder shall be in writing, addressed as follows:

To the Company:

First Western Financial, Inc. 1900 16th St Suite #1200 Denver, CO 80202 Attention: Secretary and General counsel

To the Executive:

Julie Courkamp 14730 Verbena Court Brighton, Colorado 80602

All such notices shall be conclusively deemed to be received and shall be effective (i) if sent by hand delivery or nationally recognized courier, upon receipt or (ii) if sent by electronic mail or facsimile, upon receipt by the sender of such transmission.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

IN WITNESS WHEREOF, the Company has caused this Agreement to be signed by its officer pursuant to the authority of its Board, and the Executive has executed this Agreement, as of the day and year first written above.

| FIRST WESTERN FINANCIAL, IN | C. |
|-----------------------------|----|
| By: | |
| JULIE COURKAMP | |
| _ | |

GENERAL RELEASE OF CLAIMS

| This General Release of all | Claims (this "Agreement") is entered into by Julie Courkamp (the "Executive") and |
|---------------------------------------|---|
| FWFI (the "Company"), effective as of | in connection with the termination of the Executive's employment with the |
| Company as of | |
| | |

In consideration of the promises set forth in the Second Amended and Restated Employment Agreement between the Executive and the Company, dated effective April 26, 2023 (the "<u>Employment Agreement</u>"), the Executive and the Company agree as follows:

1. <u>Return of Property.</u> All Company files, access keys and codes, desk keys, ID badges, computers, records, manuals, electronic devices, computer programs, papers, electronically stored information or documents, telephones and credit cards, and any other property of the Company in the Executive's possession must be returned no later than the date of the Executive's termination from the Company; provided, that, after the notification of a consultation with the Company, the Executive may keep one copy of such items as she may reasonably expect to use to protect her rights under this Agreement.

2. General Release and Waiver of Claims.

- (a) Release. In consideration of the payments and benefits provided to the Executive under the Employment Agreement and after consultation with counsel, the Executive and each of the Executive's respective heirs, executors, administrators, representatives, agents, insurers, successors and assigns (collectively, the "Releasors") hereby irrevocably and unconditionally release and forever discharge the Company, its subsidiaries and affiliates and each of their respective officers, employees, directors, shareholders and agents ("Releasees") from any and all claims, actions, causes of action, rights, judgments, obligations, damages, demands, accountings or liabilities of whatever kind or character (collectively, "Claims"), including, without limitation, any Claims under any federal, state, local or foreign law, that the Releasors may have, or in the future may possess, arising out of (i) the Executive's employment relationship with and service as an employee, officer or director of the Company or any subsidiaries or affiliated companies and the termination of such relationship or service, and any event, condition, circumstance or obligation that occurred, existed or arose on or prior to the date hereof; provided, however, that the Executive does not release, discharge or waive any rights to (i) payments and benefits provided under the Employment Agreement that are contingent upon the execution by the Executive of this Agreement and (ii) any indemnification rights the Executive may have under the Employment Agreement, in accordance with the Company's governance instruments or under any director and officer liability insurance maintained by the Company with respect to liabilities arising as a result of the Executive's service as an officer and employee of the Company. This Section 2(a) does not apply to any Claims that the Releasors may have as of the date the applicable rules and regulations promulgated thereunder ("ADEA"). Claims arising under ADEA are addressed in Section 2(b) of this Agreement.
- (b) <u>Specific Release of ADEA Claims</u>. In further consideration of the payments and benefits provided to the Executive under the Employment Agreement, the Releasors hereby unconditionally release and forever discharge the Releasees from any and all Claims arising under ADEA that the Releasors may have as of the date the Executive signs this Agreement. By signing this Agreement, the Executive hereby acknowledges and confirms the following: (i) the Executive was advised by the Company in connection with her termination to

consult with an attorney of her choice prior to signing this Agreement: and to have such attorney explain to the Executive the terms of this Agreement, including, without limitation, the terms relating to the Executive's release of claims arising under ADEA, and the Executive has in fact consulted with an attorney; (ii) the Executive was given a period of not fewer than 21 days to consider the terms of this Agreement and to consult with an attorney of her choosing with respect thereto; (iii) the Executive knowingly and voluntarily accepts the terms of this Agreement; and (iv) the Executive is providing this release and discharge only in exchange for consideration in addition to anything of value to which the Executive is already entitled. The Executive also understands that she has seven days following the date on which she signs this Agreement within which to revoke the release contained in this Section, by providing the Company with a written notice of her revocation of the release and waiver contained in this Section.

(c) <u>No Assignment</u>. The Executive represents and warrants that she has not assigned any of the Claims being released under this Agreement. The Company may assign this Agreement, in whole or in part, to any affiliated company or subsidiary of, or any successor in interest to, the Company.

3. Proceedings.

- (a) <u>General Agreement Relating to Proceedings</u>. The Executive has not filed, and except as provided in <u>Sections 3(b)</u> and $\underline{3(c)}$, the Executive agrees not to initiate or cause to be initiated on her behalf, any complaint, charge, claim or proceeding against the Releasees before any local, state or federal agency, court or other body relating to her employment or the termination of her employment, other than with respect to the obligations of the Company to the Executive under the Employment Agreement (each, individually, a "<u>Proceeding</u>"), and agrees not to participate voluntarily in any Proceeding. The Executive waives any right she may have to benefit in any manner from any relief (whether monetary or otherwise) arising out of any Proceeding.
- (b) <u>Proceedings Under ADEA</u>. <u>Section 3(a)</u> shall not preclude the Executive from filing any complaint, charge, claim or proceeding challenging the validity of the Executive's waiver of Claims arising under ADEA (which is set forth in <u>Section 2(b)</u> of this Agreement). However, both the Executive and the Company confirm their belief that the Executive's waiver of claims under ADEA is valid and enforceable, and that their intention is that all claims under ADEA will be waived.
- (c) <u>Certain Administrative Proceedings</u>. In addition, <u>Section 3(a)</u> shall not preclude the Executive from filing a charge with or participating in any administrative investigation or proceeding by the Equal Employment Opportunity Commission or another Fair Employment Practices agency. The Executive is, however, waiving her right to recover money in connection with any such charge or investigation. The Executive is also waiving her right to recover money in connection with a charge filed by any other entity or individual, or by any federal, state or local agency.
- 4. Remedies. In the event the Executive initiates or voluntarily participates in any Proceeding in violation of this Agreement, or if she fails to abide by any of the terms of this Agreement or her post-termination obligations contained in the Employment Agreement, or if she revokes the ADEA release contained in Section 2(b) within the seven-day period provided under Section 2(b), the Company may, in addition to any other remedies it may have, reclaim any amounts paid to her under Sections 5(b)(i)(A), (B), (C) or (D) of the Employment Agreement or terminate any benefits or payments that are subsequently due under Sections 5(b)(i)(A), (B), (C) and (D) of the Employment Agreement, without waiving the release granted herein. The Executive acknowledges and agrees that the remedy at law available to the Company for breach of any of her post-termination obligations under the Employment

Agreement or her obligations under $\underline{\text{Sections 2}}$ and $\underline{\text{3}}$ herein would be inadequate and that damages flowing from such a breach may not readily be susceptible to measurement in monetary terms. Accordingly, the Executive acknowledges, consents and agrees that, in addition to any other tights or remedies that the Company may have at law or in equity or as may otherwise be set forth in the Employment Agreement, the Company shall be entitled to seek a temporary restraining order or a preliminary or permanent injunction, or both, without bond or other security, restraining the Executive from breaching her post-termination obligations under the Employment Agreement or her obligations under $\underline{\text{Sections 2}}$ and $\underline{\text{3}}$ herein. Such injunctive relief in any court shall be available to the Company, in lieu of, or prior to or pending determination in, any arbitration proceeding.

The Executive understands that by entering into this Agreement she shall be limiting the availability of certain remedies that she may have against the Company and limiting also her ability to pursue certain claims against the Company.

- 5. <u>Severability Clause</u>. In the event that any provision or part of this Agreement is found to be invalid or unenforceable, only that particular provision or part so found, and not the entire Agreement, shall be inoperative.
- 6. <u>Non-Admission</u>. Nothing contained in this Agreement shall be deemed or construed as an admission of wrongdoing or liability on the part of the Company or the Executive.
- 7. <u>Governing Law and Forum</u>. The Executive and the Company agree that this Agreement and all matters or issues arising out of or relating to the Executive's employment with the Company shall be resolved pursuant to the provisions of Section 14 of the Employment Agreement.
 - 8. Notices. All notices or communications hereunder shall be in writing, addressed as follows:

To the Company:

First Western Financial, Inc. 1900 16th St Suite #1200 Denver, CO 80202 Attention: Secretary

To the Executive: Julie Courkamp 14730 Verbena Court Brighton, Colorado 80602

All such notices shall be conclusively deemed to be received and shall be effective (i) if sent by hand delivery or nationally recognized courier, upon receipt or (ii) if sent by electronic mail or facsimile, upon receipt by the sender of such transmission.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

THE EXECUTIVE ACKNOWLEDGES THAT SHE HAS READ THIS AGREEMENT AND THAT SHE FULLY KNOWS, UNDERSTANDS AND APPRECIATES ITS CONTENTS, AND THAT SHE HEREBY EXECUTES THE SAME AND MAKES THIS AGREEMENT AND THE RELEASE AND AGREEMENTS PROVIDED FOR HEREIN VOLUNTARILY AND OF HER OWN FREE WILL.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first set forth above.

First Western Financial, Inc.

By: <u>/s/ Scott C. Wylie</u> Scott C. Wylie, Chairman, President, and CEO

THE EXECUTIVE

/s/ Julie Courkamp Julie Courkamp, Treasurer, COO, and CFO

Dated: April 26, 2023

AMENDED AND RESTATED EMPLOYMENT AGREEMENT

THIS AMENDED AND RESTATED EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") effective as of April 26, 2023 (the "<u>Effective Date</u>") is by and between First Western Financial Inc., a Colorado corporation (the "<u>Company</u>" or "<u>FWFI</u>"), and John Sawyer, an individual resident of the State of Colorado (the "<u>Executive</u>"). This Agreement amends, supersedes and replaces, in its entirety, the Employment Agreement between the Company and the Executive dated effective as of April 8, 2020 (the "<u>Prior Agreement</u>").

WHEREAS, the Executive is currently employed by the Company as its Chief Investment Officer; and

WHEREAS, the Executive and the Company would like to amend and restate the Prior Agreement to more closely align the Executive's severance compensation to current market practice.

NOW, THEREFORE, in consideration of the foregoing, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

- 1. <u>Employment and Duties</u>.
- (a) <u>General</u>. As of the Effective Date, the Executive shall continue to serve as the Chief Investment Officer of the Company. The Executive shall report directly to the Chief Executive Officer of the Company (the "<u>CEO</u>"). The Executive shall have such duties and responsibilities, commensurate with the Executive's position, as may be assigned to the Executive from time to time by the CEO. The Executive hereby accepts such employment and agrees to render the services described above.
- (b) Exclusive Services. For so long as the Executive is employed by the Company, the Executive shall devote his full-time working time to his duties hereunder, shall faithfully serve the Company, shall in all material respects conform to and comply with the lawful directions and instructions given to his by the CEO and shall use his reasonable best efforts to promote and serve the interests of the Company. Further, the Executive shall not, directly or indirectly, render services to any other person or organization without the written consent of the CEO or otherwise engage in activities that would interfere in any material respect with his faithful performance of his duties hereunder. Notwithstanding the foregoing, (i) the Executive may serve on such other for-profit corporate boards as may be consented to by the CEO, provided that such activity does not contravene the first sentence of this Section 1(b), and (ii) the Executive may serve on not-for-profit corporate, civic or charitable boards or engage in charitable activities without remuneration therefor as may be consented to by the CEO, provided that such activity does not contravene the first sentence of this Section 1(b).
- 2. <u>Term.</u> The term of the Executive's employment under this Agreement commenced on the Effective Date, and shall expire on the December 31, 2023. This Agreement shall automatically renew for successive one-year terms commencing January 1, 2024, unless either party gives notice to the other 90 days before the end of a particular term. The period during which the Agreement is in effect shall be referred to as the "<u>Term</u>".
 - 3. <u>Definitions</u>.
- (a) "Base Amount" means the amount equal to the sum of (i) the Executive's Base Salary at the rate then in effect and (ii) amount equal to the greater of (A) the Executive's

121984509.4

target Annual Bonus at the rate then if effect or (B) the average Annual Bonus for the three fiscal years immediately preceding the date on which the Executive's termination occurs.

- (b) Termination for "<u>Cause</u>" means termination of the Executive's employment because of:
 - (i) the willful failure by the Executive to perform the Executive's duties with the Company;
 - (ii) gross incompetence or gross negligence in the discharge of the Executive's duties;
- (iii) willful dishonesty, theft, embezzlement, fraud, breach of confidentiality, or unauthorized disclosure or use of financial information, confidential client information, client or employee lists, trade secrets, or other Company confidential or proprietary information;
- (iv) willful violation of any law, rule or regulation of any governing authority or of the Company's policies and procedures, including, without limitation, the Company's employee handbook or similar document;
- (v) the willful refusal of Executive to follow the lawful directions of the CEO within a reasonable period after delivery to Executive of written notice of such directions;
- (vi) willful conduct that is grossly injurious to the reputation, financial condition, business or assets of the Company; or
 - (vii) willful breach of any material provision in an agreement with the Company.

In each of (i) through (vii) the Executive shall be given written notice of such cause for termination, and in each of (i) and (vii) the Executive shall be given an opportunity to remedy such cause for termination within sixty (60) business days of receipt of such notice.

- (c) "<u>Change in Control</u>" shall have the same meaning as the definition contained in Section 12.2 of the First Western Financial, Inc. 2016 Omnibus Incentive Plan.
- (d) Resignation for "<u>Good Reason</u>" means termination of employment by the Executive because of the occurrence of any of the following events:
 - (i) there is a material reduction in the Executive's Base Salary, unless agreed to in writing by the Executive;
 - (ii) there is a material reduction in the Executive's authority, duties, or responsibilities;
 - (iii) the Executive is required to relocate the Executive's principal place of employment with the Company (or successor to the Company, if applicable) to a place that increases the Executive's one-way commute by more than twenty-five (25 miles) as compared to the Executive's then-current principal place of employment immediately prior to such relocation (excluding regular travel in the ordinary course of business), unless such relocation is agreed to in writing by the Executive;

- (iv) the failure of any successor to assume this Agreement; and
- (v) any other action or inaction that constitutes a material breach by the Company of this Agreement after the Executive provides written notice to the Company of the facts which constitute the grounds within sixty (60) business days following the initial existence of the grounds and the Company thereafter fails to cure such grounds within sixty (60) business days following its receipt of such notice (or, in the event that such grounds cannot be corrected within such sixty (60) day period, the Company has not taken all reasonable steps within such sixty (60) day period to correct such grounds as promptly as practicable thereafter).
- 4. <u>Compensation and Other Benefits</u>. Subject to the provisions of this Agreement, the Company shall pay and provide the following compensation and other benefits to the Executive during the Term as compensation for services rendered hereunder:
- (a) <u>Base Salary</u>. The Company shall pay to the Executive an annual salary (the "<u>Base Salary</u>") payable in substantially equal installments at such intervals as may be determined by the Company in accordance with its ordinary payroll practices, as established from time to time. The Base Salary shall be reviewed annually and increased as appropriate for market changes, commencing effective January 1, 2024. Executive annualized Base Salary as of the Effective Date is \$320,000. The Base Salary shall not be decreased by the Company except with the prior written consent of the Executive.
- (b) <u>Annual Bonus</u>. Executive shall be eligible to receive annual incentive compensation (the "<u>Annual Bonus</u>"), as governed by the terms set forth in the First Western Financial, Inc. Incentive Plan for Senior Executive Officers, as may be amended from time to time. The incentive compensation performance measures and goals are reviewed by the Board of Directors annually, and adjusted as appropriate, according to the needs of the business.
- (c) <u>Long-Term Incentive Plan</u>. The Executive shall be eligible for grants under the First Western Financial, Inc. 2016 Omnibus Incentive Plan, including, but not limited to, grants of stock options, market conditioned performance share units, financial conditioned performance stock units and restricted stock units, as the Compensation Committee of the Board shall determine from time to time.
- (d) <u>Savings and Retirement Plans</u>. The Executive shall be eligible to participate in all savings and retirement plans applicable generally to other executives of the Company, in accordance with the terms of the plans, as may be amended from time to time.
- (e) <u>Welfare Benefit Plans</u>. The Executive and his eligible dependents shall be eligible to participate in and shall receive all benefits under the Company's welfare benefit plans and programs applicable generally to other executives of the Company, in accordance with the terms of the plans, as may be amended from time to time.
- (f) <u>Expenses</u>. Upon presentation of written documentation thereof, in accordance with the applicable expense reimbursement policies and procedures of the Company as in effect from time to time, the Company shall reimburse the Executive for reasonable business-related expenses incurred by the Executive in the fulfillment of his duties. Payments with respect to reimbursements of expenses shall be made promptly and in accordance with the applicable expense reimbursement policies and procedures of the Company, but in any event, on or before the last day of the calendar month following the calendar month in which the relevant expense is incurred.

- (g) <u>Vacation</u>. The Executive shall be entitled to vacation each calendar year during the Term, subject to and in accordance with the Company's vacation policy in effect from time to time.
- 5. <u>Termination of Employment</u>. The terms of any equity compensation grants outstanding as of the Effective Date, as well as any future equity compensation grants made under the First Western Financial, Inc. 2016 Omnibus Incentive Plan shall govern what the Executive receives on termination of employment, in terms of equity compensation, except as expressly provided in Section 5(b)(i)(C) and Section 5(b)(ii)(C) hereof. All other forms of remuneration the Executive is eligible for on termination of employment are set forth in this Section 5.
- (a) Termination for Cause; Resignation Without Good Reason. If, prior to the expiration of the Term, the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code"), and the Treasury Regulations ("Regulations") thereunder, by reason of the Company's termination of the Executive's employment for Cause, or if the Executive resigns from his employment hereunder other than for Good Reason, the Executive shall be entitled only to payment of (i) any unpaid Base Salary through and including the date of termination or resignation, (ii) any Annual Bonus earned, but unpaid, for the year immediately preceding the year in which the termination date occurs (which unpaid Annual Bonus amount shall be paid no later than March 15 of the year following the year in which the amount was earned), and (iii) any other amounts or benefits required to be paid or provided by law or under any plan, program, policy or practice of the Company (the amounts or benefits in (i) through (iii) being referred to collectively as the "Other Accrued Compensation and Benefits"). Except as set forth in this subsection (a), the Executive shall have no further right to receive any other compensation or benefits after such termination or resignation of employment.

(b) <u>Termination without Cause; Resignation for Good Reason.</u>

- (i) If the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Code and the Regulations thereunder, by reason of the Company's termination of the Executive's employment without Cause, or if the Executive resigns from his employment hereunder for Good Reason, the Executive shall be entitled to the following:
 - (A) The Other Accrued Compensation and Benefits.
- (B) An amount equal to one (1) times the Executive's Base Amount, which shall be payable in equal installments pursuant to the Company's normal payroll practices and subject to all legally required and customary withholdings for the twelve (12) month period following termination.
- (C) A pro-rated portion of each outstanding unvested equity award, including, without limitation, stock options, restricted stock and restricted stock units, held by the Executive shall automatically become vested and, if applicable, exercisable and any restrictions thereon shall immediately lapse. The pro-rated portion, determined separately for each vesting tranche for each such equity award, shall be based on a fraction, (I) the numerator of which is number of days that elapse from the grant date of the equity award until the Executive's termination date and the (II) the denominator of which is the total number of days in the period that begins on the date of grant of the applicable equity award and ends on the scheduled vesting date of such equity award (or, if applicable, the scheduled vesting date of the applicable vesting tranche). For purposes of determining the number of shares subject to any outstanding unvested equity awards

subject to performance vesting conditions (as opposed to solely service-based vesting conditions) that would otherwise vest pursuant to the foregoing sentence, the applicable performance goals shall be deemed achieved at the level of the Company's actual achievement of the applicable performance goals as measured as of the Executive's termination date, as determined by the Compensation Committee in its reasonable discretion. Each of the Executive's outstanding equity awards as of the Effective Date is hereby amended to incorporate the terms of this Section 5.1(b)(i)(C).

(D) The Company shall monthly pay to the Executive the amount equal to the full premium amount (determined as of the date of termination) for continued coverage under the Company's health plan pursuant to the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), for the Executive, and, to the extent that the Executive is providing coverage for his spouse or eligible dependents as of the termination date, for such individuals; provided, however, that the Company's obligation to pay such premiums shall cease immediately upon the earlier of (i) the passage of eighteen (18) months (ii) the expiration of the statutory COBRA period and (iii) the date the Executive becomes eligible for coverage under any other group health plan (as an employee or otherwise) or Medicare.

Notwithstanding the foregoing, if the Company terminates the Executive's employment without Cause, the Company shall provide the Executive with no less than ninety (90) days' written notice or payment of three (3) months Base Salary in lieu of ninety (90) days' written notice, which shall be in addition to payments described under this Section 5(b)(i).

- (E) All payments and benefits provided under this Section 5(b)(i)(B) and (D) shall commence on the first payroll date following the 60th day after the Executive's termination of employment, with the first installment payment to include any payments that would have otherwise been paid to the Executive if such payments commenced on the first payroll date following the Executive's termination date. The Company shall not be required to make the payments and provide the benefits provided for under this Section 5(b)(i)(A), (B), (C) or (D) unless the Executive executes and delivers to the Company, within sixty (60) days following the Executive's termination of employment, a release substantially in the form attached hereto as Exhibit A (the "Release"), and the Release has become effective and irrevocable in its entirety in such 60-day period. The Executive's failure or refusal to sign the Release (or the Executive's revocation of such Release in accordance with applicable laws) will result in the forfeiture of the payments and benefits under this Section 5(b)(i)(A), (B), (C) or (D). To the extent any amount payable under this Section 5 is deferred compensation subject to the Code, if the period during which the Executive has discretion to execute or revoke the Release straddles two of the Executive's taxable years, then the Company shall make the severance payments starting in the second of such taxable years, regardless of which taxable year the Executive actually delivers the executed Release to the Company. The Executive may not, directly or indirectly, designate the calendar year or timing of payments. This Section 5(b)(i) shall expressly not apply to payments made on account of a Change in Control, pursuant to Section 5(b)(ii) hereof.
- (F) If, following a termination of employment without Cause or a resignation for Good Reason, the Executive breaches the provisions of Sections 6 through 10 hereof or breaches any provision set forth in the executed copy of the general release of claims, the Executive shall not be eligible, as of the date of such breach, for the payments and benefits described in Section 5(b)(i)(A), (B), (C) or (D), and any and all obligations and agreements of the Company with respect to such payments shall

thereupon cease. This Section 5(b)(i)(F) shall expressly not apply to payments made on account of a Change in Control, pursuant to Section 5(b)(ii) hereof.

- (ii) If the Company undergoes a Change in Control and, within 24 months of such Change in Control, the Executive is terminated without Cause or resigns for Good Reason, then, in lieu of (and not in addition to) any payments or benefits payable to the Executive pursuant to Section 5(b)(i), the Executive shall be entitled to the following:
 - (A) The Other Accrued Compensation and Benefits.
- (B) An amount equal to two (2) times the Executive's Base Amount, payable in a single lump sum payment.
- (C) Each outstanding equity award, including, without limitation, stock options, restricted stock and restricted stock units, held by the Executive shall automatically become vested and, if applicable, exercisable and any restrictions thereon shall immediately lapse, in each case, with respect to that number of shares of Company common stock that would otherwise vest based on Executive's continued employment. For purposes of determining the number of shares subject to any outstanding equity awards subject to performance vesting conditions that would otherwise vest pursuant to the foregoing sentence, the applicable performance goals shall be deemed achieved at the "target" level. Each of the Executive's outstanding equity awards as of the Effective Date is hereby amended to incorporate the terms of this Section 5.1(b)(ii)(C).
- (D) Monthly payments to the Executive equal to the full premium amount (determined as of the date of termination) for continued coverage under the Company's health plan pursuant to COBRA for the Executive, and, to the extent that the Executive is providing coverage for his spouse or eligible dependents as of the termination date, for such individuals; provided, however, that the Company's obligation to pay such premiums shall cease immediately upon the earlier of (i) the passage of eighteen (18) months (ii) the expiration of the statutory COBRA period and (iii) the date the Executive becomes eligible for coverage under any other group health plan (as an employee or otherwise) or Medicare.

Notwithstanding the foregoing, if the Company (or its successor) terminates the Executive's employment without Cause, the Company (or its successor) shall provide the Executive with no less than ninety (90) days' written notice or payment of three (3) months Base Salary in lieu of ninety (90) days' written notice, which shall be in addition to payments described under this Section 5(b)(ii).

(E) Unless otherwise provided herein, all payments and benefits provided under this Section 5(b)(ii)(B) and (D) shall be paid (or, in the case of the payments described in Section 5(b)(ii)(D), commence to be paid) on the first payroll date following the 60th day after the Executive's termination of employment, with the first payment to include any payments provided under Section 5(b)(ii)(D) that would have otherwise been paid to the Executive if such payments commenced on the first payroll date following the Executive's termination date. The Company shall not be required to make the payments and provide the benefits provided for under this Section 5(b)(ii)(A), (B), (C) or (D) unless the Executive executes and delivers the Release to the Company, within sixty (60) days following the Executive's termination of employment, and the Release has become effective and irrevocable in its entirety in such 60-day period. The Executive's failure or refusal to sign the Release (or the Executive's

revocation of such Release in accordance with applicable laws) will result in the forfeiture of the payments and benefits under this Section 5(b)(ii)(A), (B), (C) or (D). To the extent any amount payable under this Section 5 is deferred compensation subject to the Code, if the period during which the Executive has discretion to execute or revoke the Release straddles two of the Executive's taxable years, then the Company shall make the severance payments starting in the second of such taxable years, regardless of which taxable year the Executive actually delivers the executed Release to the Company. The Executive may not, directly or indirectly, designate the calendar year or timing of payments.

- (F) In the event that it is determined that any payment or distribution of any type to or for the benefit of an Executive made by the Company, by any of its Affiliates, by any person who acquires ownership or effective control or ownership of a substantial portion of the Company's assets (within the meaning of Code Section 280G) or by any affiliate of such person, whether paid or payable or distributed or distributable pursuant to the terms of any equity compensation plan, this Agreement or otherwise (the "Total Payments"), would be subject to the excise tax imposed by Code Section 4999 or any interest or penalties with respect to such excise tax (the "Excise Tax"), then, notwithstanding any other provision of this Agreement or any equity compensation plan to the contrary, any right of the Executive to any payment or benefit under this Agreement or any such equity compensation plan shall be reduced or eliminated, but only to the extent necessary to avoid imposition of the Excise Tax. In no case, however, shall such cutback be made if Total Payments after the imposition of the Excise Tax are greater than Total Payments cut back as provided in this Section 5(b) (ii) to avoid the Excise Tax.
- (G) In the event that a cutback of Total Payments is permitted under Section 5(b)(ii)(F), and except as required by Code Section 409A or to the extent that Code 409A permits discretion, the Compensation Committee shall have the right, in the Compensation Committee's sole discretion, to designate those rights, payments, or benefits and all other agreements that should be reduced or eliminated so as to provide the Executive with the maximum pre-tax amount which avoids imposition of the Excise Tax. For example, the Compensation Committee may choose to cut back cash severance, if that would yield a higher pre-tax amount than cutting back equity. Notwithstanding the foregoing, to the extent any payment or benefit constitutes deferred compensation under Code Section 409A, in order to comply with Code Section 409A the Compensation Committee shall instead accomplish such reduction by first reducing or eliminating any cash payments (with the payments to be made furthest in the future being reduced first), then by reducing or eliminating any accelerated vesting of options or stock appreciation rights, then by reducing or eliminating any accelerated vesting of restricted stock or restricted stock units.
- (c) <u>Termination Due to Death or Disability</u>. The Executive's employment with the Company shall terminate automatically on the Executive's death. In the event of the Executive's Disability (as defined herein), the Company shall be entitled to terminate his employment. In the event of the Executive's death or if the Executive incurs a "Separation from Service" within the meaning of Section 409A of the Code, or the Regulations thereunder, by reason of the Executive's Disability, the Company shall pay to the Executive (or his estate, as applicable), (i) the Executive's Base Salary through and including the date of termination and any Other Accrued Compensation and Benefits (ii) a pro-rata Annual Bonus for the year of termination, based on actual audited year-end results and payable when bonuses are normally paid to employees, and (iii) three (3) months Base Salary at the rate then in effect, payable in equal installments pursuant to the Company's normal payroll practices and subject to all legally required and customary withholdings for the three (3) month following termination.

- (i) For purposes of this Agreement, "<u>Disability</u>" means a physical or mental disability or infirmity of the Executive that prevents the normal performance of substantially all his duties for a period in excess of ninety (90) consecutive days or for more than ninety (90) days in any consecutive twelve (12)-month period. Evidence of such physical or mental disability or infirmity shall be certified by a physician licensed to practice in the state of residence of the Executive, which physician is mutually agreeable to the Board and the Executive. If there is no agreement on the selection of the physician, then the Board shall select one physician and the Executive shall select one physician, and the two physicians shall attempt to mutually agree upon such physical or mental disability or infirmity. If the two physicians cannot agree, then the two physicians shall jointly select a third physician, whose opinion on such physical or mental disability or infirmity shall control.
- (d) <u>Notice of Termination</u>. Any termination of employment by the Company or the Executive shall be communicated by a written "<u>Notice of Termination</u>" to the other party hereto given in accordance with Section 24 of this Agreement. In the event of a termination by the Company for Cause, or by the Executive for Good Reason, the Notice of Termination shall (i) indicate the specific termination provision in this Agreement relied upon, (ii) set forth in reasonable detail the facts and circumstances claimed to provide a basis for termination of the Executive's employment under the provision so indicated and (iii) specify the date of termination, which date shall not be more than thirty (30) business days after the giving of such notice, provided that the date of termination will not occur before the expiration of any applicable cure period.

The failure by the Executive or the Company to set forth in the Notice of Termination any fact or circumstance which contributes to a showing of Good Reason or Cause shall not waive any right of the Executive or the Company, respectively, hereunder or preclude the Executive or the Company, respectively, from asserting such fact or circumstance in enforcing the Executive's or the Company's rights hereunder.

- (e) <u>No Further Rights</u>. The Executive shall have no further rights under this Agreement or otherwise to receive any other compensation or benefits after such termination or resignation of employment.
 - 6. <u>Confidentiality</u>.
 - (a) <u>Confidential Information</u>.
 - (i) The Executive agrees that he will not at any time, except with the prior written consent of the Company or its Affiliates or as required by applicable law, directly or indirectly, reveal to any person, entity or other organization (other than the Company or its Affiliates or its respective employees, officers, directors, shareholders or agents) or use for the Executive's own benefit any confidential or proprietary information of any member of the Company or its Affiliates ("Confidential Information") relating to the assets, liabilities, employees, goodwill, business or affairs of any member of the Company or its Affiliates, including, without limitation, any information concerning past, present or prospective clients, intellectual capital, marketing data, or other confidential information used by, or useful to, any member of and known to the Executive by reason of the Executive's employment by, shareholdings in or other association with the Company or its Affiliates, other than disclosure while employed by the Company which the Executive reasonably and in good faith believes to be in or not opposed to the interests of the Company; provided that such Confidential Information does not include any information which is available to the general public or is generally available within the relevant business or industry other than as a result of the Executive's breach of this

Agreement. Confidential Information may be in any medium or form, including, without limitation, physical documents, computer files or disks, videotapes, audiotapes, and oral communications.

- (ii) In the event that the Executive becomes legally compelled to disclose any Confidential Information, the Executive shall, if permitted by law, provide the Company with prompt written notice so that the Company may seek a protective order or other appropriate remedy. In the event that such protective order or other remedy is not obtained, the Executive shall furnish only that portion of such Confidential Information or take only such action as is legally required by binding order and shall exercise his reasonable efforts to obtain reliable assurance that confidential treatment shall be accorded any such Confidential Information. The Company shall promptly pay (upon receipt of invoices and any other documentation as may be requested by the Company) all reasonable expenses and fees incurred by the Executive, including attorneys' fees, in connection with his compliance with the immediately preceding sentence. Notwithstanding anything herein to the contrary, nothing in this Agreement shall (A) prohibit the Executive from making reports of possible violations of federal law or regulation to any governmental agency or entity in accordance with the provisions of and rules promulgated under Section 21F of the Securities Exchange Act of 1934 or Section 806 of the Sarbanes-Oxley Act of 2002, or of any other whistleblower protection provisions of state or federal law or regulation, or (B) require notification or prior approval by the Company of any reporting described in clause (A).
- (b) <u>Confidentiality of Agreement</u>. The Executive agrees that, except as may be required by applicable law or legal process, during the Term and thereafter, he shall not disclose the terms of this Agreement to any person or entity other than the Executive's accountants, financial advisors, attorneys or spouse, provided that such accountants, financial advisors, attorneys and spouse agree not to disclose the terms of this Agreement to any other person or entity.
- (c) <u>Exclusive Property</u>. The Executive confirms that all Confidential Information is and shall remain the exclusive property of the Company or its Affiliates. All business records, papers and documents kept or made by the Executive relating to the business of the Company or its Affiliates shall be and remain the property of the Company or its Affiliates. Upon the request and at the expense of the Company or its Affiliates, the Executive shall promptly make all disclosures, execute all instruments and papers and perform all acts reasonably necessary to vest and confirm in the Company or its Affiliates, fully and completely, all rights created or contemplated by this Section 6.
- 7. <u>Noncompetition</u>. The Executive agrees that, for a period commencing on the January 1, 2020 and ending 365 days following the Executive's termination of employment (the "<u>Restricted Period</u>"), the Executive shall not, without the prior written consent of the Company, directly or indirectly, and whether as principal or investor or as an employee, officer, director, manager, partner, consultant, agent or otherwise, alone or in association with any other person, firm, corporation or other business organization, carry on a Competing Business (as defined herein) within the then current footprint of the Company. For purposes of this Section 7: (a) carrying on a "<u>Competing Business</u>" means to engage in the competing business of any business carried on by the Company or its Affiliates. Notwithstanding the foregoing nothing herein shall limit the Executive's right to own not more than one (1) percent of any of the debt or equity securities of any business organization that is then filing reports with the Securities and Exchange Commission pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended. The Executive acknowledges that this noncompetition provision will not unreasonably impair or infringe upon the Executive's right to work or earn a living during the Restricted Period, and the Executive further acknowledges that these restrictions are necessary to

protect the trade secrets of the Company as the Executive's breach of this provision would necessarily involve the Executive's use of the Company's trade secrets.

Initial

Non-Solicitation and Non-Acceptance. The Executive agrees that for the Restricted Period the Executive shall not, directly or indirectly, (a) interfere with or attempt to interfere with the relationship between any person who is, or was during the Restricted Period or the 3-month period immediately preceding the commencement of the Restricted Period, an employee, officer, representative or agent of the Company or its Affiliates and any member of the Company or its Affiliates, or solicit, induce or attempt to solicit or induce any of them to terminate their employment or service relationship with any member of the Company or its Affiliates or violate the terms of their respective service contracts, or any employment arrangements, with such entities, provided that the foregoing shall not prevent general employment or service solicitations that do not specifically target any such persons; or (b) induce or attempt to induce any customer or client of any member of the Company or its Affiliates to cease doing business with any member of the Company or its Affiliates, or in any way interfere with the relationship between any member of the Company or its Affiliates and any customer or client of any member of the Company or its Affiliates. The Executive will make reasonable best efforts to not accept, and when in his control, not allow the acceptance of any of the Company relationships in this Section 8 at a new employer during the Restricted Period. The Executive acknowledges that this non-solicitation provision will not unreasonably impair or infringe upon the Executive's right to work or earn a living during the Restricted Period, and the Executive further acknowledges that these restrictions are necessary to protect the trade secrets of the Company as the Executive's breach of this provision would necessarily involve the Executive's use of the Company's trade secrets.

Initial

- 9. <u>No Conflicting Agreement</u>. The Executive represents, warrants and covenants to the Company that the Executive is not a party to any agreement, whether written or oral, that would be breached by or would prevent or interfere with the executive of the Executive of this Agreement or the fulfillment by the Executive of the Executive's obligations hereunder.
- 10. <u>Nondisparagement</u>. Each party represents, warrants and covenants to the other that at no time during the Term or thereafter shall such party make, or cause or assist any other person to make, any statement or other communication to any third party which impugns or attacks, or is otherwise critical of, the reputation, business or character of the other party or any of its respective directors, officers or employees, as applicable; provided this Section shall not prohibit truthful testimony by or on behalf of either party in any judicial or administrative proceeding.
- 11. Section 409A of the Code. This Agreement is intended to meet the requirements of Section 409A of the Code and shall be interpreted and construed consistent with that intent. Notwithstanding any other provision of this Agreement, to the extent that the right to any payment (including the provision of benefits) hereunder provides for the "deferral of compensation" within the meaning of Section 409A(d)(l) of the Code, if the Executive is a "Specified Employee" within the meaning of Section 409A(a)(2)(B)(i) of the Code on the date of the Executive's "Separation from Service" within the meaning of Section 409A(a)(2)(A)(i) of the Code (the "Separation Date"), then no such payment shall be made or commence during the period beginning on the Separation Date and ending on the date that is six months following the Separation Date or, if earlier, on the date of the Executive's death. The amount of any payment that would otherwise be paid to the Executive during this period shall instead be paid to the Executive on the fifteenth day of the first calendar month following the end of the period.

12. <u>Certain Remedi</u>es.

- (a) <u>Forfeiture/Payment Obligations</u>. In the event the Executive fails to comply with Sections 6 through 10, other than any isolated, insubstantial and inadvertent failure, the Executive agrees that he will forfeit any amounts not already paid pursuant to Section 5(b)(i) (A), (B), (C) or (D) of this Agreement. Notwithstanding the previous sentence, the Executive shall be given written notice of each alleged failure to comply with Sections 6 through 10, and the Executive shall be given an opportunity to remedy such failure within (60) sixty business days of the receipt of such notice. For purposes of clarity, the Executive's failure to comply with Sections 6 through 10 shall not result in forfeiture of amounts required to be paid but not already paid on account of a Change in Control pursuant to Section 5(b)(ii) of this Agreement.
- (b) <u>Injunctive Relief.</u> Without intending to limit the remedies available to the Company or its Affiliates, including, but not limited to, that set forth in Section 12(a) hereof, the Executive agrees that a breach of any of the covenants contained in Sections 6 through 10 of this Agreement may result in material and irreparable injury to the Company or its Affiliates for which there is no adequate remedy at law, that it will not be possible to measure damages for such injuries precisely and that, in the event of such a breach or threat thereof, any member of the Company or its Affiliates shall be entitled to seek a temporary restraining order or a preliminary or permanent injunction, or both, without bond or other security, restraining the Executive from engaging in activities prohibited by the covenants contained in Sections 6 through 10 of this Agreement or such other relief as may be required specifically to enforce any of the covenants contained in this Agreement. Such injunctive relief in any court shall be available to the Company or its Affiliates in lieu of, or prior to or pending determination in, any arbitration proceeding.
- 13. <u>Defense of Claims</u>. The Executive agrees that, during the Term, and for a period of seven (7) years after termination of the Executive's employment, upon request from the Company, the Executive will reasonably cooperate with the Company in the defense of any claims or actions that may be made by or against the Company that affect the Executive's prior areas of responsibility, except if the Executive's reasonable interests are adverse to the Company in such claim or action. The Company agrees to promptly pay in advance or reimburse the Executive for, as requested by the Executive, all of the Executive's reasonable travel and other direct costs and expenses incurred, or to be reasonably incurred, to comply with the Executive's obligations under this Section 13, including, but not limited to, legal costs and expenses.
- Alternative Dispute Resolution. The Company and the Executive agree that any dispute that arises out of or relates to Executive's employment or termination of employment with the Company, including any dispute that the Executive may have with any present or former officer, manager, director, employee, agent, attorney or insurer of the Company, shall first be submitted to mediation through the Institute for Conflict Prevention & Resolution ("CPR") (or such other nationally-recognized alternative dispute resolution service as the Executive and Company may agree). The Executive and the Company shall use their reasonable efforts to commence and conclude such mediation in a prompt manner. If the dispute is not resolved through mediation within thirty (30) days after notice thereof, such dispute shall be resolved by binding arbitration in accordance with the rules and procedures of the CPR (or such other nationally recognized alternative dispute resolution service as the Executive and the Company may agree). Judgment upon the award rendered by the arbitrator may be entered in any court having in person and subject matter jurisdiction. The Company and the Executive hereby submit to the jurisdiction of the federal and state courts in Denver, Colorado, for the purpose of confirming any such award and entering judgment thereon. The Company shall pay for all administrative costs and fees charged by the CPR (or such other nationally recognized alternative dispute resolution service) as well as the fees charged by the arbitrator. Each party shall pay for his or its attorneys' fees and costs.

11

- 15. No assignability; Binding Agreement.
- (a) <u>By the Executive</u>. This Agreement and any and all rights, duties, obligations or interests hereunder shall not be assignable or delegable by the Executive.
- (b) <u>By the Company</u>. This Agreement and all of the Company's rights and obligations hereunder shall not be assignable by the Company except as incident to a reorganization, merger or consolidation, or transfer of all or substantially all of the Company's assets.
- (c) <u>Binding Effect</u>. This Agreement shall be binding upon, and inure to the benefit of, the parties hereto, any successors to or assigns of the Company and the Executive's heirs and the personal representatives of the Executive's estate.
- 16. <u>Withholding</u>. Any payments made or benefits provided to the Executive under this Agreement shall be reduced by any applicable withholding taxes or other amounts required to be withheld by law or contract.
- 17. <u>Amendment; Waiver</u>. This Agreement may not be modified, amended or waived in any manner, except by an instrument in writing signed by both parties hereto. The waiver by either party of compliance with any provision of this Agreement by the other party shall not operate or be construed as a waiver of any other provision of this Agreement, or of any subsequent breach by such party of a provision of this Agreement.
- 18. <u>Governing Law and Forum</u>. The Executive and the Company agree that this Agreement and all matters or issues arising out of or relating to the Executive's employment with the Company shall be governed by the laws of the State of Colorado applicable to contracts entered into and performed entirely therein. Any action to enforce this Agreement shall be brought solely in the state or federal courts located in the County of Denver, Colorado.
- 19. <u>Survival of Certain Provisions</u>. Unless expressly provided otherwise, the rights and obligations set forth in this Agreement shall survive any termination or expiration of this Agreement.
- 20. <u>Entire Agreement; Supersedes Previous Agreements</u>. This Agreement contains the entire agreement and understanding of the parties hereto with respect to the matters covered herein, and supersedes all prior or contemporaneous negotiations, commitments, agreements and writings with respect to the subject matter hereof (including, without limitation, the Prior Agreement), all such other negotiations, commitments, agreements and writings shall have no further force or effect, and the parties to any such other negotiation, commitment, agreement or writing shall have no further rights or obligations thereunder.
- 21. <u>Severability</u>. The provisions of the Agreement shall be deemed severable, and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions herein.
- 22. <u>Counterparts</u>. This Agreement may be executed by either of the parties hereto in counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same instrument.
- 23. <u>Headings</u>. The headings of sections herein are included solely for convenience of reference and shall not control the meaning or interpretation of any of the provisions of this Agreement.

24. <u>Notices</u>. All notices or communications hereunder shall be in writing, addressed as follows:

To the Company:

First Western Financial, Inc. 1900 16th St Suite #1200 Denver, CO 80202 Attention: Secretary and General counsel

To the Executive:

John Sawyer 701 Race Street Denver, CO 80206

All such notices shall be conclusively deemed to be received and shall be effective (i) if sent by hand delivery or nationally recognized courier, upon receipt or (ii) if sent by electronic mail or facsimile, upon receipt by the sender of such transmission.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

IN WITNESS WHEREOF, the Company has caused this Agreement to be signed by its officer pursuant to the authority of its Board, and the Executive has executed this Agreement, as of the day and year first written above.

| FIRST WESTERN FINANCIAL, II | NC. |
|-----------------------------|-----|
| By: | |
| JOHN SAWYER | |
| _ | |

GENERAL RELEASE OF CLAIMS

| This General Release of all Claims | (this "Agreement") is entered into by John Sawyer (the "Executive") and |
|---------------------------------------|---|
| FWFI (the "Company"), effective as of | _ in connection with the termination of the Executive's employment with the |
| Company as of | |
| | |

In consideration of the promises set forth in the Amended and Restated Employment Agreement between the Executive and the Company, dated effective April 26, 2023 (the "Employment Agreement"), the Executive and the Company agree as follows:

1. <u>Return of Property</u>. All Company files, access keys and codes, desk keys, ID badges, computers, records, manuals, electronic devices, computer programs, papers, electronically stored information or documents, telephones and credit cards, and any other property of the Company in the Executive's possession must be returned no later than the date of the Executive's termination from the Company; provided, that, after the notification of a consultation with the Company, the Executive may keep one copy of such items as he may reasonably expect to use to protect his rights under this Agreement.

2. General Release and Waiver of Claims.

- (a) Release. In consideration of the payments and benefits provided to the Executive under the Employment Agreement and after consultation with counsel, the Executive and each of the Executive's respective heirs, executors, administrators, representatives, agents, insurers, successors and assigns (collectively, the "Releasors") hereby irrevocably and unconditionally release and forever discharge the Company, its subsidiaries and affiliates and each of their respective officers, employees, directors, shareholders and agents ("Releasees") from any and all claims, actions, causes of action, rights, judgments, obligations, damages, demands, accountings or liabilities of whatever kind or character (collectively, "Claims"), including, without limitation, any Claims under any federal, state, local or foreign law, that the Releasors may have, or in the future may possess, arising out of (i) the Executive's employment relationship with and service as an employee, officer or director of the Company or any subsidiaries or affiliated companies and the termination of such relationship or service, and any event, condition, circumstance or obligation that occurred, existed or arose on or prior to the date hereof; provided, however, that the Executive does not release, discharge or waive any rights to (i) payments and benefits provided under the Employment Agreement that are contingent upon the execution by the Executive of this Agreement and (ii) any indemnification rights the Executive may have under the Employment Agreement, in accordance with the Company's governance instruments or under any director and officer liability insurance maintained by the Company with respect to liabilities arising as a result of the Executive's service as an officer and employee of the Company. This Section 2(a) does not apply to any Claims that the Releasors may have as of the date the applicable rules and regulations promulgated thereunder ("ADEA"). Claims arising under ADEA are addressed in Section 2(b) of this Agreement.
- (b) <u>Specific Release of ADEA Claims</u>. In further consideration of the payments and benefits provided to the Executive under the Employment Agreement, the Releasors hereby unconditionally release and forever discharge the Releasees from any and all Claims arising under ADEA that the Releasors may have as of the date the Executive signs this Agreement. By signing this Agreement, the Executive hereby acknowledges and confirms the following: (i) the Executive was advised by the Company in connection with his termination to

consult with an attorney of his choice prior to signing this Agreement: and to have such attorney explain to the Executive the terms of this Agreement, including, without limitation, the terms relating to the Executive's release of claims arising under ADEA, and the Executive has in fact consulted with an attorney; (ii) the Executive was given a period of not fewer than 21 days to consider the terms of this Agreement and to consult with an attorney of his choosing with respect thereto; (iii) the Executive knowingly and voluntarily accepts the terms of this Agreement; and (iv) the Executive is providing this release and discharge only in exchange for consideration in addition to anything of value to which the Executive is already entitled. The Executive also understands that he has seven days following the date on which he signs this Agreement within which to revoke the release contained in this Section, by providing the Company with a written notice of his revocation of the release and waiver contained in this Section.

(c) <u>No Assignment</u>. The Executive represents and warrants that he has not assigned any of the Claims being released under this Agreement. The Company may assign this Agreement, in whole or in part, to any affiliated company or subsidiary of, or any successor in interest to, the Company.

3. Proceedings.

- (a) General Agreement Relating to Proceedings. The Executive has not filed, and except as provided in Sections 3(b) and 3(c), the Executive agrees not to initiate or cause to be initiated on his behalf, any complaint, charge, claim or proceeding against the Releasees before any local, state or federal agency, court or other body relating to his employment or the termination of his employment, other than with respect to the obligations of the Company to the Executive under the Employment Agreement (each, individually, a "Proceeding"), and agrees not to participate voluntarily in any Proceeding. The Executive waives any right he may have to benefit in any manner from any relief (whether monetary or otherwise) arising out of any Proceeding.
- (b) <u>Proceedings Under ADEA</u>. <u>Section 3(a)</u> shall not preclude the Executive from filing any complaint, charge, claim or proceeding challenging the validity of the Executive's waiver of Claims arising under ADEA (which is set forth in <u>Section 2(b)</u> of this Agreement). However, both the Executive and the Company confirm their belief that the Executive's waiver of claims under ADEA is valid and enforceable, and that their intention is that all claims under ADEA will be waived.
- (c) <u>Certain Administrative Proceedings</u>. In addition, <u>Section 3(a)</u> shall not preclude the Executive from filing a charge with or participating in any administrative investigation or proceeding by the Equal Employment Opportunity Commission or another Fair Employment Practices agency. The Executive is, however, waiving his right to recover money in connection with any such charge or investigation. The Executive is also waiving his right to recover money in connection with a charge filed by any other entity or individual, or by any federal, state or local agency.
- 4. Remedies. In the event the Executive initiates or voluntarily participates in any Proceeding in violation of this Agreement, or if he fails to abide by any of the terms of this Agreement or his post-termination obligations contained in the Employment Agreement, or if he revokes the ADEA release contained in Section 2(b) within the seven-day period provided under Section 2(b), the Company may, in addition to any other remedies it may have, reclaim any amounts paid to his under Sections 5(b)(i)(A), (B), (C) or (D) of the Employment Agreement or terminate any benefits or payments that are subsequently due under Sections 5(b)(i)(A), (B), (C) and (D) of the Employment Agreement, without waiving the release granted herein. The Executive acknowledges and agrees that the remedy at law available to the Company for breach of any of his post-termination obligations under the Employment Agreement or his obligations

under <u>Sections 2</u> and <u>3</u> herein would be inadequate and that damages flowing from such a breach may not readily be susceptible to measurement in monetary terms. Accordingly, the Executive acknowledges, consents and agrees that, in addition to any other tights or remedies that the Company may have at law or in equity or as may otherwise be set forth in the Employment Agreement, the Company shall be entitled to seek a temporary restraining order or a preliminary or permanent injunction, or both, without bond or other security, restraining the Executive from breaching his post-termination obligations under the Employment Agreement or his obligations under <u>Sections 2</u> and <u>3</u> herein. Such injunctive relief in any court shall be available to the Company, in lieu of, or prior to or pending determination in, any arbitration proceeding.

The Executive understands that by entering into this Agreement he shall be limiting the availability of certain remedies that he may have against the Company and limiting also his ability to pursue certain claims against the Company.

- 5. <u>Severability Clause</u>. In the event that any provision or part of this Agreement is found to be invalid or unenforceable, only that particular provision or part so found, and not the entire Agreement, shall be inoperative.
- 6. <u>Non-Admission</u>. Nothing contained in this Agreement shall be deemed or construed as an admission of wrongdoing or liability on the part of the Company or the Executive.
- 7. <u>Governing Law and Forum</u>. The Executive and the Company agree that this Agreement and all matters or issues arising out of or relating to the Executive's employment with the Company shall be resolved pursuant to the provisions of Section 14 of the Employment Agreement.
 - 8. <u>Notices</u>. All notices or communications hereunder shall be in writing, addressed as follows:

To the Company:

First Western Financial, Inc. 1900 16th St Suite #1200 Denver, CO 80202 Attention: Secretary

To the Executive: John Sawyer 701 Race Street Denver, CO 80206

All such notices shall be conclusively deemed to be received and shall be effective (i) if sent by hand delivery or nationally recognized courier, upon receipt or (ii) if sent by electronic mail or facsimile, upon receipt by the sender of such transmission.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK

THE EXECUTIVE ACKNOWLEDGES THAT HE HAS READ THIS AGREEMENT AND THAT HE FULLY KNOWS, UNDERSTANDS AND APPRECIATES ITS CONTENTS, AND THAT HE HEREBY EXECUTES THE SAME AND MAKES THIS AGREEMENT AND THE RELEASE AND AGREEMENTS PROVIDED FOR HEREIN VOLUNTARILY AND OF HIS OWN FREE WILL.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first set forth above.

First Western Financial, Inc.

By: <u>/s/ Scott C. Wylie</u> Scott C. Wylie, Chairman, President, and CEO

THE EXECUTIVE

<u>/s/ John Sawyer</u> John Sawyer, Chief Investment Officer

Dated: April 26, 2023

FIRST WESTERN FINANCIAL, INC. 2016 OMNIBUS INCENTIVE PLAN (As Amended and Restated April 26, 2023)

- 1. Establishment & Purpose.
- 1.1 First Western Financial, Inc., a Colorado corporation, sets forth herein the terms of its 2016 Omnibus Incentive Plan.
- 1.2 The purpose of the First Western Financial, Inc. 2016 Omnibus Incentive Plan is to further align the interests of eligible participants with those of the Company's stockholders by providing long-term cash and equity incentive compensation opportunities tied to the performance of the Company and/or its Common Stock. The Plan is intended to advance the interests of the Company and increase stockholder value by attracting, retaining and motivating key personnel upon whose judgment, initiative and effort the successful conduct of the Company's business is largely dependent. The Plan was originally adopted on November 18, 2016 (the "Original Effective Date"). This amendment and restatement of the Plan amends the Plan effective as April 26, 2023 (the "Effective Date") to provide for continued vesting of Awards held by a Retirement Eligible Employee following his or her Retirement (as those terms are defined below), subject to certain conditions as specified herein.
 - 2. <u>Definitions</u>. Wherever the following capitalized terms are used in the Plan, they shall have the meanings specified below:
- "Affiliate" means, with respect to the Company, any company or other trade or business that controls, is controlled by or is under common control with the Company within the meaning of Rule 405 of Regulation C under the Securities Act, including, without limitation, any Subsidiary.
- "Award" means an award of a Stock Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit, Cash Performance Award, or Stock Award granted under the Plan.
- "Award Agreement" means a notice or an agreement (whether written or electronic) entered into between the Company and a Participant setting forth the terms and conditions of an Award granted to a Participant as provided in Section 16.2 hereof.
 - "Beneficial Owner" shall have the meaning ascribed to such term in Rule 13d-3 under the Exchange Act.
 - "Board" means the Board of Directors of the Company.
- "Cash Performance Award" means an Award that is denominated by a cash amount to an Eligible Person under Section 10 hereof and payable based on or conditioned upon the attainment of pre-established business and/or individual Performance Goals over a specified performance period and subject to such conditions as are set forth in the Plan and the applicable Award Agreement.
 - "Cause" shall have the meaning set forth in Section 13.2 hereof.
 - "Change in Control" shall have the meaning set forth in Section 12.2 hereof.
 - "Code" means the Internal Revenue Code of 1986, as amended.
- "Committee" means (i) the Compensation Committee of the Board, (ii) such other committee of the Board appointed by the Board to administer the Plan or (iii) as provided in Section 3.1 hereof, the full Board.
 - "Common Stock" means the Company's common stock, par value \$0.001 per share, as the same may be reclassified, exchanged or recapitalized.
 - "Company" means First Western Financial, Inc., a Colorado corporation, or any successor thereto.
- "Date of Grant" means the date on which an Award under the Plan is granted or approved for grant by the Committee or such later date as the Committee may specify to be the effective date of an Award.

"Disability" shall have the meaning set forth below, except with respect to any Participant who has an effective employment agreement or service agreement with the Company or one of its Subsidiaries that defines "Disability" or a like term, in which event the definition of "Disability" as set forth in such agreement shall be deemed to be the definition of "Disability" herein solely for such Participant and only for so long as such agreement remains effective. In all other events, the term "Disability" shall mean Participant's inability to perform the essential duties, responsibilities and functions of Participant's position with the Company and its Subsidiaries for a period of ninety (90) consecutive days or for a total of one hundred eighty (180) days during any twelve (12)-month period as a result of any mental or physical illness, disability or incapacity even with reasonable accommodations for such illness, disability or incapacity even with reasonable accommodations for such illness, disability or incapacity even with reasonable and which condition is expected to last for a continuous period of not less than twelve (12) months, all as determined by the Committee in its reasonable good faith judgment. Participant shall cooperate in all respects with the Company if a question arises as to whether he has become disabled (including, without limitation, submitting to reasonable examinations by one or more medical doctors and other health care specialists selected by the Company and authorizing such medical doctors and other health care specialists selected by the Company. Notwithstanding anything to the contrary contained herein, and solely for purposes of any Incentive Stock Option, "Disability" shall mean a permanent and total disability (within the meaning of Section 22(e)(3) of the Code).

"Effective Date" shall have the meaning set forth in Section 1.2 hereof.

"Eligible Person" means any person who is an employee, Non-Employee Director, consultant or other personal service provider of the Company or any of its Subsidiaries.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder from time to time.

"Fair Market Value" means, as of any given date, the value of a share of Common Stock determined as follows:

- (a) For purposes of any Awards granted on the date of a Public Offering, the Fair Market Value will be the initial price to the public as set forth in the final prospectus included within the registration statement in Form S-1 filed with the Securities and Exchange Commission for the Public Offering.
- (b) For purposes of any Awards granted on any other date, the Fair Market Value will be the closing price as reported on NASDAQ or any other principal exchange on which the Common Stock is then listed on such date, or if the Common Stock was not traded on such date, then on the next preceding trading day that the Common Stock was traded on such exchange, as reported by such responsible reporting service as the Committee may select. If the Common Stock is not listed on any such exchange, "Fair Market Value" shall be such value as determined by the Board or the Committee in its discretion and, to the extent necessary, shall be determined in a manner consistent with Section 409A of the Code and the regulations thereunder.

The determination of fair market value for purposes of tax withholding may be made in the Committee's discretion subject to applicable laws and is not required to be consistent with the determination of Fair Market Value for other purposes.

"Family Member" means a person who is a spouse, former spouse, child, stepchild, grandchild, parent, stepparent, grandparent, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother, sister, brother-in-law, or sister-in-law, including adoptive relationships, of the Grantee, any person sharing the Grantee's household (other than a tenant or employee), a trust in which any one or more of these persons have more than fifty percent of the beneficial interest, a foundation in which any one or more of these persons (or the Grantee) control the management of assets, and any other entity in which one or more of these persons (or the Grantee) own more than fifty percent of the voting interests; provided, however, that to the extent required by applicable law, the term Family Member shall be limited to a person who is a spouse, former spouse, child, stepchild, grandchild, parent, stepparent, grandparent, mother-in-law, son-in-law, daughter-in-law, brother, sister, brother-in-law, or sister-in-law, including adoptive relationships of the Grantee or a trust or foundation for the exclusive benefit of any one or more of these persons.

"Grant" means an Award under the Plan.

"Grantee" means a person who receives or holds a Grant under the Plan.

- "Incentive Stock Option" means a Stock Option granted under Section 6 hereof that is intended to meet the requirements of Section 422 of the Code and the regulations thereunder.
 - "NASDAQ" means The NASDAQ Global Market.
 - "Non-Employee Director" means a member of the Board who is not an employee of the Company or any of its Subsidiaries.
 - "Nonqualified Stock Option" means a Stock Option granted under Section 6 hereof that is not an Incentive Stock Option.
 - "Participant" means any Eligible Person who holds an outstanding Award under the Plan.
 - "Performance Criteria" shall have the meaning set forth in Section 10.3 hereof.
 - "Performance Goals" shall have the meaning set forth in Section 10.4 hereof.
- "Performance Stock Unit" means a Restricted Stock Unit denominated as a Performance Stock Unit under Section 9.2 hereof, to be paid or distributed based on or conditioned upon the attainment of pre-established business and/or individual Performance Goals over a specified performance period.
- "Person" shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof, including a "group" as defined in Section 13(d) thereof.
- "Plan" means the First Western Financial, Inc. 2016 Omnibus Incentive Plan (As Amended and Restated May 9, 2018) as set forth herein, effective and as may be amended from time to time as provided herein.
- "Public Offering" means the sale of shares of the Common Stock to the public pursuant to an effective registration statement (other than a registration statement on Form S-4 or S-8 or any similar or successor form) filed under the Securities Act in connection with an underwritten offering.
- "Restricted Stock Award" means a grant of shares of Common Stock to an Eligible Person under Section 8 hereof that are issued subject to such vesting and transfer restrictions as the Committee shall determine, and such other conditions, as are set forth in the Plan and the applicable Award Agreement.
- "Restricted Stock Unit" means a contractual right granted to an Eligible Person under Section 9 hereof representing a notional unit interest equal in value to a share of Common Stock to be paid or distributed at such times, and subject to such conditions, as set forth in the Plan and the applicable Award Agreement.
 - "Reporting Person" means a person who is required to file reports under Section 16(a) of the Exchange Act.
- "Retirement" means, with respect to a Retirement Eligible Employee, such Retirement Eligible Employee's termination of employment with the Company and its Subsidiaries due to the Retirement Eligible Individual's voluntary resignation, where all of the following conditions are met: (a) the Retirement Eligible Employee has provided at least one (1) year advanced written notice of resignation to the Company; (b) the Retirement Eligible Employee terminates his or her employment on the date of resignation specified in such written notice of resignation (or such other date as may be mutually agreed in writing by the Company and the Retirement Eligible Employee following the Company's receipt of the written notice of resignation); (c) if the Retirement Eligible Employee is not subject to a non-solicitation agreement as of the date of his or her termination, the Retirement Eligible Employee has executed and delivered to the Company a non-solicitation agreement on a form approved by the Company in its sole discretion; (d) if the Retirement Eligible Employee is not subject to a non-competition agreement as of the date of his or her termination, the Retirement Eligible Employee has entered into a non-competition agreement on a form approved by the Company in its sole discretion; and (e) the Retirement Eligible Employee executes and delivers, and does not revoke, a general release and waiver on a form approved by the Company in its sole discretion, and such general release and waiver becomes binding and irrevocable not later than sixty (60) days following the Retirement Eligible Employee's resignation from the Company and its Subsidiaries.
- "Retirement Eligible Employee" means an employee who has attained age 65 and completed 10 years of employment with Company and its Subsidiaries. For avoidance of doubt, an employee's prior service with an entity

that is acquired (or the assets of which are acquired) by the Company or a Subsidiary shall not count towards an employee's service with the Company and its Subsidiaries for this purpose.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder from time to time.

"Service" means a Participant's employment with the Company or any Subsidiary or a Participant's service as a Non-Employee Director, consultant or other service provider with the Company or any Subsidiary, as applicable.

"Stock Appreciation Right" means a contractual right granted to an Eligible Person under Section 7 hereof entitling such Eligible Person to receive a payment, representing the excess of the Fair Market Value of a share of Common Stock over the base price per share of the right, at such time, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.

"Stock Award" means a grant of shares of Common Stock to an Eligible Person under Section 11 hereof that are issued free of transfer restrictions and forfeiture conditions.

"Stock Option" means a contractual right granted to an Eligible Person under Section 6 hereof to purchase shares of Common Stock at such time and price, and subject to such conditions, as are set forth in the Plan and the applicable Award Agreement.

"Subsidiary" means an entity (whether or not a corporation) that is wholly or majority owned or controlled, directly or indirectly, by the Company or any other affiliate of the Company that is so designated, from time to time, by the Committee, during the period of such affiliated status; provided, however, that with respect to Incentive Stock Options, the term "Subsidiary" shall include only an entity that qualifies under Section 424(f) of the Code as a "subsidiary corporation" with respect to the Company.

"Ten-Percent Stockholder" means an individual who owns more than ten percent (10%) of the total combined voting power of all classes of outstanding stock of the Company, its parent or any of its Subsidiaries. In determining stock ownership, the attribution rules of Section 424(d) of the Code shall be applied.

3. Administration.

- 3.1 *Committee Members*. The Plan shall be administered by a Committee comprised of no fewer than two members of the Board who are appointed by the Board to administer the Plan. To the extent deemed necessary by the Board, each Committee member shall satisfy the requirements for (i) an "independent director" under rules adopted by NASDAQ or any other principal exchange on which the Common Stock is then listed, or (ii) a "nonemployee director" for purposes of such Rule 16b-3 under the Exchange Act. Notwithstanding the foregoing, the mere fact that a Committee member shall fail to qualify under any of the foregoing requirements shall not invalidate any Award made by the Committee which Award is otherwise validly made under the Plan. Neither the Company nor any member of the Committee shall be liable for any action or determination made in good faith by the Committee with respect to the Plan or any Award thereunder. The Board shall have the authority to execute the powers of the Committee under the Plan.
- 3.2 Committee Authority. The Committee shall have all powers and discretion necessary or appropriate to administer the Plan and to control its operation, including, but not limited to, the power to (i) determine the Eligible Persons to whom Awards shall be granted under the Plan and to grant Awards, (ii) prescribe the restrictions, terms and conditions of all Awards, (iii) interpret the Plan and terms of the Awards, (iv) adopt rules for the administration, interpretation and application of the Plan as are consistent therewith, and interpret, amend or revoke any such rules, (v) make all determinations with respect to a Participant's Service and the termination of such Service for purposes of any Award, (vi) correct any defect(s) or omission(s) or reconcile any ambiguity(ies) or inconsistency(ies) in the Plan or any Award thereunder, (vii) make all determinations it deems advisable for the administration of the Plan; (viii) to decide all disputes arising in connection with the Plan and to otherwise supervise the administration of the Plan; (ix) subject to the terms of the Plan, amend the terms of an Award in any manner that is not inconsistent with the Plan, (x) accelerate the vesting or, to the extent applicable, exercisability of any Award at any time (including, but not limited to, upon a Change in Control or upon termination of Service under certain circumstances, as set forth in the Award Agreement or otherwise), and (xi) adopt such procedures and subplans as are necessary or appropriate to permit participation in the Plan by Eligible Person who are foreign nationals or employed outside of the United States. The Committee's determinations under the Plan need not be uniform and may be made by the Committee selectively among Participants and Eligible Persons, whether or not such persons are similarly situated. The Committee shall, in its discretion, consider such factors as it deems relevant in making its

interpretations, determinations and actions under the Plan including, without limitation, the recommendations or advice of any officer or employee of the Company or such attorneys, consultants, accountants or other advisors as it may select. All interpretations, determinations, and actions by the Committee shall be final, conclusive, and binding upon all parties.

3.3 Delegation of Authority. The Committee shall have the right, from time to time, to delegate in writing to one or more officers of the Company the authority of the Committee to grant and determine the terms and conditions of Awards granted under the Plan, subject to any applicable laws. In no event shall any such delegation of authority be permitted with respect to Awards granted to any member of the Board or to any Eligible Person who is subject to Rule 16b-3 under the Exchange Act (as determined in accordance with applicable guidance as of the applicable date of determination). The Committee shall also be permitted to delegate, to any appropriate officer or employee of the Company, responsibility for performing certain ministerial functions under the Plan. In the event that the Committee's authority is delegated to officers or employees in accordance with the foregoing, all provisions of the Plan relating to the Committee shall be interpreted in a manner consistent with the foregoing by treating any such reference as a reference to such officer or employee for such purpose. Any action undertaken in accordance with the Committee's delegation of authority hereunder shall have the same force and effect as if such action were undertaken directly by the Committee and shall be deemed for all purposes of the Plan to have been taken by the Committee.

4. Shares Subject to the Plan.

- 4.1 *Number of Shares Reserved*. Subject to adjustment as provided in Section 4.4 hereof, the total number of shares of Common Stock that are reserved for issuance under the Plan shall be 200,000 plus any shares remaining available for grant under the First Western Financial, Inc. 2008 Stock Incentive Plan (the "2008 Plan") as of the Original Effective Date (collectively, the "Share Reserve"). Each share of Common Stock subject to an Award shall reduce the Share Reserve by one share; provided that Awards that are required to be paid in cash pursuant to their terms shall not reduce the Share Reserve. Any shares of Common Stock delivered under the Plan shall consist of authorized and unissued shares or treasury shares.
- 4.2 Share Replenishment. To the extent that an Award granted under this Plan is canceled, expired, forfeited, surrendered, settled by delivery of fewer shares than the number underlying the Award, settled in cash or otherwise terminated without delivery of the shares to the Participant, the shares of Common Stock retained by or returned to the Company will (i) not be deemed to have been delivered under the Plan, (ii) be available for future Awards under the Plan, and (iii) increase the Share Reserve by one share for each share that is retained by or returned to the Company. Shares that are (x) withheld from an Award in payment of the exercise or purchase price or taxes relating to such an Award or (y) not issued or delivered as a result of the net settlement of an outstanding Stock Option or Stock Appreciation Right will (i) not be deemed to have been delivered under the Plan, (ii) be available for future Awards under the Plan, (iii) increase the Share Reserve by one Share for each Share that is retained by or returned to the Company and (iv) continue to be counted as outstanding for purposes of determining whether the Award limit specified in Section 4.3 has been attained. In addition to the foregoing, any shares that become available for issuance pursuant to Section 4 of the 2008 Plan as a result of the forfeiture, cancellation or termination for no consideration of an award under the 2008 Plan will (i) not be available for future awards under the 2008 Plan, (ii) be available for future Awards under this Plan, and (iii) increase the Share Reserve by one share for each share that is retained by or returned to the Company, subject to a maximum of 1,500,000 thereof
- 4.3 Awards Granted to Non-Employee Directors. The maximum number of shares of Common Stock that may be subject to Stock Options, Stock Appreciation Rights, Restricted Stock Awards, Restricted Stock Units and Stock Awards granted to any Non-Employee Director during any calendar year shall be limited to 100,000 shares of Common Stock for all such Award types in the aggregate (subject to adjustment as provided in Section 4.4 hereof). If an Award is settled in cash, the number of shares of Common Stock on which the Award is based shall not count toward the individual share limit set forth in this Section 4.3 but shall count against the annual Cash Performance Award limit set forth in Section 10.7.
- 4.4 Adjustments. If there shall occur any change with respect to the outstanding shares of Common Stock by reason of any recapitalization, reclassification, stock dividend, extraordinary cash dividend, stock split, reverse stock split or other distribution with respect to the shares of Common Stock or any merger, reorganization, consolidation, combination, spin-off, stock purchase or other similar corporate change or any other change affecting the Common Stock (other than regular cash dividends to shareholders of the Company), the Committee shall, in the manner and to the extent it considers appropriate and equitable to the Participants and consistent with the terms of the Plan, cause an adjustment to be made to (i) the maximum number of shares of Common Stock provided in Sections 4.1 and 4.3 hereof (including the maximum number of shares of Common Stock that may become payable to a Non-Employee Director provided in Sections 4.3 hereof), (ii) the number and kind of shares of Common Stock, units or other rights subject to then outstanding Awards, (iii) the exercise or base price for each

- 5 -

share or unit or other right subject to then outstanding Awards, and (iv) any other terms of an Award that are affected by the event. Notwithstanding the foregoing, (a) any such adjustments shall, to the extent necessary, be made in a manner consistent with the requirements of Section 409A of the Code and (b) in the case of Incentive Stock Options, any such adjustments shall, to the extent practicable, be made in a manner consistent with the requirements of Section 424(a) of the Code.

5. Eligibility and Awards.

- 5.1 Designation of Participants. Any Eligible Person may be selected by the Committee to receive an Award and become a Participant. The Committee has the authority, in its discretion, to determine and designate from time to time those Eligible Persons who are to be granted Awards, the types of Awards to be granted, the number of shares of Common Stock or units subject to Awards to be granted and the terms and conditions of such Awards consistent with the terms of the Plan, and to grant any such Awards. In selecting Eligible Persons to be Participants, and in determining the type and amount of Awards to be granted under the Plan, the Committee shall consider any and all factors that it deems relevant or appropriate. Designation of a Participant in any year shall not require the Committee to designate such person to receive an Award in any other year or, once designated, to receive the same type or amount of Award as granted to such Participant in any other year.
- 5.2 Determination of Awards. The Committee shall determine the terms and conditions of all Awards granted to Participants in accordance with its authority under Section 3.2 hereof. An Award may consist of one type of right or benefit hereunder or of two or more such rights or benefits granted in tandem.
- 5.3 *Award Agreements*. Each Award granted to an Eligible Person shall be represented by an Award Agreement. The terms of all Awards under the Plan, as determined by the Committee, will be set forth in each individual Award Agreement as described in Section 16.2 hereof.

6. Stock Options.

- 6.1 *Grant of Stock Options*. A Stock Option may be granted to any Eligible Person selected by the Committee, except that an Incentive Stock Option may only be granted to an Eligible Person satisfying the conditions of Section 6.7(a) hereof. Each Stock Option shall be designated on the Date of Grant, in the discretion of the Committee, as an Incentive Stock Option or as a Nonqualified Stock Option.
- 6.2 Exercise Price. The exercise price per share of a Stock Option shall not be less than one hundred percent (100%) of the Fair Market Value of a share of Common Stock on the Date of Grant. The Committee may in its discretion specify an exercise price per share that is higher than the Fair Market Value of a share of Common Stock on the Date of Grant.
- 6.3 *Vesting of Stock Options*. The Committee shall, in its discretion, prescribe the time or times at which or the conditions upon which, a Stock Option or portion thereof shall become vested and/or exercisable. The requirements for vesting and exercisability of a Stock Option may be based on the continued Service of the Participant with the Company or a Subsidiary for a specified time period (or periods), on the attainment of a specified Performance Goal(s) or on such other terms and conditions as approved by the Committee in its discretion, all as set forth in the Award Agreement. If the vesting requirements of a Stock Option are not satisfied, the Award shall be forfeited.
- 6.4 Term of Stock Options. The Committee shall in its discretion prescribe in an Award Agreement the period during which a vested Stock Option may be exercised; provided, however, that the maximum term of a Stock Option shall be ten (10) years from the Date of Grant. The Committee may provide that a Stock Option will cease to be exercisable upon or at the end of a specified time period following a termination of Service for any reason as set forth in the Award Agreement or otherwise. Subject to Section 409A of the Code and the provisions of this Section 6, the Committee may extend at any time the period in which a Stock Option may be exercised.
- 6.5 Stock Option Exercise; Tax Withholding. Subject to such terms and conditions as specified in an Award Agreement, a Stock Option may be exercised in whole or in part at any time during the term thereof by notice in the form required by the Company, together with payment of the aggregate exercise price and applicable withholding tax. Payment of the exercise price may be made: (i) in cash or by cash equivalent acceptable to the Committee or (ii) to the extent permitted by the Committee in its sole discretion and set forth in the Award Agreement or otherwise (including by a policy or resolution of the Committee), (A) in shares of Common Stock valued at the Fair Market Value of such shares on the date of exercise, (B) through an open-market, broker-assisted sales transaction pursuant to which the Company is promptly delivered the amount of proceeds necessary to satisfy the exercise price, (C) by reducing the number of shares of Common Stock otherwise deliverable upon the exercise

of the Stock Option by the number of shares of Common Stock having a Fair Market Value on the date of exercise equal to the exercise price, (D) by a combination of the methods described above or (E) by such other method as may be approved by the Committee and set forth in the Award Agreement. In addition to and at the time of payment of the exercise price, the Participant shall pay to the Company the full amount of any and all applicable income tax, employment tax and other amounts required to be withheld in connection with such exercise, payable under such of the methods described above for the payment of the exercise price as may be approved by the Committee and set forth in the Award Agreement.

6.6 *Limited Transferability of Nonqualified Stock Options*. All Nonqualified Stock Options shall be exercisable during the Participant's lifetime only by the Participant or by the Participant's guardian or legal representative. The Nonqualified Stock Options and the rights and privileges conferred thereby shall be nontransferable except as otherwise provided in Section 16.3 hereof.

6.7 Additional Rules for Incentive Stock Options.

- (a) *Eligibility*. An Incentive Stock Option may only be granted to an Eligible Person who is considered an employee for purposes of Treasury Regulation § 1.421-1(h) with respect to the Company or any Subsidiary that qualifies as a "subsidiary corporation" with respect to the Company for purposes of Section 424(f) of the Code.
- (b) Annual Limits. No Incentive Stock Option shall be granted to a Participant as a result of which the aggregate Fair Market Value (determined as of the Date of Grant) of the Common Stock with respect to which incentive stock options under Section 422 of the Code are exercisable for the first time in any calendar year under the Plan and any other stock option plans of the Company or any subsidiary or parent corporation, would exceed \$100,000, determined in accordance with Section 422(d) of the Code. This limitation shall be applied by taking such incentive stock options into account in the order in which they were granted.
- (c) Additional Limitations. In the case of any Incentive Stock Option granted to an Eligible Person who owns, either directly or indirectly (taking into account the attribution rules contained in Section 424(d) of the Code), stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Subsidiary, the exercise price shall not be less than one hundred ten percent (110%) of the Fair Market Value of a share of Common Stock on the Date of Grant and the maximum term shall be five (5) years.
- (d) *Termination of Employment*. An Award of an Incentive Stock Option may provide that such Stock Option may be exercised not later than (i) three (3) months following termination of employment of the Participant with the Company and all Subsidiaries (other than as set forth in clause (ii) of this Section 6.7(d)) or (ii) one year following termination of employment of the Participant with the Company and all Subsidiaries due to death or permanent and total disability within the meaning of Section 22(e)(3) of the Code, in each case as and to the extent determined by the Committee to comply with the requirements of Section 422 of the Code.
- (e) Other Terms and Conditions; Nontransferability. No Incentive Stock Options granted under the Plan may be granted more than ten (10) years following the date that the Plan is adopted or the date that the Plan is approved by the Company's stockholders, whichever is earlier. The Award Agreement representing any Incentive Stock Option granted hereunder shall contain such additional terms and conditions, not inconsistent with the terms of the Plan, as are deemed necessary or desirable by the Committee, which terms, together with the terms of the Plan, shall be intended and interpreted to cause such Incentive Stock Option to qualify as an "incentive stock option" under Section 422 of the Code. A Stock Option that is granted as an Incentive Stock Option shall, to the extent it fails to qualify as an "incentive stock option" under the Code, be treated as a Nonqualified Stock Option. An Incentive Stock Option shall by its terms be nontransferable other than by will or by the laws of descent and distribution, and shall be exercisable during the lifetime of a Participant only by such Participant.
- (f) Disqualifying Dispositions. If shares of Common Stock acquired by exercise of an Incentive Stock Option are disposed of within two years following the Date of Grant or one year following the transfer of such shares to the Participant upon exercise, the Participant shall, promptly following such disposition, notify the Company in writing of the date and terms of such disposition and provide such other information regarding the disposition as the Company may reasonably require.
- 6.8 *Repricing Prohibited*. Subject to the anti-dilution adjustment provisions contained in Section 4.4 hereof, without the prior approval of the Company's stockholders, neither the Committee nor the Board shall cancel a Stock Option when the exercise price per share exceeds the Fair Market Value of one share of Common Stock in exchange for cash or another Award (other than in connection with a Change in Control) or cause the cancellation, substitution or amendment of a Stock Option that would have the effect of reducing the exercise price of such a Stock Option previously granted under the Plan or otherwise approve any modification to such a Stock Option that

would be treated as a "repricing" under the then applicable rules, regulations or listing requirements adopted by NASDAQ or any other principal exchange on which the Common Stock is then listed.

6.9 *Dividend Equivalent Rights*. Subject to the anti-dilution adjustment provisions contained in Section 4.4 hereof, dividends shall not be paid with respect to Stock Options. Dividend equivalent rights shall be granted with respect to the shares of Common Stock subject to Stock Options to the extent permitted by the Committee or set forth in the Award Agreement.

7. Stock Appreciation Rights.

- 7.1 *Grant of Stock Appreciation Rights*. Stock Appreciation Rights may be granted to any Eligible Person selected by the Committee. Stock Appreciation Rights may be granted on a basis that allows for the exercise of the right by the Participant or that provides for the automatic payment of the right upon a specified date or event. Stock Appreciation Rights shall be non-transferable, except as provided in Section 16.3 hereof.
- 7.2 Stand-Alone and Tandem Stock Appreciation Rights. A Stock Appreciation Right may be granted without any related Stock Option, or may be granted in tandem with a Stock Option, either on the Date of Grant or at any time thereafter during the term of the Stock Option. The Committee shall in its discretion provide in an Award Agreement the time or times at which or the conditions upon which, a Stock Appreciation Right or portion thereof shall become vested and/or exercisable. The requirements for vesting and exercisability of a Stock Appreciation Right may be based on the continued Service of a Participant with the Company or a Subsidiary for a specified time period (or periods), on the attainment of a specified Performance Goal(s) or on such other terms and conditions as approved by the Committee in its discretion. If the vesting requirements of a Stock Appreciation Right are not satisfied, the Award shall be forfeited. A Stock Appreciation Right will be exercisable or payable at such time or times as determined by the Committee; provided, that the maximum term of a Stock Appreciation Right shall be ten (10) years from the Date of Grant. The Committee may provide that a Stock Appreciation Right will cease to be exercisable upon or at the end of a period following a termination of Service for any reason. The base price of a Stock Appreciation Right granted without any related Stock Option shall be determined by the Committee in its discretion; provided, however, that the base price per share of any such stand-alone Stock Appreciation Right shall not be less than one hundred percent (100%) of the Fair Market Value of a share of Common Stock on the Date of Grant.
- 7.3 Payment of Stock Appreciation Rights. A Stock Appreciation Right will entitle the holder, upon exercise or other payment of the Stock Appreciation Right, as applicable, to receive an amount determined by multiplying: (i) the excess of the Fair Market Value of a share of Common Stock on the date of exercise or payment of the Stock Appreciation Right over the base price of such Stock Appreciation Right, by (ii) the number of shares as to which such Stock Appreciation Right is exercised or paid. Payment of the amount determined under the foregoing may be made, as approved by the Committee and set forth in the Award Agreement, in shares of Common Stock valued at their Fair Market Value on the date of exercise or payment, in cash or in a combination of shares of Common Stock and cash, subject to applicable tax withholding requirements.
- 7.4 Repricing Prohibited. Subject to the anti-dilution adjustment provisions contained in Section 4.4 hereof, without the prior approval of the Company's stockholders, neither the Committee nor the Board shall cancel a Stock Appreciation Right when the base price per share exceeds the Fair Market Value of one share of Common Stock in exchange for cash or another Award (other than in connection with a Change in Control) or cause the cancellation, substitution or amendment of a Stock Appreciation Right that would have the effect of reducing the base price of such a Stock Appreciation Right previously granted under the Plan or otherwise approve any modification to such Stock Appreciation Right that would be treated as a "repricing" under the then applicable rules, regulations or listing requirements adopted by NASDAQ or any other principal exchange on which the Common Stock is then listed.
- 7.5 *Dividend Equivalent Rights*. Subject to the anti-dilution adjustment provisions contained in Section 4.4 hereof, dividends shall not be paid with respect to Stock Appreciation Rights. Dividend equivalent rights shall be granted with respect to the shares of Common Stock subject to Stock Appreciation Rights to the extent permitted by the Committee or set forth in the Award Agreement.

8. Restricted Stock Awards.

8.1 *Grant of Restricted Stock Awards*. A Restricted Stock Award may be granted to any Eligible Person selected by the Committee. The Committee may require the payment by the Participant of a specified purchase price in connection with any Restricted Stock Award.

- 8.2 *Vesting Requirements*. The restrictions imposed on shares granted under a Restricted Stock Award shall lapse in accordance with the vesting requirements specified by the Committee in the Award Agreement. The requirements for vesting of a Restricted Stock Award may be based on the continued Service of the Participant with the Company or a Subsidiary for a specified time period (or periods), on the attainment of a specified Performance Goal(s), or on such other terms and conditions as approved by the Committee in its discretion. If the vesting requirements of a Restricted Stock Award shall not be satisfied or, if applicable, the Performance Goal(s) with respect to such Restricted Stock Award are not attained, the Award shall be forfeited and the shares of Stock subject to the Award shall be returned to the Company.
- 8.3 *Transfer Restrictions*. Shares granted under any Restricted Stock Award may not be transferred, assigned or subject to any encumbrance, pledge or charge until all applicable restrictions are removed or have expired, except as provided in Section 16.3 hereof. Failure to satisfy any applicable restrictions shall result in the subject shares of the Restricted Stock Award being forfeited and returned to the Company. The Committee may require in an Award Agreement that certificates (if any) representing the shares granted under a Restricted Stock Award bear a legend making appropriate reference to the restrictions imposed, and that certificates (if any) representing the shares granted or sold under a Restricted Stock Award will remain in the physical custody of an escrow holder until all restrictions are removed or have expired.
- 8.4 Rights as Stockholder. Subject to the foregoing provisions of this Section 8 and the applicable Award Agreement, the Participant shall have all rights of a stockholder with respect to the shares granted to the Participant under a Restricted Stock Award, including the right to vote the shares and receive all dividends and other distributions paid or made with respect thereto, unless otherwise provided in the applicable Award Agreement or the Committee determines otherwise at the time the Restricted Stock Award is granted. The Committee may provide in an Award Agreement for the payment of dividends and distributions to the Participant at such times as paid to stockholders generally, at the times of vesting or other payment of the Restricted Stock Award or otherwise.
- 8.5 Section 83(b) Election. If a Participant makes an election pursuant to Section 83(b) of the Code with respect to a Restricted Stock Award, the Participant shall file, within thirty (30) days following the Date of Grant, a copy of such election with the Company and with the Internal Revenue Service, in accordance with the regulations under Section 83 of the Code. The Committee may provide in an Award Agreement that the Restricted Stock Award is conditioned upon the Participant's making or refraining from making an election with respect to the Award under Section 83(b) of the Code.

9. Restricted Stock Units.

- 9.1 *Grant of Restricted Stock Units*. A Restricted Stock Unit may be granted to any Eligible Person selected by the Committee. The value of each Restricted Stock Unit is equal to the Fair Market Value of the Common Stock on the applicable date or time period of determination, as specified by the Committee. Restricted Stock Units shall be subject to such restrictions and conditions as the Committee shall determine. In addition, a Restricted Stock Unit may be designated as a "Performance Stock Unit", the vesting requirements of which may be based, in whole or in part, on the attainment of preestablished business and/or individual Performance Goal(s) over a specified performance period as approved by the Committee in its discretion. Restricted Stock Units shall be non-transferable, except as provided in Section 16.3 hereof.
- 9.2 Vesting of Restricted Stock Units. On the Date of Grant, the Committee shall, in its discretion, determine any vesting requirements with respect to Restricted Stock Units, which shall be set forth in the Award Agreement. The requirements for vesting of a Restricted Stock Unit may be based on the continued Service of the Participant with the Company or a Subsidiary for a specified time period (or periods) or on such other terms and conditions as approved by the Committee (including Performance Goal(s)) in its discretion. If the vesting requirements of a Restricted Stock Units Award are not satisfied, the Award shall be forfeited.
- 9.3 *Payment of Restricted Stock Units*. Restricted Stock Units shall become payable to a Participant at the time or times determined by the Committee and set forth in the Award Agreement, which may be upon or following the vesting of the Award. Payment of a Restricted Stock Unit may be made, as approved by the Committee and set forth in the Award Agreement, in cash or in shares of Common Stock or in a combination thereof, subject to applicable tax withholding requirements. Any cash payment of a Restricted Stock Unit shall be made based upon the Fair Market Value of the Common Stock
- 9.4 Dividend Equivalent Rights. Subject to the anti-dilution adjustment provisions contained in Section 4.4 hereof, Restricted Stock Units may or may not, in the discretion of the Committee, be granted together with a dividend equivalent right with respect to the shares of Common Stock subject to the Award, which may be accumulated and may be deemed reinvested in additional Restricted Stock Units or may be accumulated in cash, as determined by the Committee in its discretion. Dividend equivalent rights will be paid at such times as determined

by the Committee in its discretion (including without limitation at the times paid to stockholders generally or at the times of vesting or payment of the Restricted Stock Unit). Dividend equivalent rights may be subject to forfeiture under the same conditions as apply to the underlying Restricted Stock Units.

9.5 *No Rights as Stockholder*. The Participant shall not have any rights as a stockholder with respect to the shares subject to a Restricted Stock Unit until such time as shares of Common Stock are delivered to the Participant pursuant to the terms of the Award Agreement.

10. Cash Performance Awards and Performance Criteria.

- 10.1 *Grant of Cash Performance Awards*. A Cash Performance Award may be granted to any Eligible Person selected by the Committee. Payment amounts may be based on specified levels of attainment with respect to the Performance Goals, including, if applicable, specified threshold, target and maximum performance levels. The requirements for payment may be also based upon the continued Service of the Participant with the Company or a Subsidiary during the respective performance period and on such other conditions as determined by the Committee and set forth in an Award Agreement. With respect to Cash Performance Awards and other Awards the vesting and/or payout of which are tied to one or more Performance Goals, the Committee will determine the duration of the performance period, the Performance Criteria, the applicable Performance Goals relating to the Performance Criteria, and the amount and terms of payment and/or vesting upon achievement of the Performance Goals. Cash Performance Awards shall be non-transferable, except as provided in Section 16.3 hereof.
- 10.2 *Award Agreements*. Each Cash Performance Award shall be evidenced by an Award Agreement that shall specify the performance period and such other terms and conditions as the Committee, in its discretion, shall determine. The Committee may accelerate the vesting of a Cash Performance Award upon a Change in Control or termination of Service under certain circumstances, as set forth in the Award Agreement.
- 10.3 Performance Criteria. For purposes of Cash Performance Awards, Performance Stock Units and other Awards, the Performance Criteria shall be one or any combination of any performance measure selected by the Committee, which may include, but shall not be limited to, any the following performance measures for the Company or any identified Subsidiary or business unit: (a) net earnings; (b) earnings per share; (c) net debt; (d) revenue or sales growth; (e) net or operating income; (f) net operating profit; (g) return measures (including, but not limited to, return on assets, capital, equity or sales); (h) cash flow (including, but not limited to, operating cash flow, distributable cash flow and free cash flow); (i) earnings before or after taxes, interest, depreciation, amortization and/or rent; (j) share price (including, but not limited to growth measures and total stockholder return); (k) expense control or loss management; (l) customer satisfaction; (m) market share; (n) economic value added; (o) working capital; (p) the formation of joint ventures or the completion of other corporate transactions; (q) gross or net profit margins; (r) revenue mix; (s) operating efficiency; (t) product diversification; (u) market penetration; (v) measurable achievement in quality, operation or compliance initiatives; (w) quarterly dividends or distributions; (x) employee retention or turnover; (y) assets under management; (z) return on average tangible common equity (defined as a ratio, the numerator of which is income before amortization of intangibles, and the denominator of which is average tangible common equity); (aa) "efficiency ratio" determined as the ratio of total noninterest operating expenses (less amortization of intangibles) divided by total tax-equivalent revenues; (bb) "burden ratio" determined as the ratio of total noninterest operating expenses (less amortization of intangibles) less noninterest income over total tax-equivalent revenues; (cc) noninterest income to total revenue ratio; (dd) noninterest income to average assets ratio; (ee) net interest margin; (ff) net interest margin — tax equivalent; (gg) ratio of noninterest expense (less amortization of intangibles) to average assets; (hh) credit quality measures (including non-performing asset ratio, net charge-off ratio, ratio of allowance to non-performing loans, and classified assets as a percentage of tier 1 capital plus allowance of loan losses); (ii) noninterest bearing deposits as a percentage of total deposits; (jj) brokered deposits as a percentage of total deposits; (kk) loan growth; (ll) deposit growth; (mm) core deposit growth (defined as deposit growth excluding time deposits); (nn) noninterest income growth; (oo) budgeted noninterest income; (pp) yield on earning assets; (qq) tax-equivalent yield on earning assets; (rr) loan yield; (ss) tax-equivalent loan yield; (tt) cost of funds; (uu) net interest income; (vv) pre-provision, pre-tax income; (ww) regulatory criteria or measures, including compliance with a regulatory enforcement action or regulation (including a regulation that would exclude certain income from being included in any of the above criteria); (xx) other strategic milestones based on objective criteria established by the Company provided that, with respect to Covered Employees, such strategic milestones must be approved by the stockholders of the Company prior to the payment of any Cash Performance Awards, Performance Stock Units and other Awards intended to qualify as "performance-based compensation"; or (yy) any combination of or a specified increase in any of the foregoing. Each of the Performance Criteria shall be applied and interpreted in accordance with an objective formula or standard established by the Committee at the time the applicable Award is granted including, without limitation, GAAP.

10.4 *Performance Goals*. For purposes of Cash Performance Awards and other Awards the vesting and/or payout of which are tied to one or more Performance Goals, the "Performance Goals" shall be the levels of

achievement relating to the Performance Criteria selected by the Committee for the Award. The Performance Goals shall be written and shall be expressed as an objective formula or standard that precludes discretion to increase the amount of compensation payable that would otherwise be due upon attainment of the goal. The Performance Goals may be (but are not required to be) applied on an absolute basis or relative to an identified index, peer group, or one or more competitors or other companies (including particular business segments or divisions or such companies). The Performance Goals need not be the same for all Participants.

- 10.5 *Adjustments*. At the time that an Award is granted, the Committee may provide for the Performance Goals or the manner in which performance will be measured against the Performance Goals to be adjusted in such objective manner as it deems appropriate, including, without limitation, adjustments to reflect charges for restructurings, non-operating income, the impact of corporate transactions or discontinued operations, extraordinary and other unusual or non-recurring items and the cumulative effects of accounting or tax law changes. In addition, with respect to a Participant hired or promoted following the beginning of a performance period, the Committee may determine to prorate the Performance Goals and/or the amount of any payment in respect of such Participant's Cash Performance Awards for the partial performance period.
- 10.6 *Negative Discretion*. Notwithstanding anything else contained in the Plan to the contrary, the Committee shall, to the extent provided in an Award Agreement, have the right, in its discretion, (i) to reduce or eliminate the amount otherwise payable to any Participant under an Award and (ii) to establish rules or procedures that have the effect of limiting the amount payable to any Participant to an amount that is less than the amount that otherwise would be payable under an Award. The Committee may exercise such discretion in a non-uniform manner among Participants.
- 10.7 *Certification*. Following the conclusion of the performance period of a Cash Performance Award or other Award the vesting and/or payout of which is tied to one or more Performance Goals, the Committee shall certify in writing whether the Performance Goals for that performance period have been achieved, or certify the degree of achievement, if applicable.
- 10.8 *Payment*. Upon certification of the Performance Goals for a Cash Performance Award or other Award, the Committee shall determine the level of vesting or amount of payment to the Participant pursuant to the Award, if any. Notwithstanding the foregoing, Cash Performance Awards may be paid, at the discretion of the Committee, in any combination of cash or shares of Common Stock, based upon the Fair Market Value of such shares at the time of payment.

11. Stock Awards.

- 11.1 *Grant of Stock Awards*. A Stock Award may be granted to any Eligible Person selected by the Committee. A Stock Award may be granted for past Services, in lieu of bonus or other cash compensation, as directors' compensation or for any other valid purpose as determined by the Committee. The Committee shall determine the terms and conditions of such Awards, and such Awards may be made without vesting requirements. In addition, the Committee may, in connection with any Stock Award, require the payment of a specified purchase price.
- 11.2 *Rights as Stockholder*. Subject to the foregoing provisions of this Section 11 and the applicable Award Agreement, upon the issuance of the Common Stock under a Stock Award the Participant shall have all rights of a stockholder with respect to the shares of Common Stock, including the right to vote the shares and receive all dividends and other distributions paid or made with respect thereto.

12. Change in Control.

12.1 Effect on Awards. Upon the occurrence of a Change in Control, unless otherwise provided in the Award Agreement, the Committee is authorized (but not obligated) to make adjustments in the terms and conditions of outstanding Awards, including without limitation the following (or any combination thereof): (a) continuation or assumption of such outstanding Awards under the Plan by the Company (if it is the surviving company or corporation) or by the surviving company or corporation or its parent; (b) substitution by the surviving company or corporation or its parent of awards with substantially the same terms for outstanding Awards (with appropriate adjustments to the type of consideration payable upon settlement of the Awards); (c) accelerated exercisability, vesting and/or payment under outstanding Awards immediately prior to or upon the occurrence of such event or upon a termination of employment following such event; and (d) if all or substantially all of the Company's outstanding shares of Common Stock transferred in exchange for cash consideration in connection with such Change in Control: (i) upon written notice, provide that any outstanding Stock Options and Stock Appreciation Rights are exercisable during a reasonable period of time immediately prior to the scheduled consummation of the event or such other reasonable period as determined by the Committee (contingent upon the consummation of the event), and

at the end of such period, such Stock Options and Stock Appreciation Rights shall terminate to the extent not so exercised within the relevant period; and (ii) cancellation of all or any portion of outstanding Awards for fair value (in the form of cash, shares, other property or any combination thereof) as determined in the sole discretion of the Committee; <u>provided</u>, <u>that</u>, in the case of Stock Options and Stock Appreciation Rights, the fair value may equal the excess, if any, of the value of the consideration to be paid in the Change in Control transaction to holders of shares of Common Stock (or, if no such consideration is paid, Fair Market Value of the shares of Common Stock subject to such outstanding Awards or portion thereof being canceled) over the aggregate exercise or base price, as applicable, with respect to such Awards or portion thereof being canceled, or if no such excess, zero.

12.2 Definition of Change in Control. Unless otherwise defined in an Award Agreement, "Change in Control" shall mean the occurrence of one of the following events:

(a) Any Person, becomes the Beneficial Owner, directly or indirectly, of more than fifty percent (50%) of the combined voting power, excluding any Person who holds fifty percent (50%) or more of the voting power on the Effective Date of the Plan (the "Initial Owners"), of the then outstanding voting securities of the Company entitled to vote generally in the election of its directors (the "Outstanding Company Voting Securities") including by way of merger, consolidation or otherwise; provided, however, that for purposes of this definition, the following acquisitions shall not constitute a Change in Control: (i) any acquisition of voting securities of the Company directly from the Company, including without limitation, a public offering of securities; (ii) any acquisition by the Company or any of its Subsidiaries of Outstanding Company Voting Securities, including an acquisition by any employee benefit plan or related trust sponsored or maintained by the Company or any of its Subsidiaries; or (iii) any acquisition after which the Initial Owners and their affiliates remain the Beneficial Owners of more Outstanding Voting Securities than any other Person.

(b) Consummation of a reorganization, merger, or consolidation to which the Company is a party or a sale or other disposition of all or substantially all of the assets of the Company (a "Business Combination"), unless, following such Business Combination: (i) any individuals and entities that were the Beneficial Owners of Outstanding Company Voting Securities immediately prior to such Business Combination are the Beneficial Owners, directly or indirectly, of more than fifty percent (50%) of the combined voting power of the outstanding voting securities entitled to vote generally in the election of directors (or election of members of a comparable governing body) of the entity resulting from the Business Combination (including, without limitation, an entity which as a result of such transaction owns all or substantially all of the Company or all or substantially all of the Company's assets either directly or through one or more Subsidiaries) (the "Successor Entity") in substantially the same proportions as their ownership immediately prior to such Business Combination; (ii) no Person (excluding any Successor Entity or any employee benefit plan or related trust of the Company, such Successor Entity, or any of their Subsidiaries) is the Beneficial Owner, directly or indirectly, of more than fifty percent (50%) of the Company, such Successor Entity, or any of their Subsidiaries) is the Beneficial Owner, directly or indirectly, of more than fifty percent (50%) of the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors (or comparable governing body) of the Successor Entity, except to the extent that such ownership existed prior to the Business Combination; and (iii) at least a majority of the members of the board of directors (or comparable governing body) of the Successor Entity were Incumbent Directors (including persons deemed to be Incumbent Directors) at the time of the execution of the initial agreement or of the Board providing for such Busine

Notwithstanding the foregoing, to the extent necessary to comply with Section 409A of the Code with respect to the payment of "nonqualified deferred compensation," "Change in Control" shall be limited to a "change in control event" as defined under Section 409A of the Code. For the avoidance of doubt, neither a Public Offering nor any changes to the size or members of the Board in connection with or as a result of a Public Offering shall constitute or be deemed to result in a Change in Control.

13. Forfeiture Events.

13.1 *General*. The Committee may specify in an Award Agreement at the time of the Award that the Participant's rights, payments and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture or recoupment upon the occurrence of certain specified events, in addition to any otherwise applicable vesting or performance conditions of an Award. Such events may include, but shall not be limited to, termination of Service for Cause, violation of material Company policies, breach of noncompetition, confidentiality or other restrictive covenants that may apply to the Participant or other similar conduct by the Participant that is detrimental to the business or reputation of the Company.

13.2 Termination for Cause.

(a) *Treatment of Awards*. Unless otherwise provided by the Committee and set forth in an Award Agreement, if (i) a Participant's Service with the Company or any Subsidiary shall be terminated for Cause or (ii)

- 12 -

within one (1) year following termination of Service for any other reason, the Committee determines in its discretion that, after termination, the Participant breached any of the material terms contained in any non-competition agreement, confidentiality agreement or similar restrictive covenant agreement to which such Participant is a party, such Participant's rights, payments and benefits with respect to an Award shall be subject to cancellation, forfeiture and/or recoupment, as provided in Section 13.3 below. The Company shall have the power to determine whether the Participant has been terminated for Cause, the date upon which such termination for Cause occurs and whether the Participant engaged in conduct that violated any continuing obligation or duty of the Participant in respect of the Company or any Subsidiary. Any such determination shall be final, conclusive and binding upon all Persons. In addition, if the Company shall reasonably determine that a Participant has committed or may have committed any act which could constitute the basis for a termination of such Participant's Service for Cause or violates any continuing obligation or duty of the Participant in respect of the Company or any Subsidiary, the Company may suspend the Participant's rights to exercise any Stock Option or Stock Appreciation Right, receive any payment or vest in any right with respect to any Award pending a determination by the Company of whether an act or omission could constitute the basis for a termination for Cause as provided in this Section 13.2.

- (b) Definition of Cause. Unless otherwise defined in an Award Agreement, "Cause" shall mean:
- (i) the Participant has committed a deliberate and premeditated act against the interests of the Company including, without limitation: an act of fraud, embezzlement, misappropriation or breach of fiduciary duty against the Company, including, but not limited to, the offer, payment, solicitation or acceptance of any unlawful bribe or kickback with respect to the Company's business; or
- (ii) the Participant has been convicted by a court of competent jurisdiction of, or pleaded guilty or nolo contendere to, any felony or any crime involving moral turpitude; or
- (iii)the Participant has failed to perform or neglected the material duties incident to his employment or other engagement with the Company on a regular basis, and such refusal or failure shall have continued for a period of twenty (20) days after written notice to the Participant specifying such refusal or failure in reasonable detail; or
- (iv)the Participant has been chronically absent from work (excluding vacations, illnesses, Disability or leaves of absence approved by the Company); or
- (v) the Participant has refused, after explicit written notice, to obey any lawful resolution of or direction by the Board which is consistent with the duties incident to his employment or other engagement with the Company and such refusal continues for more than twenty (20) days after written notice is given to the Participant specifying such refusal in reasonable detail; or
- (vi)the Participant has breached any of the material terms contained in any employment agreement, non-competition agreement, confidentiality agreement or similar type of agreement to which such Participant is a party; or
- (vii)the Participant has engaged in (x) the unlawful use (including being under the influence) or possession of illegal drugs on the Company's premises or (y) habitual drunkenness on the Company's premises.

Any voluntary termination of employment or other engagement by the Participant in anticipation of an involuntary termination of the Participant's Service for Cause shall be deemed to be a termination for "Cause." Notwithstanding the foregoing, in the event that a Participant is party to an employment, severance or similar agreement with the Company or any of its affiliates and such agreement contains a definition of "Cause," the definition of "Cause" set forth above shall be deemed replaced and superseded, with respect to such Participant, by the definition of "Cause" used in such employment, severance or similar agreement.

13.3 Right of Recapture.

(a) *General*. If at any time within one (1) year (or such longer time specified in an Award Agreement or other agreement with a Participant) after the date on which a Participant exercises a Stock Option or Stock Appreciation Right or on which a Stock Award, Restricted Stock Award or Restricted Stock Unit vests or becomes payable or on which a Cash Performance Award is paid to a Participant, or on which income otherwise is realized by a Participant in connection with an Award, (i) a Participant's Service is terminated for Cause or (ii) after a Participant's Service otherwise terminates for any other reason, the Committee determines in its discretion that, after termination, the Participant breached any of the material terms contained in any non-competition agreement, confidentiality agreement or similar restrictive covenant agreement to which such Participant is a party, then any

gain realized by the Participant from the exercise, vesting, payment or other realization of income by the Participant in connection with an Award, shall be paid by the Participant to the Company upon notice from the Company, subject to applicable state law. Such gain shall be determined as of the date or dates on which the gain is realized by the Participant, without regard to any subsequent change in the Fair Market Value of a share of Common Stock. To the extent not otherwise prohibited by law, the Company shall have the right to offset such gain against any amounts otherwise owed to the Participant by the Company (whether as wages, vacation pay or pursuant to any benefit plan or other compensatory arrangement).

- (b) Accounting Restatement. If a Participant receives compensation pursuant to a performance-based Award under the Plan (whether a performance-vesting Stock Option, Cash Performance Award or otherwise) based on financial statements that are subsequently required to be restated in a way that would decrease the value of such compensation, the Participant will, to the extent not otherwise prohibited by law, forfeit and repay to the Company the difference between what the Participant received and what the Participant should have received based on the accounting restatement, (i) in accordance with the Company's compensation recovery, "clawback" or similar policy, as may be in effect from time to time, or (ii) in accordance with any compensation recovery, "clawback" or similar policy made applicable by law including the provisions of Section 945 of the Dodd-Frank Wall Street Reform and Consumer Protection Act and the rules, regulations and requirements adopted thereunder by the Securities and Exchange Commission and/or any national securities exchange on which the Company's equity securities may be listed (the "Policy"). By accepting an Award hereunder, the Participant acknowledges and agrees that the Policy shall apply to such Award, and all incentive-based compensation payable pursuant to such Award shall be subject to forfeiture and repayment pursuant to the terms of the Policy.
- 14. Retirement. Unless otherwise provided in an Award Agreement that expressly refers to this Section 14 (or any successor provision of this Section 14), an Award held by a Retirement Eligible Employee shall continue to vest (and, in the case of an Award the payout of which is tied to one or more Performance Goals, be earned) in accordance with the terms of such Award following the Retirement of such Retirement Eligible Employee, notwithstanding the Retirement Eligible Employee's termination of employment with the Company and its Subsidiaries. For avoidance of doubt, any such Award shall continue to be subject to the terms of the Plan, including, without limitation, Section 13.2 and Section 13.3. Each Award that is outstanding on the Effective Date is hereby amended to incorporate the provisions of this Section 14.
- 15. <u>Transfer, Leave of Absence, Etc.</u> For purposes of the Plan, except as otherwise determined by the Committee, the following events shall not be deemed a termination of employment: (a) a transfer of a Participant's employment to the Company from a Subsidiary or from the Company to a Subsidiary, or from one Subsidiary to another; or (b) an approved leave of absence for military service or sickness, or for any other purpose approved by the Company, if the employee's right to re-employment is guaranteed either by a statute or by contract or under the policy pursuant to which the leave of absence was granted or if the Committee otherwise so provides in writing.

16. General Provisions.

16.1 Status of Plan. The Committee may authorize the creation of trusts or other arrangements to meet the Company's obligations to deliver stock or make payments with respect to Awards.

16.2 Award Agreement. Each Award under the Plan shall be evidenced by an Award Agreement in a written or electronic form approved by the Committee setting forth the number of shares of Common Stock or Restricted Stock Units subject to the Award, the exercise price, base price or purchase price of the Award, the time or times at which an Award will become vested, exercisable or payable and the term of the Award. The Award Agreement also may set forth the effect on an Award of a Change in Control or a termination of Service under certain circumstances. The Award Agreement shall be subject to and incorporate, by reference or otherwise, all of the applicable terms and conditions of the Plan, and also may set forth other terms and conditions applicable to the Award as determined by the Committee consistent with the limitations of the Plan. The grant of an Award under the Plan shall not confer any rights upon the Participant holding such Award other than such terms, and subject to such conditions, as are specified in the Plan as being applicable to such type of Award (or to all Awards) or as are expressly set forth in the Award Agreement. The Committee need not require the execution of an Award Agreement by a Participant, in which case, acceptance of the Award by the Participant shall constitute agreement by the Participant to the terms, conditions, restrictions and limitations set forth in the Plan and the Award Agreement as well as the administrative guidelines of the Company in effect from time to time. In the event of any conflict between the provisions of the Plan and any Award Agreement, the provisions of the Plan shall prevail.

16.3 No Assignment or Transfer; Beneficiaries. Except as provided in Section 6.7(e) hereof or as otherwise determined by the Committee, Awards under the Plan shall not be assignable or transferable by the Participant, and shall not be subject in any manner to assignment, alienation, pledge, encumbrance or charge. Notwithstanding the foregoing, in the event of the death of a Participant, except as otherwise provided by the

Committee in an Award Agreement, an outstanding Award may be exercised by or shall become payable to the Participant's beneficiary as designated by the Participant in the manner prescribed by the Committee or, in the absence of an authorized beneficiary designation, by a legatee or legatees of such Award under the participant's last will or by such Participant's executors, personal representatives or distributees of such Award in accordance with the Participant's will or the laws of descent and distribution. The Committee may provide in the terms of an Award Agreement or in any other manner prescribed by the Committee that the Participant shall have the right to designate a beneficiary or beneficiaries who shall be entitled to any rights, payments or other benefits specified under an Award following the Participant's death.

(a) Notwithstanding the foregoing, if authorized in the applicable Award Agreement, a Grantee may transfer, not for value, all or part of an Option that is not an Incentive Stock Option to any Family Member. For the purpose of this Section 16.3(a), a "not for value" transfer is a transfer which is (i) a gift, (ii) a transfer under a domestic relations order in settlement of marital property rights; or (iii) unless applicable law does not permit such transfers, a transfer to an entity in which more than fifty percent of the voting interests are owned by Family Members (or the Grantee) in exchange for an interest in that entity. Following a transfer under this Section 16.3(a), any such Option shall continue to be subject to the same terms and conditions as were applicable immediately prior to transfer, and shares of Stock acquired pursuant to the Option shall be subject to the same restrictions on transfer of shares as would have applied to the Grantee. Subsequent transfers of transferred Options are prohibited except to Family Members of the original Grantee in accordance with this Section 16.3(a) or by will or the laws of descent and distribution. The events of termination of Service under an Option shall continue to be applied with respect to the original Grantee, following which the Option shall be exercisable by the transferee only to the extent, and for the periods specified in the applicable Award Agreement, and the shares may be subject to repurchase by the Company or its assignee.

16.4 Deferrals of Payment. The Committee may in its discretion permit a Participant to defer the receipt of payment of cash or delivery of shares of Common Stock that would otherwise be due to the Participant by virtue of the exercise of a right or the satisfaction of vesting or other conditions with respect to an Award; provided, however, that such discretion shall not apply in the case of a Stock Option or Stock Appreciation Right. If any such deferral is to be permitted by the Committee, the Committee shall establish rules and procedures relating to such deferral in a manner intended to comply with the requirements of Section 409A of the Code, including, without limitation, the time when an election to defer may be made, the time period of the deferral and the events that would result in payment of the deferred amount, the interest or other earnings attributable to the deferral and the method of funding, if any, attributable to the deferred amount.

16.5 *No Right to Employment or Continued Service*. Nothing in the Plan, in the grant of any Award or in any Award Agreement shall confer upon any Eligible Person or any Participant any right to continue in the Service of the Company or any of its Subsidiaries or interfere in any way with the right of the Company or any of its Subsidiaries to terminate the employment or other service relationship of an Eligible Person or a Participant for any reason at any time.

16.6 Stock Certificates. The Committee may determine in its discretion the manner of delivery of Common Stock to be issued under the Plan, which may be by delivery of stock certificates, electronic account entry into new or existing accounts or any other means as the Committee, in its discretion, deems appropriate. The Committee may require that the stock certificates (if any) be held in escrow by the Company for any shares of Common Stock or cause the shares to be legended in order to comply with the securities laws or other applicable restrictions or should the shares of Common Stock be represented by book or electronic account entry rather than a certificate, the Committee may take such steps to restrict transfer of the shares of Common Stock as the Committee considers necessary or advisable.

16.7 *Trading Policy Restrictions*. Option exercises and other Awards under the Plan shall be subject to such Company insider-trading-policy-related restrictions, terms and conditions to the extent established by the Committee, or in accordance with policies set by the Committee, from time to time.

16.8 Section 409A Compliance. To the maximum extent possible, it is intended that the Plan and all Awards hereunder comply with, or be exempt from, the requirements of Section 409A of the Code and the Treasury Regulations and other guidance issued thereunder, and that the Plan and all Award Agreements shall be interpreted and applied by the Committee in a manner consistent with this intent in order to avoid the imposition of any additional tax under Section 409A of the Code. Subject to Section 16.4 hereof, any payments due pursuant to this Plan shall be payable to the Participant no later than two-and-a-half months following the end of the taxable year in which the payments are earned (subject to a reasonable delay in payment due to an unforeseeable event making it administratively impracticable to make the payment by such time), and in no event shall the payments be made later than the end of the taxable year following the taxable year in which the payments are earned. In the event that any payment under this Plan is contingent upon the execution of a release, and the applicable release spans two of the

Participant's taxable years, the applicable payments must be made in the second of the two taxable years. In the event that any (i) provision of the Plan or an Award Agreement, (ii) Award, payment, transaction or (iii) other action or arrangement contemplated by the provisions of the Plan is determined by the Committee to not comply with the applicable requirements of Section 409A of the Code and the Treasury Regulations and other guidance issued thereunder, the Committee shall have the authority to take such actions and to make such changes to the Plan or an Award Agreement as the Committee deems necessary to comply with such requirements. No payment that constitutes deferred compensation under Section 409A of the Code that would otherwise be made under the Plan or an Award Agreement upon a termination of Service will be made or provided unless and until such termination is also a "separation from service," as determined in accordance with Section 409A of the Code. Notwithstanding the foregoing or anything elsewhere in the Plan or an Award Agreement to the contrary, if a Participant is a "specified employee" as defined in Section 409A of the Code at the time of termination of Service with respect to an Award, then solely to the extent necessary to avoid the imposition of any additional tax under Section 409A of the Code, the commencement of any payments or benefits under the Award shall be deferred until the date that is six (6) months following the Participant's termination of Service (or, if earlier, the date of death of the specified employee) and shall instead be paid (in a manner set forth in the Award Agreement) on the payment date that immediately follows the end of such six- (6-) month period (or death) or as soon as administratively practicable within thirty (30) days thereafter, but in no event later than the end of the applicable taxable year. In no event whatsoever shall the Company be liable for any additional tax, interest or penalties that may be imposed on a Participant by Section 409A of the C

16.9 Securities Law Compliance.

- (a) *General*. No shares of Common Stock will be issued or transferred pursuant to an Award unless and until all then applicable requirements imposed by Federal and state securities and other laws, rules and regulations and by any regulatory agencies having jurisdiction, and by any exchanges upon which the shares of Common Stock may be listed, have been fully met. As a condition precedent to the issuance of shares pursuant to the grant or exercise of an Award, the Company may require the Participant to take any reasonable action to meet such requirements. The Committee may impose such conditions on any shares of Common Stock issuable under the Plan as it may deem advisable, including, without limitation, restrictions under the Securities Act of 1933, as amended, under the requirements of any exchange upon which such shares of the same class are then listed, and under any blue sky or other securities laws applicable to such shares. The Committee may also require the Participant to represent and warrant at the time of issuance or transfer that the shares of Common Stock are being acquired only for investment purposes and without any current intention to sell or distribute such shares
- (b) *Compliance with Rule 701*. To the extent that any Awards are granted prior to a Public Offering and the filing of an effective registration statement on Form S-8, the Plan is intended to be a written compensatory benefit plan within the meaning of Rule 701 promulgated under the Securities Act and, therefore, such Awards are subject to the restrictions set forth in Rule 701, and are "restricted securities," as such term is defined in Rule 144 promulgated under the Securities Act, and any resale of the Shares underlying such Awards must be in compliance with the registration requirements of the Securities Act or an exemption therefrom. Awards issued pursuant to the Plan prior to a Public Offering and the filling of an effective registration statement on Form S-8 shall in no event exceed the limitations set forth in Rule 701(d), as applicable from time to time.
- 16.10 *Substitute Awards in Corporate Transactions*. Nothing contained in the Plan shall be construed to limit the right of the Committee to grant Awards under the Plan in connection with the acquisition, whether by purchase, merger, consolidation or other corporate transaction, of the business or assets of any corporation or other entity. Without limiting the foregoing, the Committee may grant Awards under the Plan to an employee, director or other individual service provider of another corporation who becomes an Eligible Person by reason of any such corporate transaction in substitution for awards previously granted by such corporation or entity to such person. The terms and conditions of the substitute Awards may vary from the terms and conditions that would otherwise be required by the Plan solely to the extent the Committee deems necessary for such purpose. Any such substitute awards shall not (a) reduce the number of shares of Common Stock available for issuance under the Plan, (b) be subject to or counted against the Award limits specified in Section 4.3, 4.4 or 10.7 hereof or (c) replenish the Share Reserve upon the occurrence of any event set forth in Section 4.2 hereof.
- 16.11 *Tax Withholding*. The Company shall have the power and the right to deduct or withhold automatically from any amount deliverable under the Award or otherwise, or require a Participant to remit to the Company or the applicable Subsidiary, up to an amount based on the maximum statutory tax rates in the Participant's applicable tax jurisdiction or such other rate that will not trigger a negative accounting impact on the Company.. With respect to required withholding, Participants may elect (subject to the Company's automatic withholding right set out above) to satisfy the withholding requirement with respect to any taxable event arising as a result of the Plan, in whole or in part, by the methods described in Section 6.5 hereof with respect to Stock Options

- 16 -

or by a method similar to the methods described in Section 6.5 hereof with respect to Awards other than Stock Options (except as otherwise set forth in an Award Agreement).

- 16.12 *Unfunded Plan*. The adoption of the Plan and any reservation of shares of Stock or cash amounts by the Company to discharge its obligations hereunder shall not be deemed to create a trust or other funded arrangement. Except upon the issuance of Common Stock pursuant to an Award, any rights of a Participant under the Plan shall be those of a general unsecured creditor of the Company, and neither a Participant nor the Participant's permitted transferees or estate shall have any other interest in any assets of the Company by virtue of the Plan. Notwithstanding the foregoing, the Company shall have the right to implement or set aside funds in a grantor trust, subject to the claims of the Company's creditors or otherwise, to discharge its obligations under the Plan. The Plan is not subject to the U.S. Employee Retirement Income Security Act of 1974, as amended from time to time.
- 16.13 Other Compensation and Benefit Plans. The adoption of the Plan shall not affect any other share incentive or other compensation plans in effect for the Company or any Subsidiary, nor shall the Plan preclude the Company or any Subsidiary from establishing any other forms of share incentive or other compensation or benefit program for employees of the Company or any Subsidiary. The amount of any compensation deemed to be received by a Participant pursuant to an Award shall not constitute includable compensation for purposes of determining the amount of benefits to which a Participant is entitled under any other compensation or benefit plan or program of the Company or a Subsidiary, including, without limitation, under any pension or severance benefits plan, except to the extent specifically provided by the terms of any such plan.
- 16.14 *Plan Binding on Transferees*. The Plan shall be binding upon the Company, its transferees and assigns, and the Participant, the Participant's executor, administrator and permitted transferees and beneficiaries.
- 16.15 *Severability*. If any provision of the Plan or any Award Agreement shall be determined to be illegal or unenforceable by any court of law in any jurisdiction, the remaining provisions hereof and thereof shall be severable and enforceable in accordance with their terms, and all provisions shall remain enforceable in any other jurisdiction.
- 16.16 *Governing Law*. The Plan and all rights hereunder shall be subject to and interpreted in accordance with the laws of the State of Colorado, without reference to the principles of conflicts of laws, and to applicable Federal securities laws.
- 16.17 *No Fractional Shares*. No fractional shares of Common Stock shall be issued or delivered pursuant to the Plan or any Award, and the Committee shall determine (i) whether cash, other securities or other property shall be paid or transferred in lieu of any fractional shares of Common Stock or (ii) whether such fractional shares or any rights thereto shall be canceled, terminated or otherwise eliminated (in the case of this clause (ii), with no consideration paid therefor).
- 16.18 *No Guarantees Regarding Tax Treatment*. Neither the Company nor the Committee make any guarantees to any person regarding the tax treatment of Awards or payments made under the Plan. Neither the Company nor the Committee has any obligation to take any action to prevent the assessment of any tax on any person with respect to any Award under Section 409A of the Code, Section 4999 of the Code, Section 280G of the Code or otherwise, and neither the Company nor the Committee shall have any liability to a person with respect thereto.
- 16.19 *Data Protection*. By participating in the Plan, each Participant consents to the collection, processing, transmission and storage by the Company, its Subsidiaries and any third party administrators of any data of a professional or personal nature for the purposes of administering the Plan.
- 16.20 Awards to Non-U.S. Participants. To comply with the laws in countries other than the United States in which the Company or any of its Subsidiaries or affiliates operates or has employees, Non-Employee Directors or consultants, the Committee, in its sole discretion, shall have the power and authority to (i) modify the terms and conditions of any Award granted to Participants outside the United States to comply with applicable foreign laws, (ii) take any action, before or after an Award is made, that it deems advisable to obtain approval or comply with any necessary local government regulatory exemptions or approvals and (iii) establish subplans and modify exercise procedures and other terms and procedures, to the extent such actions may be necessary or advisable. Any subplans and modifications to Plan terms and procedures established under this Section 16.20 by the Committee shall be attached to this Plan document as appendices.

16.21 *Regulatory Compliance*. The Committee shall take all necessary steps to ensure the Plan is being interpreted and administered in accordance with the Dodd-Frank Wall Street Reform and Consumer Protection Act and any regulations promulgated thereunder, as well as Regulation O and other regulations promulgated by the Board of Governors of the Federal Reserve System.

17. Restrictions on Transfer of Shares of Stock.

- 17.1 Right of First Refusal. Subject to Section 17.4 below, a Grantee (or such other individual who is entitled to exercise an Option or otherwise acquire shares pursuant to a Grant under the terms of this Plan) shall not sell, pledge, assign, gift, transfer, or otherwise dispose of any shares of Stock acquired pursuant to a Grant to any person or entity without first offering such shares to the Company for purchase on the same terms and conditions as those offered the proposed transferee. The Company may assign its right of first refusal under this Section 17.1 in whole or in part, to (1) any holder of stock or other securities of the Company (a "Stockholder"), (2) any Affiliate or (3) any other person or entity that the Board determines has a sufficient relationship with or interest in the Company. The Company shall give reasonable written notice to the Grantee of any such assignment of its rights. The restrictions of this Section 17.1 apply to any person to whom Stock that was originally acquired pursuant to a Grant is sold, pledged, assigned, bequeathed, gifted, transferred or otherwise disposed of, without regard to the number of such subsequent transferees or the manner in which they acquire the Stock, but the restrictions of this Section 17.1 do not apply to a transfer of Stock that occurs as a result of the death of the Grantee or of any subsequent transferee (but shall apply to the executor, the administrator or personal representative, the estate, and the legatees, beneficiaries and assigns thereof).
- 17.2 Repurchase and Other Rights. Stock issued upon exercise of a Grant or pursuant to the Grant of Restricted Stock may be subject to such right of repurchase or other transfer restrictions as the Board may determine, consistent with applicable law. Any such additional restriction shall be set forth in the Award Agreement.

17.3 Installment Payments.

- (a) *General Rule*. In the case of any purchase of Stock or an Option under this Section 17, the Company or its permitted assignee may pay the Grantee, transferee of the Option or other registered owner of the Stock the purchase price in three or fewer annual installments. Interest shall be credited on the installments at the applicable federal rate (as determined for purposes of Section 1274 of the Code) in effect on the date on which the purchase is made. The Company or its permitted assignee shall pay at least one-third of the total purchase price each year, plus interest on the unpaid balance, with the first payment being made on or before the 60th day after the purchase.
- (b) Exception in the Case of Stock Repurchase Right. If an Award Agreement authorizes, upon the Grantee's termination of Service, the repurchase of shares of Stock acquired by the Grantee pursuant to the exercise of an Option or under a Grant of Restricted Stock, to the extent required by applicable law, payment shall be made in cash or by cancellation of indebtedness within the later of 90 days from the date of termination of Service or 90 days from the date of exercise or purchase, as the case may be.
- 17.4 *Publicly Traded Stock*. If the Stock is listed on an established national or regional stock exchange or is admitted to quotation on The Nasdaq Stock Market, Inc., or is publicly traded in an established securities market, the foregoing transfer restrictions of Sections 17.1 and 17.2 shall terminate as of the first date that the Stock is so listed, quoted or publicly traded.
- 17.5 *Legend*. In order to enforce the restrictions imposed upon shares of Stock under this Plan or as provided in an Award Agreement, the Board may cause a legend or legends to be placed on any certificate representing shares issued pursuant to this Plan that complies with the applicable securities laws and regulations and makes appropriate reference to the restrictions imposed under it.
 - 18. Term; Amendment and Termination; Stockholder Approval.
- 18.1 *Term*. The Plan shall be effective as of the date of adoption by the Board (the "*Effective Date*"). Subject to Section 18.2 hereof, the Plan shall terminate on the tenth anniversary of the Original Effective Date.
- 18.2 Amendment and Termination. The Board may from time to time and in any respect, amend, modify, suspend or terminate the Plan; provided, that, except as provided in Section 16.8 or 16.20 or as otherwise determined by the Committee as it deems necessary to comply with applicable laws, no amendment, modification, suspension or termination of the Plan shall adversely affect any Award theretofore granted without the consent of

the Participant or the permitted transferee of the Award. The Company will obtain approval by the Company's stockholders of any Plan amendment to the extent it is necessary or advisable to comply with the legal and regulatory requirements relating to administration of equity-based awards and the related issuance of shares of Common Stock thereunder, including by not limited to U.S. federal and state corporate laws, U.S. federal and state securities laws, the Code, and any stock exchange or quotation system on which the Common Stock is listed or quoted.

* * * *